



## Report to the NHS Somerset Integrated Care Board on 30 March 2023

# Title: Delegation of Pharmacy, Optometry and Dental Services

Enclosure F

Version Number / Status:	1
Executive Lead	Dr Bernie Marden, Chief Medical Officer
Clinical Lead:	Dr Bernie Marden, Chief Medical Officer
Author:	Tanya Whittle, Deputy Director of Primary Care and Contracting

### Summary and Purpose of Paper

Over the last 12 months, NHS Somerset ICB has been working with the regional team and fellow ICBs in the South West to take forward the delegation of pharmacy, optometry and dental services. All parties have agreed an operating model whereby a collaborative commissioning hub (comprising current NHSE staff) continue to undertakes these functions on behalf of the 7 ICBs. This collaborative work culminated in the completion of a safe delegation checklist, which was submitted to NHSE in February 2023. At the time it was recognised that not all items would be addressed fully by 31 March 2023 and further work would be required to ensure a smooth transition.

On 23 March 2023, Jonathan Higman, on behalf of NHS Somerset ICB, signed and submitted the attached nationally-produced delegation agreement, along with a locally-produced memorandum of understanding. The covering letter and current risk register for these services is also attached for your information. Following a more detailed review, the risks will be incorporated into the overall risk management approach for NHS Somerset.

Whilst much progress has been made in understanding the functions associated with delegation, we recognise that further work is still required and will be taken forward through the development and implementation of a transition workplan, which colleagues in NHS Somerset will play an active role in. The areas of priority for the transition plan are highlighted in the attached covering letter.

### **Recommendations and next steps**

The Board is asked to note the signing of the delegation agreement and the further work required to ensure a smooth transition of the delegation of the functions associated with the commissioning of pharmacy, dental and optometry services. The Board is also asked to note the risks that have been identified in relation to these commissioning functions.

The Primary Care Commissioning Committee will continue to receive updates on the delegation of these services as well as receiving regular reports relating to the commissioning agenda.

Not applicable					
A quality SOP has been developed and included within the memorandum of understanding.					
The approach to safeguarding has been included as part of the local memorandum of understanding.					
Not applicable					
The transition work has been taken forward with colleagues across the 7 ICBs in the South West and the regional team. A communications plan will also be part of the transition.					
A finance SOP has been developed and included within the memorandum of understanding.					
The Primary Care Commissioning Committee terms of reference will be updated to reflect this change in accountability. Integrated Care Boards will be responsible for any legal costs associated with these functions. Discussions are ongoing about an appropriate transfer of funding.					
Not applicable.					
There is risk that NHS Somerset Integrated Care Board is unable to deliver obligations regarding delegated pharmacy, optometry and dental services post 04/23					
Likelihood RAG Rating GBAF Ref					
4 16 566					
as e C Di Di					



Our Ref: JH/jch/230323-18

23 March 2023



Wynford House Lufton Way Lufton Yeovil Somerset BA22 8HR

Tel: 01935 384000

somicb.enquiries@nhs.net

Rachel Pearce Regional Director Commissioning (South West) NHS England

(via email to: Rachel.pearce12@nhs.net)

# Dear Rachel

NHS Somerset Integrated Care Board is keen to embrace the opportunities that POD delegation will bring, particularly the opportunity to collaborate and integrate services to improve provision and access for the population of Somerset. Please find enclosed the signed delegation agreement and memorandum of understanding for Somerset, which has been shared with our Board members via email. The delegation agreement, this letter and the risk register will also be discussed at next week's public Board meeting.

In conjunction with our SW ICB colleagues and the regional team, we have been making preparations for delegation over the last 12 months. Whilst much progress has been made, we recognise that there are still some significant areas of work that need to be encapsulated within a transition plan. Our delegation agreement and MOU has been signed with the following understanding:

- A transition plan will be jointly developed and address the amber/red ratings identified within the final version of the Safe Delegation Checklist, with the first draft being available before the end of March 2023.
- The transition plan will include specific workstreams relating to information governance, clinical governance (Including risk management processes, patient safety, complaints management post 1 July 2023), decision-making/governance structures and management of conflicts of interest.
- The transition planning framework will have a structure to ensure actions are progressed at pace and monitored accordingly.
- The Data Processing Agreement is finalised and signed by 31 March 2023.
- Whilst a structure chart has been received, there has not been clarity on the current level of vacancies within the structure. We ask that the hub structure at at 1 April 2023 is appended to the MOU.
- The risk register was received on 22 March 2022 and we note that there are 6 risks rated at 16 and 2 risks rated at 20. The ICB will need to review the detail of this risk register and may require further assurances through the transition plan as a result of this review.





We recognise that there are separate discussions regarding the hosting of the hub from 1 July 2023 so we do not plan to comment on this area specifically as part of our response to the delegation agreement.

We remain committed to working with the NHS England regional team, Collaborative Commissioning Hub staff and fellow ICBs in the South West to ensure we operate as 'one team' to achieve our shared ambitions for health and service improvement this year.

Should you have any further queries please do not hesitate to contact me.

Yours sincerely

Jonathan Higman Chief Executive

Encs

Copy: Bernie Marden, Chief Medical Officer Suresh Ariaratnam, Non-Executive Director Tanya Whittle, Deputy Director of Primary Care and Contracting Dated: 23 March 2023

(1) NHS ENGLAND

- and -

# (2) NHS SOMERSET INTEGRATED CARE BOARD

# **Delegation Agreement in Respect of**

- (i) Primary Medical Care Services
- (ii) Primary Dental Services and Prescribed Dental Services
- (iii) Primary Ophthalmic Services
- (iv) Pharmaceutical Services and Local Pharmaceutical Services

### Table of contents

Clause heading and number

1.	PARTICULARS	3		
2.	INTERPRETATION	5		
3.	BACKGROUND	5		
4.	TERM	5		
5.	PRINCIPLES	5		
6.	DELEGATION	6		
7.	EXERCISE OF DELEGATED FUNCTIONS	7		
8.	PERFORMANCE OF THE RESERVED FUNCTIONS	8		
9.	FINANCE	8		
10.	INFORMATION, PLANNING AND REPORTING	11		
11.	FURTHER ARRANGEMENTS	11		
12.	STAFFING AND WORKFORCE	12		
13.	BREACH	12		
14.	ESCALATION RIGHTS	13		
15.	LIABILITY AND INDEMNITY	13		
16.	CLAIMS AND LITIGATION	14		
17.	DATA PROTECTION, FREEDOM OF INFORMATION AND TRANSPARENCY	16		
18.	IT INTER-OPERABILITY	16		
19.	CONFLICTS OF INTEREST AND TRANSPARENCY ON GIFTS AND HOSPITALITY	17		
20.	PROHIBITED ACTS AND COUNTER-FRAUD	17		
21.	CONFIDENTIAL INFORMATION OF THE PARTIES	18		
22.	INTELLECTUAL PROPERTY	19		
23.	NOTICES	19		
24.	DISPUTES	19		
25.	VARIATIONS	20		
26.	TERMINATION	20		
27.	CONSEQUENCE OF TERMINATION	21		
28.	PROVISIONS SURVIVING TERMINATION	22		
29.	COSTS	23		
30.	SEVERABILITY	23		
31.	GENERAL	23		
SCHEDULE 1 – DEFINITIONS AND INTERPRETATION				
SCHEDULE 2 – DELEGATED FUNCTION				
SCHED	DULE 3 – RESERVED FUNCTION	67		

SCHEDULE 4 – FURTHER INFORMATION GOVERNANCE AND SHARING PROVISIONS	. 73
SCHEDULE 5 – FINANCIAL PROVISIONS AND DECISION MAKING LIMITS	. 80
SCHEDULE 6 – MANDATED ASSISTANCE AND SUPPORT	. 82
SCHEDULE 7 – LOCAL TERMS	. 84
SCHEDULE 8 – DEPLOYMENT OF NHS ENGLAND STAFF TO THE ICB	. 85
SCHEDULE 9 – MANDATED GUIDANCE	. 88
SCHEDULE 10 – ADMINISTRATIVE AND MANAGEMENT SERVICES	. 90

#### DELEGATION AGREEMENT FOR SPECIFIED FUNCTIONS

#### 1. **PARTICULARS**

1.1 This Agreement records the particulars of the agreement made between NHS England and the Integrated Care Board (ICB) named below.

Integrated Care Board	NHS Somerset Integrated Care Board	
Area	Area of the ICB as defined in its Constitution	
Date of Agreement	The date stated on the front page of this document	
ICB Representative	Tanya Whittle, Deputy Director of Primary Care and Contracting	
ICB Email Address for Notices	tanya.whittle2@nhs.net	
NHS England Representative	[Insert details of name of the manager of this Agreement for NHS England]	

NHS England Email Address for [Insert Address] Notices

1.2 The following Delegated Functions are included in this Agreement<sup>1</sup>:

Delegated Functions	Schedule	Included	Effective Date of Delegation
Primary Medical Services Functions	Schedule 2A –	Yes	1 July 2022
Primary Dental Services and Prescribed Dental Services Functions	Schedule 2B –	Yes	1 April 2023
Primary Ophthalmic Services Functions	Schedule 2C –	Yes	1 April 2023
Pharmaceutical Services and Local Pharmaceutical Services Functions	Schedule 2D –	Yes	1 April 2023

- 1.3 This Agreement comprises:
  - 1.3.1 the Particulars (clause 1);
  - 1.3.2 the Terms and Conditions (clauses 2 to 31);

<sup>&</sup>lt;sup>1</sup> This table <u>must</u> be completed to indicate which services are included in the Delegation.

- 1.3.3 the Schedules; and
- 1.3.4 the Mandated Guidance

Signed by NHS England

[Name]

[Title]

(for and on behalf of NHS England)

 Signed by
 NHS Somerset Integrated Care Board

 Jonathan Higman
 Jonether Executive

 for and on behalf of NHS Somerset Integrated Care Board

#### TERMS AND CONDITIONS

#### 2. **INTERPRETATION**

- 2.1 This Agreement is to be interpreted in accordance with SCHEDULE 1 (Definitions and Interpretation).
- 2.2 If there is any conflict or inconsistency between the provisions of this Agreement, that conflict or inconsistency must be resolved according to the following order of priority:
  - 2.2.1 the Particulars and Terms and Conditions (clauses 1 to 31);
  - 2.2.2 all Schedules excluding Local Terms;
  - 2.2.3 Mandated Guidance; and
  - 2.2.4 Local Terms.
- 2.3 This Agreement constitutes the entire agreement and understanding between the Parties relating to the Delegation and supersedes all previous agreements, promises and understandings between them, whether written or oral, relating to its subject matter.
- 2.4 Where it is indicated that a provision in this Agreement is not used, that provision is not relevant and has no application in this Agreement.
- 2.5 Where a particular clause is included in this Agreement but is not relevant to the ICB because that clause relates to matters which do not apply to the ICB (for example, if the clause only relates to functions that are not Delegated Functions in respect of the ICB), that clause is not relevant and has no application to this Agreement.

#### 3. BACKGROUND

- 3.1 By this Agreement NHS England delegates the Delegated Functions to the ICB under section 65Z5 of the NHS Act while retaining the Reserved Functions.
- 3.2 Arrangements made under section 65Z5 may be made on such terms and conditions (including terms as to payment) as may be agreed between NHS England and the ICB.
- 3.3 This Agreement sets out the terms that apply to the exercise of the Delegated Functions by the ICB and the Parties' associated responsibilities and measures required to ensure the effective and efficient exercise of the Delegated Functions and Reserved Functions.

#### 4. **TERM**

4.1 This Agreement has effect from the Date of Agreement set out in the Particulars and will remain in force unless terminated in accordance with clause 26 *(Termination)* below.

#### 5. **PRINCIPLES**

- 5.1 In complying with the terms of this Agreement, NHS England and the ICB must:
  - 5.1.1 at all times have regard to the Triple Aim;
  - 5.1.2 at all times act in good faith and with integrity towards each other;
  - 5.1.3 have regard to the intention that commissioning functions in respect of Primary Medical Services, Primary Dental Services and Primary Ophthalmic Services will in future be directly conferred on the ICB;
  - 5.1.4 consider how it can meet its legal duties to involve patients and the public in shaping the provision of services, including by working with local

communities, under-represented groups and those with protected characteristics for the purposes of the Equality Act 2010;

- 5.1.5 consider how in performing their obligations they can address health inequalities;
- 5.1.6 at all times exercise functions effectively, efficiently and economically;
- 5.1.7 act in a timely manner;
- 5.1.8 share information and best practice, and work collaboratively to identify solutions and enhance the evidence base for the commissioning and provision of health services, eliminate duplication of effort, mitigate risk and reduce cost; and
- 5.1.9 have regard to the needs and views of the other Party and as far as is lawful and reasonably practicable, take such needs and views into account.

#### 6. **DELEGATION**

- 6.1 In accordance with its statutory powers under section 65Z5 of the NHS Act, NHS England hereby delegates the exercise of the Delegated Functions to the ICB to empower it to commission a range of services for its Population, as further described in this Agreement ("**Delegation**").
- 6.2 The Delegated Functions are the functions described as being delegated to the ICB as have been identified as included in clause 1 (*Particulars*) and included as a Schedule to this Agreement.
- 6.3 The Delegation in respect of each Delegated Function has effect from the relevant Effective Date of Delegation.
- 6.4 NHS England may by Contractual Notice allocate Contracts to the ICB such that they are included as part of the Delegation. The Delegated Functions must be exercised both in respect of the relevant Contract and any related matters concerning any Provider that is a party to Contract or Arrangement.
- 6.5 Subsequent to the Effective Date of Delegation and for the duration of this Agreement, unless otherwise agreed any new Contract entered into in respect of the Delegated Functions shall be managed by the ICB in accordance with the provisions of this Agreement.
- 6.6 NHS England may by Contractual Notice add or remove Contracts to or from the list of those allocated to the ICB for the purposes of this Agreement. In particular, NHS England may add or remove Contracts where this is associated with an extension or reduction of the scope of the Delegated Functions.
- 6.7 Decisions of the ICB in respect of the Delegated Functions and made in accordance with the terms of this Agreement shall be binding on NHS England and the ICB.
- 6.8 Unless expressly provided for in this Agreement, the ICB is not authorised by this Agreement to take any step or make any decision in respect of Reserved Functions. Any such purported decision of the ICB is invalid and not binding on NHS England unless ratified in writing by NHS England in accordance with the NHS England Scheme of Delegation and Standing Financial Instructions. NHS England may, acting reasonably and solely to the extent that the decision relates to the Delegated Functions, substitute its own decision for any decision which the ICB purports to make where NHS England reasonably considers that the impact of the ICB decision could, in relation to the Delegated Functions, cause the ICB to be acting unlawfully, in breach of this Agreement including Mandated Guidance, or in breach of any Contract. The ICB must

provide any information, assistance and support as NHS England requires to enable it to determine whether to make any such decision.

6.9 The terms of clause 6.8 are without prejudice to the ability of NHS England to enforce the terms of this Agreement or otherwise take action in respect of any failure by the ICB to comply with this Agreement.

#### 7. EXERCISE OF DELEGATED FUNCTIONS

- 7.1 The ICB agrees that it will exercise the Delegated Functions in accordance with:
  - 7.1.1 the terms of this Agreement including Mandated Guidance;
  - 7.1.2 any Contractual Notices;
  - 7.1.3 all applicable Law and Guidance;
  - 7.1.4 the ICB's constitution;
  - 7.1.5 the requirements of any assurance arrangements made by NHS England, and;
  - 7.1.6 Good Practice.
- 7.2 In exercising the Delegated Functions, the ICB must comply with the Mandated Guidance set out at SCHEDULE (*Mandated Guidance*) or otherwise referred to in the Schedules to this Agreement and such further Mandated Guidance as may be issued by NHS England from time to time, including on the Direct Commissioning Guidance Webpage.
- 7.3 NHS England may, at its discretion, issue Contractual Notices from time to time relating to the manner in which the Delegated Functions must be exercised by the ICB. Contractual Notices will have effect as variations to this Agreement.
- 7.4 The ICB must establish effective, safe, efficient and economic arrangements for the discharge of the Delegated Functions.
- 7.5 The ICB must give due consideration to whether any of the Delegated Functions should be exercised collaboratively with other NHS bodies or Local Authorities including, without limitation, by means of arrangements under section 65Z5 and section 75 of the NHS Act.
- 7.6 The ICB must develop an operational scheme(s) of delegation defining those individuals or groups of individuals, including committees, who may discharge aspects of the Delegated Functions. For the purposes of this clause, the ICB may include the operational scheme(s) of delegation within its general organisational scheme of delegation.
- 7.7 Subject to clauses 7.1 to 7.6, the ICB may determine the arrangements for the exercise of the Delegated Functions.
- 7.8 The ICB must perform the Delegated Functions:
  - 7.8.1 in such a manner as to ensure NHS England's compliance with NHS England's statutory duties in respect of the Delegated Functions and to enable NHS England to fulfil its Reserved Functions; and
  - 7.8.2 having regard to NHS England's accountability to the Secretary of State and Parliament in respect of both the Delegated Functions and Reserved Functions.

#### 8. **PERFORMANCE OF THE RESERVED FUNCTIONS**

- 8.1 NHS England will exercise the Reserved Functions, including but not limited to those set out in the relevant Schedules to this Agreement.
- 8.2 For the avoidance of doubt, the Parties acknowledge that the Delegation may be amended, and additional functions may be delegated to the ICB, in which event consequential changes to this Agreement shall be agreed with the ICB pursuant to clause 25 (*Variations*) of this Agreement.
- 8.3 Where appropriate NHS England will work collaboratively with the ICB when exercising the Reserved Functions.
- 8.4 If there is any conflict or inconsistency between functions that are named as Delegated Functions and functions that are named as Reserved Functions, then such functions shall be interpreted as Reserved Functions unless and until NHS England confirms otherwise. In the event that an ICB identifies such a conflict or inconsistency it will inform NHS England as soon as is reasonably practicable.
- 8.5 The Parties acknowledge that where the ICB shall provide administrative and management services to NHS England in relation to certain Reserved Functions these shall be as set out in clause 9.14. and SCHEDULE (*Administrative and Management Services*).
- 8.6 The Parties further acknowledge that NHS England may ask the ICB to provide certain administrative and management services to NHS England in relation to other Reserved Functions.
- 8.7 Notwithstanding any arrangement for or provision of administrative or management services in respect of certain Reserved Functions, NHS England shall retain and be accountable for the exercise of such Reserved Functions.

#### 9. FINANCE

- 9.1 Without prejudice to any other provision in this Agreement, the ICB must comply with such financial processes as required by NHS England for the management, reporting and accounting of funds used for the purposes of the Delegated Functions.
- 9.2 The ICB acknowledges that it will receive funds from NHS England in respect of the Delegated Functions (the "Delegated Funds") and that these are in addition to the funds allocated to it within its Annual Allocation.
- 9.3 Subject to clause 9.4 and any provisions in the Schedules or Mandated Guidance, the ICB may use:
  - 9.3.1 its Annual Allocation and the Delegated Funds in the exercise of the Delegated Functions; and
  - 9.3.2 the Delegated Funds and its Annual Allocation in the exercise of the ICB's functions other than the Delegated Functions.
- 9.4 The ICB's expenditure on the Delegated Functions must be no less than that necessary to:
  - 9.4.1 ensure that NHS England is able to fulfil its functions, including without limitation the Reserved Functions, effectively and efficiently;
  - 9.4.2 meet all liabilities arising under or in connection with all Contracts allocated to the ICB in accordance with clauses 6.4 to 6.6 in so far as they relate to the Delegated Functions;

- 9.4.3 meet national commitments from time to time on expenditure on specific Delegated Functions including, without limitation, the Community Pharmacy Contractual Framework.
- 9.5 NHS England may increase or reduce the Delegated Funds in any Financial Year, by sending a notice to the ICB of such increase or decrease:
  - 9.5.1 in order to take into account any monthly adjustments or corrections to the Delegated Funds that NHS England considers appropriate, including without limitation adjustments following any changes to the Delegated Functions, changes in allocations, changes in Contracts, to implement Mandated Guidance under Clause 7.4 or otherwise;
  - 9.5.2 in order to comply with a change in the amount allocated to NHS England by the Secretary of State pursuant to section 223B of the NHS Act;
  - 9.5.3 to take into account any Losses of NHS England for which the ICB is required to indemnify NHS England under clause 15;
  - 9.5.4 to take into account any adjustments that NHS England considers appropriate (including without limitation in order to make corrections or otherwise to reflect notional budgets) to reflect funds transferred (or that should have been transferred) to the ICB in respect of the Delegated Functions and/or funds transferred (or that should have been transferred) to the ICB and in respect of which the ICB has management or administrative responsibility under Schedule 10 of this Agreement; or
  - 9.5.5 in order to ensure compliance by NHS England with its obligations under the NHS Act (including without limitation, Part 11 of the NHS Act) or any action taken or direction made by the Secretary of State in respect of NHS England under the NHS Act.
- 9.6 NHS England acknowledges that the intention of paragraph 9.5 is to reflect genuine corrections and adjustments to the Delegated Funds and may not be used to change the allocation of the Delegated Funds unless there are significant or exceptional circumstances that would require such corrections or adjustments.
- 9.7 The ICB acknowledges that it must comply with its statutory financial duties, including those under Part 11 of the NHS Act to the extent that these sections apply in relation to the receipt of the Delegated Funds.
- 9.8 NHS England may in respect of the Delegated Funds:
  - 9.8.1 notify the ICB regarding the required payment of sums by the ICB to NHS England in respect of charges referable to the valuation or disposal of assets and such conditions as to records, certificates or otherwise;
  - 9.8.2 by notice, require the ICB to take such action or step in respect of the Delegated Funds, in order to ensure compliance by NHS England of its duties or functions under the NHS (including without limitation, Part 11 of the NHS Act) or any action taken or direction made by the Secretary of State under the NHS Act.
- 9.9 The Schedules to this Agreement identify further financial provisions in respect of the exercise of the Delegated Functions including but not limited to SCHEDULE 5 (*Financial Provisions and Decision Making Limits*).
- 9.10 NHS England may issue Mandated Guidance in respect of the financial arrangements in respect of the Delegated Functions.

#### Payment and Transfer

- 9.11 NHS England will pay the Delegated Funds to the ICB using the revenue transfer process as used for the Annual Allocation or using such other process as notified to the ICB from time to time.
- 9.12 Without prejudice to any other obligation upon the ICB, the ICB agrees that it must use its resources for the purposes of the Delegated Functions in accordance with:
  - 9.12.1 the terms and conditions of this Agreement including any Mandated Guidance issued by NHS England from time to time in relation to the use of resources for the purposes of the Delegated Functions (including in relation to the form or contents of any accounts);
  - 9.12.2 the business rules as set out in NHS England's planning guidance or such other documents issued by NHS England from time to time;
  - 9.12.3 any Capital Investment Guidance; and
  - 9.12.4 the HM Treasury guidance *Managing Public Money* (dated September 2022)
- 9.13 Without prejudice to any other obligation upon the ICB, the ICB agrees that it must provide:
  - 9.13.1 all information, assistance and support to NHS England in relation to the audit and/or investigation (whether internal or external and whether under Law or otherwise) in relation to the use of or payment of resources for the purposes of the Delegated Functions and the discharge of those functions;
  - 9.13.2 such reports in relation to the expenditure on the Delegated Functions as set out in Mandated Guidance, the Schedules to this Agreement or as otherwise required by NHS England.

Administrative and/or Management Services

9.14 The provisions of SCHEDULE (*Administrative and Management Services*) in relation to Administrative and/or Management Services shall apply.

#### Pooled Funds

- 9.15 Subject to the provisions of this Agreement, the ICB may, for the purposes of exercising the Delegated Functions under this Agreement, establish and maintain a pooled fund in respect of any part of the Delegated Funds with:
  - 9.15.1 NHS England in accordance with sections 13V or 65Z6 of the NHS Act;
  - 9.15.2 one or more ICBs in accordance with section 65Z6 of the NHS Act as part of a Further Arrangement; or
  - 9.15.3 NHS England and one or more ICBs in accordance with section 13V of the NHS Act; and
  - 9.15.4 NHS England and one or more ICBs in accordance with section 65Z6 of the NHS Act.
- 9.16 At the date of this Agreement, details of the pooled funds (including any terms as to the governance and payments out of such pooled fund) of NHS England and the ICB are set out in the Local Terms.

#### 10. INFORMATION, PLANNING AND REPORTING

- 10.1 The ICB must provide to NHS England:
  - 10.1.1 all information or explanations in relation to the exercise of the Delegated Functions (including in relation to this Agreement), (and in such form) as requested by NHS England from time to time; and
  - 10.1.2 all such information (and in such form), that may be relevant to NHS England in relation to the exercise by NHS England of its other duties or functions including, without limitation, the Reserved Functions.
- 10.2 The provisions of this clause 10 are without prejudice to the ability of NHS England to exercise its other powers and duties in obtaining information from and assessing the performance of the ICB.

#### Forward Plan and Annual Report

- 10.3 Before the start of each Financial Year, the ICB must describe in its joint forward plan prepared in accordance with section 14Z52 of the NHS Act how it intends to exercise the Delegated Functions.
- 10.4 The ICB must report on its exercise of the Delegated Functions in its annual report prepared in accordance with section 14Z58 of the NHS Act.

#### Risk Register

10.5 The ICB must maintain a risk register in respect of its exercise of the Delegated Functions and periodically review its content. The risk register must follow such format as may be notified by NHS England to the ICB from time to time.

#### 11. FURTHER ARRANGEMENTS

- 11.1 The ICB must give due consideration to whether any of the Delegated Functions should be exercised collaboratively with other NHS bodies or Local Authorities including, without limitation, by means of arrangements under section 65Z5 and section 75 of the NHS Act.
- 11.2 The ICB may only make arrangements with another person (a "Sub-Delegate") concerning the exercise of the Delegated Functions ("Further Arrangements"), including without limitation arrangements under section 65Z5 and section 75 of the NHS Act, with the prior written approval of NHS England.
- 11.3 The approval of any Further Arrangements may:
  - 11.3.1 include approval of the terms of the proposed Further Arrangements; and
  - 11.3.2 require conditions to be met by the ICB and the Sub-Delegate in respect of that arrangement.
- 11.4 All Further Arrangements must be made in writing.
- 11.5 The ICB must not:
  - 11.5.1 terminate Further Arrangements; or
  - 11.5.2 make any material changes to the terms of Further Arrangements,
  - 11.5.3 without the prior written approval of NHS England.

- 11.6 If the ICB enters into a Further Arrangement it must ensure that the Sub-Delegate does not make onward arrangements for the exercise of any or all of the Delegated Functions without the prior written approval of NHS England.
- 11.7 The terms of this clause 11 do not prevent the ICB from making arrangements for assistance and support in the exercise of the Delegated Functions with any person, where such arrangements reserve the consideration and making of any decision in respect of a Delegated Function to the ICB.
- 11.8 NHS England requires the ICB to make arrangements for assistance and support in the exercise of the Delegated Functions with those persons described in the Schedules including, but not limited to SCHEDULE 6 (*Mandated Assistance and Support*) and with such other persons as NHS England may require from time to time.
- 11.9 Where Further Arrangements are made, and unless NHS England has otherwise given prior written agreement, any positive obligation or duty on the part of the ICB under this Agreement that is relevant to those Further Arrangements shall also require the ICB to ensure that all Sub-Delegates comply with that positive obligation or duty and support the ICB in doing so. In the same way, any negative duty or obligation on the part of the ICB under this Agreement that is relevant to Further Arrangement shall also require the ICB to ensure that all Sub-Delegates comply with that positive obligation on the part of the ICB under this Agreement that is relevant to Further Arrangement shall also require the ICB to ensure that all Sub-Delegates comply with that negative obligation or duty and support the ICB in doing so.

#### 12. STAFFING AND WORKFORCE

- 12.1 The Staffing Model in respect of each Delegated Function shall at the Effective Date of Delegation be as approved by the relevant National Moderation Panel.
- 12.2 Where the staffing arrangements include the deployment of NHS England Staff to the ICB for the purposes of carrying out the relevant Delegated Functions then the provisions of SCHEDULE 8 (*Deployment of NHS England Staff to the* ICB) shall apply.
- 12.3 The ICB must comply with any Mandated Guidance issued by NHS England from time to time in relation to the NHS England Staff.
- 12.4 For the avoidance of doubt, any breach by the ICB of the terms of this clause 12 (Staffing), including any breach of any Mandated Guidance issued in accordance with clause 12.3 above, will be a breach of the terms and conditions of this Agreement for the purposes of clauses 9.5 and 15.2.

#### 13. BREACH

- 13.1 If the ICB does not comply with the terms of this Agreement, then NHS England may:
  - 13.1.1 exercise its rights under this Agreement; and/or
  - 13.1.2 take such steps as it considers appropriate in the exercise of its other functions concerning the ICB.
- 13.2 Without prejudice to clause 13.1, if the ICB does not comply with the terms of this Agreement (including if the ICB exceeds its delegated authority under the Delegation), NHS England may (at its sole discretion):
  - 13.2.1 waive its rights in relation to such non-compliance in accordance with clause 13.3;
  - 13.2.2 ratify any decision in accordance with clause 6.8;
  - 13.2.3 substitute a decision in accordance with clause 6.9;

- 13.2.4 revoke the whole or part of the Delegation and terminate this Agreement in accordance with clause 26 *(Termination)* below;
- 13.2.5 exercise the Escalation Rights in accordance with clause 14 *(Escalation Rights)*; and/or
- 13.2.6 exercise its rights under common law.
- 13.3 NHS England may waive any non-compliance by the ICB with the terms of this Agreement provided that the ICB provides a written report to NHS England as required by clause 13.4 and, after considering the ICB's written report, NHS England is satisfied that the waiver is justified.
- 13.4 If:
  - 13.4.1 the ICB does not comply (or, based on the risk register maintained by the ICB in accordance with clause 10.5 or any other information available to it the ICB considers that it may not be able to comply) with this Agreement; or
  - 13.4.2 NHS England notifies the ICB that it considers the ICB has not complied, or may not be able to comply with, this Agreement,

then the ICB must provide a written report to NHS England within ten (10) Operational Days of the non-compliance (or the date on which the ICB considers that it may not be able to comply with this Agreement) or such notification pursuant to clause 13.4.2 setting out:

- 13.4.3 details of and reasons for the non-compliance (or likely non-compliance) with the Agreement and/or the Delegation; and
- 13.4.4 a plan for how the ICB proposes to remedy the non-compliance.

#### 14. **ESCALATION RIGHTS**

- 14.1 If the ICB does not comply with this Agreement, NHS England may exercise the following Escalation Rights:
  - 14.1.1 NHS England may require a suitably senior representative of the ICB to attend a review meeting within ten (10) days of NHS England becoming aware of the non-compliance; and
  - 14.1.2 NHS England may require the ICB to prepare an action plan and report within twenty (20) days of the review meeting (to include details of the non-compliance and a plan for how the ICB proposes to remedy the non-compliance).
- 14.2 Nothing in clause 14 (*Escalation Rights*) will affect NHS England's right to substitute a decision in accordance with clause 6.9, revoke the Delegation and/or terminate this Agreement in accordance with clause 26 (*Termination*) below.

#### 15. LIABILITY AND INDEMNITY

15.1 NHS England is liable in respect of any Losses arising in respect of NHS England's negligence, fraud, recklessness or deliberate breach in respect of the Delegated Functions and occurring after the Effective Date of Delegation and, if the ICB suffers any Losses in respect of such actions by NHS England, NHS England shall make such adjustments to the Annual Allocation (or other amounts payable to the ICB) in order to reflect any Losses suffered by the ICB (except to the extent that the ICB is liable for such Losses pursuant to clause 15.3).

- 15.2 For the avoidance of doubt, NHS England remains liable for a Claim relating to facts, events or circumstances concerning the Delegated Functions before the Effective Date of Delegation.
- 15.3 The ICB is liable to (and shall pay) NHS England for any Losses suffered by NHS England that result from or arise out of the ICB's negligence, fraud, recklessness or breach of the Delegation (including any actions that are taken that exceed the authority conferred by the Delegation) or this Agreement and, in respect of such Losses, NHS England may, at its discretion and without prejudice to any other rights, either require payment from the ICB or make such adjustments to the Delegated Funds pursuant to clause 9.5. The ICB shall not be liable to the extent that the Losses arose prior to the Effective Date of Delegation.
- 15.4 Each Party acknowledges and agrees that any rights acquired, or liabilities (including liabilities in tort) incurred, in respect of the exercise by the ICB of any Delegated Function are enforceable by or against the ICB only, in accordance with section 65Z5(6) of the NHS Act.
- 15.5 The ICB indemnifies NHS England and shall keep it indemnified on a continuing basis from and against any and all Losses which NHS England may incur by reason of any claim by any NHS England Staff:
  - 15.5.1 arising out of a breach of duty by the ICB (whether under common law, statute or otherwise) to the extent that such claim is not met by either the ICB's or NHS England's insurance or indemnity cover;
  - 15.5.2 under the Equality Act 2010 or Part V of the Employment Rights Act 1996 arising out of acts or omissions by the ICB (or any of its employees, directors or officers);
  - 15.5.3 arising from any acts or omissions by the ICB resulting in the termination of their employment, including any claim arising from any instruction by the ICB to NHS England to discipline or dismiss any person.
- 15.6 Each Party shall co-operate with the other in making all reasonable efforts to minimise any liabilities and Losses in connection with the employment of NHS England Staff in Delegated Functions.
- 15.7 Each Party will at all times take all reasonable steps to minimise and mitigate any Losses or other matters for which one Party is entitled to be indemnified by or to bring a claim against the other under this Agreement.

#### 16. CLAIMS AND LITIGATION

- 16.1 Nothing in this clause 16 (*Claims and Litigation*) shall be interpreted as affecting the reservation to NHS England of the Reserved Functions.
- 16.2 Except in the circumstances set out in clause **Error! Reference source not found.**16.5 and subject always to compliance with this clause 16 *(Claims and Litigation)*, the ICB shall be responsible for and shall retain the conduct of any Claim.
- 16.3 The ICB must:
  - 16.3.1 comply with any policy issued by NHS England from time to time in relation to the conduct of or avoidance of Claims and/or the pro-active management of Claims;
  - 16.3.2 if it receives any correspondence, issue of proceedings, claim document or other document concerning any Claim or potential Claim, immediately notify NHS England and send to NHS England all copies of such correspondence;

- 16.3.3 co-operate fully with NHS England in relation to such Claim and the conduct of such Claim;
- 16.3.4 provide, at its own cost, to NHS England all documentation and other correspondence that NHS England requires for the purposes of considering and/or resisting such Claim; and/or
- 16.3.5 at the request of NHS England, take such action or step or provide such assistance as may in NHS England's discretion be necessary or desirable having regard to the nature of the Claim and the existence of any time limit in relation to avoiding, disputing, defending, resisting, appealing, seeking a review or compromising such Claim or to comply with the requirements of the provider of an Indemnity Arrangement in relation to such Claim.
- 16.4 Subject to clauses 16.3 and 16.5 and SCHEDULE 5 (*Financial Provisions and Decision Making Limits*) the ICB is entitled to conduct the Claim in the manner it considers appropriate and is also entitled to pay or settle any Claim on such terms as it thinks fit.

#### NHS England Stepping into Claims

- 16.5 NHS England may, at any time following discussion with the ICB, send a notice to the ICB stating that NHS England will take over the conduct of the Claim and the ICB must immediately take all steps necessary to transfer the conduct of such Claim to NHS England unless and until NHS England transfers conduct back to the ICB. In such cases:
  - 16.5.1 NHS England shall be entitled to conduct the Claim in the manner it considers appropriate and is also entitled to pay or settle any Claim on such terms as it thinks fit, provided that if NHS England wishes to invoke clause 16.5.3 it agrees to seek the ICB's views on any proposal to pay or settle that Claim prior to finalising such payment or settlement; and
  - 16.5.2 the Delegation shall be treated as being revoked to the extent that and for so long as NHS England has assumed responsibility for exercising those of the Delegated Functions that are necessary for the purposes of having conduct of the Claim; and
  - 16.5.3 NHS England may, at its discretion and without prejudice to any other rights, either require payment from the ICB for such Claim Losses or make an adjustment to the Delegated Funds pursuant to clause 9.5.3 for the purposes of meeting any Claim Losses associated with that Claim.

#### Claim Losses

- 16.6 The ICB and NHS England shall notify each other within a reasonable time period of becoming aware of any Claim Losses.
- 16.7 The ICB acknowledges that NHS England will pay to the ICB the funds that are attributable to the Delegated Functions. Accordingly, the ICB acknowledges that it must pay any Claim Losses out of either the Delegated Funds or its Annual Allocation. NHS England may, in respect of any Claim Losses, at its discretion and without prejudice to any other rights, either require payment from the ICB for such Claim Losses or pursuant to clause 9.5.3 make such adjustments to the Delegated Funds to take into account the amount of any Claim Losses (other than any Claim Losses in respect of which NHS England has retained any funds, provisions or other resources to discharge such Claim Losses). For the avoidance of doubt, in circumstances where NHS England suffers any Claim Losses, then NHS England shall be entitled to recoup such Claim Losses pursuant to clause 9.5.3. If and to the extent that NHS England has retained any funds, provisions or other resources to discharge any Claim Losses pursuant to clause 9.5.3. If and to the extent that NHS England has retained any funds, provisions or other NHS England has retained any funds, provisions or other RNHS England has retained any funds, provisions or other RNHS England has retained any funds, provisions or other resources to discharge such Claim Losses pursuant to clause 9.5.3. If and to the extent that NHS England has retained any funds, provisions or other resources to discharge such Claim Losses, then NHS England may

either use such funds to discharge the Claim Loss or make an upward adjustment to the amounts paid to the ICB pursuant to clause 9.5.3.

#### 17. DATA PROTECTION, FREEDOM OF INFORMATION AND TRANSPARENCY

- 17.1 The Parties must ensure that all Personal Data processed by or on behalf of them in the course of carrying out the Delegated Functions and Reserved Functions is processed in accordance with the relevant Party's obligations under Data Protection Legislation and Data Guidance and the Parties must assist each other as necessary to enable each other to comply with these obligations.
- 17.2 The ICB must respond to any information governance breach in accordance with IG Guidance for Serious Incidents. If the ICB is required under Data Protection Legislation to notify the Information Commissioner's Office or a Data Subject of an information governance breach then as soon as reasonably practical and in any event on or before the first such notification is made the ICB must fully inform NHS England of the information governance breach. This clause does not require the ICB to provide NHS England with information which identifies any individual affected by the information governance breach where doing so would breach Data Protection Legislation.
- 17.3 Whether or not a Party is a Data Controller or Data Processor will be determined in accordance with Data Protection Legislation and any Data Guidance from a Regulatory or Supervisory Body. The Parties acknowledge that a Party may act as both a Data Controller and a Data Processor.
- 17.4 Each Party acknowledges that the other is a public authority for the purposes of the Freedom of Information Act 2000 ("**FOIA**") and the Environmental Information Regulations 2004 ("**EIR**").
- 17.5 Each Party may be statutorily required to disclose further information about the Agreement and the Relevant Information in response to a specific request under FOIA or EIR, in which case:
  - 17.5.1 each Party shall provide the other with all reasonable assistance and cooperation to enable them to comply with their obligations under FOIA or EIR;
  - 17.5.2 each Party shall consult the other regarding the possible application of exemptions in relation to the information requested; and
  - 17.5.3 subject only to clause 16 (*Claims and Litigation*), each Party acknowledges that the final decision as to the form or content of the response to any request is a matter for the Party to whom the request is addressed.
- 17.6 NHS England may, from time to time, issue a FOIA or EIR protocol or update a protocol previously issued relating to the dealing with and responding to FOIA or EIR requests in relation to the Delegated Functions. The ICB shall comply with such FOIA or EIR protocols.
- 17.7
- 17.8 SCHEDULE 3

#### **Reserved Functions**

#### 1. Introduction

- 1.1 In accordance with clause 8.4 of this Agreement, all functions of NHS England other than those defined as Delegated Functions are Reserved Functions.
- 1.2 This **Error! Reference source not found.** (*Reserved Functions*) sets out further provision regarding the carrying out of the Reserved Functions.

1.3 The ICB will work collaboratively with NHS England and will support and assist NHS England to carry out the Reserved Functions.

#### 2. Management of the national performers list

- 2.1 Subject to Paragraph 2.2, NHS England will continue to perform its functions under the National Health Service (Performers Lists) (England) Regulations 2013.
- 2.2 The ICB will carry out administrative tasks in respect of the Performers Lists as described at:
  - 2.2.1 Paragraph 9 of Part 2, Schedule 2A;
  - 2.2.2 Paragraph 9 of Part 2, Schedule 2B; and
  - 2.2.3 Paragraph 6 of Part 2, Schedule 2C.
- 2.3 NHS England's functions in relation to the management of the national performers list include:
  - 2.3.1 considering applications and decision-making in relation to inclusion on the national performers list, inclusion with conditions and refusals;
  - 2.3.2 identifying, managing and supporting primary care performers where concerns arise; and
  - 2.3.3 managing suspension, imposition of conditions and removal from the national performers list.
- 2.4 NHS England may hold local Performance Advisory Group ("PAG") meetings to consider all complaints or concerns that are reported to NHS England in relation to a named performer and NHS England will determine whether an initial investigation is to be carried out.
- 2.5 NHS England may notify the ICB of all relevant PAG meetings at least seven (7) days in advance of such meetings. NHS England may require a representative of the ICB to attend such meetings to discuss any performer concerns and/or quality issues that may impact on individual performer cases.
- 2.6 The ICB must develop a mechanism to ensure that all complaints regarding any named performer are escalated to the Local NHS England Team for review. The ICB will comply with any Mandated Guidance issued by NHS England in relation to the escalation of complaints about a named performer.

#### 3. Management of the revalidation and appraisal process

- 3.1 NHS England will continue to perform its functions under the Medical Profession (Responsible Officers) Regulations 2010 (as amended by the Medical Profession (Responsible Officers) (Amendment) Regulations 2013).
- 3.2 All functions in relation to GP appraisal and revalidation will remain the responsibility of NHS England, including:
  - 3.2.1 the funding of GP appraisers;
  - 3.2.2 quality assurance of the GP appraisal process; and
  - 3.2.3 the responsible officer network.
- 3.3 Funding to support the GP appraisal is incorporated within the global sum payment to Primary Medical Services Provider.

- 3.4 The ICB must not remove or restrict the payments made to Primary Medical Services Provider in respect of GP appraisal.
- 3.5 Appraisal arrangements in respect of all other primary care practitioner groups shall also be Reserved Functions.

#### 4. Administration of payments and related performers list management activities

- 4.1 NHS England reserves its functions in relation to the administration of payments to individual performers and related performers list management activities under the National Health Service (Performers Lists) (England) Regulations 2013 and other relevant legislation.
- 4.2 NHS England may continue to pay practitioners who are suspended from the national performers list in accordance with relevant determinations made by the Secretary of State.
- 4.3 For the avoidance of doubt, the ICB is responsible for any ad hoc or discretionary payments to Primary Medical Services Providers (including those under section 96 of the NHS Act) in accordance with **Error! Reference source not found.**A (*Delegated Functions*) Part 2 paragraphs 5.1 and 5.1 of this Agreement, including where such payments may be considered a consequence of actions taken under the National Health Service (Performers Lists) (England) Regulations 2013.

#### 5. Section 7A and Capital Expenditure Functions

- 5.1 NHS England retains the Section 7A Functions and will be responsible for taking decisions in relation to the Section 7A Functions.
- 5.2 In accordance with Schedule 10 Part 2, the ICB will provide certain management and/or administrative services to NHS England in relation to the Section 7A Functions.
- 5.3 NHS England retains the Capital Expenditure Functions and will be responsible for taking decisions in relation to the Capital Expenditure Functions.
- 5.4 In accordance with Schedule 10 Part 1, the ICB will provide certain management and/or administrative services to NHS England in relation to the Capital Expenditure Functions.

# 6. Such other ancillary activities that are necessary in order to exercise the Reserved Functions

- 6.1 NHS England will continue to comply with its obligations under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 6.2 The ICB must assist NHS England's controlled drug accountable officer ("CDAO") to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 6.3 The ICB must nominate a relevant senior individual within the ICB (the "ICB CD Lead") to liaise with and assist NHS England to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 6.4 The ICB CD Lead must, in relation to the Delegated Functions:
  - 6.4.1 on request provide NHS England's CDAO with all reasonable assistance in any investigation involving the Delegated Functions;
  - 6.4.2 report all complaints involving controlled drugs to NHS England's CDAO;
  - 6.4.3 report all incidents or other concerns involving the safe use and management of controlled drugs to NHS England's CDAO;

- 6.4.4 analyse the controlled drug prescribing data available; and
- 6.4.5 on request supply (or ensure organisations from whom the ICB commissions services involving the regular use of controlled drugs supply) periodic self-declaration and/or self-assessments to NHS England's CDAO.

#### 7. Reserved Functions – Primary Medical Services

- 7.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Medical Services Functions"):
  - 7.1.1 determining the outcomes expected from Primary Medical Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes and the Department of Health and Social Care mandate;
  - 7.1.2 designing and delivering national transformation programmes in support of national priorities;
  - 7.1.3 the negotiation and agreement of matters concerning General Medical Services contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary medical services nationally;
  - 7.1.4 the development of national standard Primary Medical Service contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Medical Services providers;
  - 7.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 7.1.6 the provision of nationally contracted services delivering digital, logistical and support services for Primary Medical Services in England (including but not limited to):
    - 7.1.6.1 Payments;
    - 7.1.6.2 Pensions;
    - 7.1.6.3 Patient Registration;
    - 7.1.6.4 Medical Records;
    - 7.1.6.5 Performer List;
    - 7.1.6.6 Supplies;
    - 7.1.6.7 Call and Recall for Cervical screening (CSAS); and
    - 7.1.6.8 Pharmacy Market Management.
- 7.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 8. **Reserved Functions – Primary Dental Services**

8.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):

- 8.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
- 8.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
- 8.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
- 8.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
- 8.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
  - 8.1.5.1 Payments;
  - 8.1.5.2 Pensions;
  - 8.1.5.3 Performer List; and
  - 8.1.5.4 Market Management.
- 8.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 9. Reserved Functions – Primary Ophthalmic Services

- 9.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Ophthalmic Functions"):
  - 9.1.1 the Primary Ophthalmic Services Contracts policy and associated documentation;
  - 9.1.2 the negotiation and agreement of matters concerning Primary Ophthalmic Services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Ophthalmic Services nationally; and
  - 9.1.3 the provision of nationally contracted services delivering digital, logistical and support services for Primary Ophthalmic Services in England (including but not limited to):
    - 9.1.3.1 Payments;
    - 9.1.3.2 Performers List;
    - 9.1.3.3 Market Management/Entry; and
    - 9.1.3.4 Contract management, assurance and post-payment verification.
- 9.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 10. Reserved Functions – Pharmaceutical Services and Local Pharmaceutical Services

- 10.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Pharmaceutical Functions"):
  - 10.1.1 publication of Pharmaceutical Lists;
  - 10.1.2 functions of NHS England as a determining authority in relation to pharmaceutical remuneration under Part 12 of the Pharmaceutical Regulations;
  - 10.1.3 functions in respect of lists of performers of pharmaceutical services and assistants, noting that as at the date of this Agreement regulations for the purposes of these functions have not been made;
  - 10.1.4 the negotiation and agreement of matters concerning NHS pharmaceutical services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Pharmaceutical Services nationally;
  - 10.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 10.1.6 administration of the pharmacist pre-registration training grant scheme.

#### 11. Reserved Functions – Primary Dental Services

- 11.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 11.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 11.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 11.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
  - 11.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 11.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
    - 11.1.5.1 Payments
    - 11.1.5.2 Pensions
    - 11.1.5.3 Performer List
    - 11.1.5.4 Market Management.

11.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 12. Reserved Functions - Prescribed Dental Services

- 12.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Prescribed Dental Services Functions"):
  - 12.1.1 determining the outcomes expected from Prescribed Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 12.1.2 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act the negotiation and agreement of matters concerning those contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 12.1.3 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the development of standard contracts and national contract variations and guidance;
  - 12.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 12.1.5 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the provision of nationally contracted services delivering digital, logistical and support services in England (including but not limited to):
    - 12.1.5.1 Payments
    - 12.1.5.2 Pensions
    - 12.1.5.3 Performer List
    - 12.1.5.4 Market Management.
- 12.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

17.9 **SCHEDULE 4** (*Further Information Governance and Sharing* Provisions) makes further provision about information sharing and information governance.

#### 18. **IT INTER-OPERABILITY**

- 18.1 NHS England and the ICB will work together to ensure that all relevant IT systems operated by NHS England and the ICB in respect of the Delegated Functions and the Reserved Functions are inter-operable and that data may be transferred between systems securely, easily and efficiently.
- 18.2 The Parties will use their respective reasonable endeavours to help develop initiatives to further this aim.

#### 19. CONFLICTS OF INTEREST AND TRANSPARENCY ON GIFTS AND HOSPITALITY

- 19.1 The ICB must and must ensure that, in delivering the Delegated Functions, all Staff comply with Law, with Managing Conflicts of Interest in the NHS and other Guidance, and with Good Practice, in relation to gifts, hospitality and other inducements and actual or potential conflicts of interest.
- 19.2 Without prejudice to the general obligations set out in clause 19.1, the ICB must maintain a register of interests in respect of all persons making decisions concerning the Delegated Functions. This register must be publicly available. For the purposes of this clause, the ICB may rely on an existing register of interests rather than creating a further register.

#### 20. **PROHIBITED ACTS AND COUNTER-FRAUD**

- 20.1 The ICB must not commit any Prohibited Act.
- 20.2 If the ICB or its Staff commits any Prohibited Act in relation to this Agreement with or without the knowledge of NHS England, NHS England will be entitled:
  - 20.2.1 to revoke the Delegation; and
  - 20.2.2 to recover from the ICB the amount or value of any gift, consideration or commission concerned; and
  - 20.2.3 to recover from the ICB any loss or expense sustained in consequence of the carrying out of the Prohibited Act.
- 20.3 The ICB must put in place and maintain appropriate arrangements, including without limitation Staff training, to address counter-fraud issues, having regard to any relevant Guidance (including from the NHS Counter Fraud Authority).
- 20.4 If requested by NHS England or the NHS Counter Fraud Authority, the ICB must allow a person duly authorised to act on behalf of the NHS Counter Fraud Authority or on behalf of NHS England to review, in line with the appropriate standards and counter-fraud arrangements put in place by the ICB.
- 20.5 The ICB must implement any reasonable modifications to its counter-fraud arrangements required by a person referred to in clause 20.4 in order to meet the appropriate standards within whatever time periods as that person may reasonably require.
- 20.6 The ICB must, on becoming aware of:
  - 20.6.1 any suspected or actual bribery, corruption or fraud involving public funds; or

- 20.6.2 any suspected or actual security incident or security breach involving Staff or involving NHS resources;
- 20.6.3 promptly report the matter to NHS England and to the NHS Counter Fraud Authority.
- 20.7 On the request of NHS England or NHS Counter Fraud Authority, the ICB must allow the NHS Counter Fraud Authority or any person appointed by NHS England, as soon as it is reasonably practicable and in any event not later than 5 Operational Days following the date of the request, access to:
  - 20.7.1 all property, premises, information (including records and data) owned or controlled by the ICB; and
  - 20.7.2 all Staff who may have information to provide;
  - 20.7.3 relevant to the detection and investigation of cases of bribery, fraud or corruption, or security incidents or security breaches directly or indirectly in connection with this Agreement.

#### 21. CONFIDENTIAL INFORMATION OF THE PARTIES

- 21.1 Except as this Agreement otherwise provides, Confidential Information is owned by the disclosing Party and the receiving Party has no right to use it.
- 21.2 Subject to clauses 21.3 to 21.5, the receiving Party agrees:
  - 21.2.1 to use the disclosing Party's Confidential Information only in connection with the receiving Party's performance under this Agreement;
  - 21.2.2 not to disclose the disclosing Party's Confidential Information to any third party or to use it to the detriment of the disclosing Party; and
  - 21.2.3 to maintain the confidentiality of the disclosing Party's Confidential Information.
- 21.3 The receiving Party may disclose the disclosing Party's Confidential Information:
  - 21.3.1 in connection with any Dispute Resolution;
  - 21.3.2 in connection with any litigation between the Parties;
  - 21.3.3 to comply with the Law;
  - 21.3.4 to any appropriate Regulatory or Supervisory Body;
  - 21.3.5 to its Staff, who in respect of that Confidential Information will be under a duty no less onerous than the Receiving Party's duty under clause 21.2;
  - 21.3.6 to NHS Bodies for the purposes of carrying out their functions;
  - 21.3.7 as permitted under or as may be required to give effect to clause 20 (*NHS Counter-Fraud*); and
  - 21.3.8 as permitted under any other express arrangement or other provision of this Agreement.
- 21.4 The obligations in clauses 21.1 and 21.2 will not apply to any Confidential Information which:
  - 21.4.1 is in or comes into the public domain other than by breach of this Agreement;

- 21.4.2 the receiving Party can show by its records was in its possession before it received it from the disclosing Party; or
- 21.4.3 the receiving Party can prove it obtained or was able to obtain from a source other than the disclosing Party without breaching any obligation of confidence.
- 21.5 This clause 21 does not prevent NHS England from making use of or disclosing any Confidential Information disclosed by the ICB where necessary for the purposes of exercising its functions in relation to the ICB.
- 21.6 The Parties acknowledge that damages would not be an adequate remedy for any breach of this clause 21 by the receiving Party, and in addition to any right to damages, the disclosing Party will be entitled to the remedies of injunction, specific performance and other equitable relief for any threatened or actual breach of this clause 21.
- 21.7 This clause 21 will survive the termination of this Agreement for any reason for a period of 5 years.
- 21.8 This clause 21 will not limit the application of the Public Interest Disclosure Act 1998 in any way whatsoever.

#### 22. INTELLECTUAL PROPERTY

- 22.1 The ICB grants to NHS England a fully paid-up, non-exclusive, perpetual licence to use the ICB Deliverables for the purposes of the exercise of its statutory and contractual functions.
- 22.2 NHS England grants the ICB a fully paid-up, non-exclusive licence to use the NHS England Deliverables for the purpose of performing this Agreement and the Delegated Functions.
- 22.3 The ICB must co-operate with NHS England to enable it to understand and adopt Best Practice (including the dissemination of Best Practice to other commissioners or providers of NHS services), and must supply such materials and information in relation to Best Practice as NHS England may reasonably request, and (to the extent that any IPR attaches to Best Practice), grants NHS England a fully paid-up, non-exclusive, perpetual licence for NHS England to use Best Practice IPR for the commissioning and provision of NHS services and to share any Best Practice IPR with other commissioners of NHS services (and other providers of NHS services) to enable those parties to adopt such Best Practice.

#### 23. NOTICES

- 23.1 Any notices given under this Agreement must be sent by e-mail to the other Party's address set out in the Particulars.
- 23.2 Notices by e-mail will be effective when sent in legible form, but only if, following transmission, the sender does not receive a non-delivery message.

#### 24. **DISPUTES**

- 24.1 This clause does not affect NHS England's right to exercise its functions for the purposes of assessing and addressing the performance of the ICB.
- 24.2 If a Dispute arises out of or in connection with this Agreement then the Parties must follow the procedure set out in this clause:
  - 24.2.1 either Party must give to the other written notice of the Dispute, setting out its nature and full particulars ("**Dispute Notice**"), together with relevant

supporting documents. On service of the Dispute Notice, the Agreement Representatives must attempt in good faith to resolve the Dispute;

- 24.2.2 if the Agreement Representatives are, for any reason, unable to resolve the Dispute within twenty (20) days of service of the Dispute Notice, the Dispute must be referred to the Chief Executive Officer (or equivalent person) of the ICB and a director of or other person nominated by NHS England (and who has authority from NHS England to settle the Dispute) who must attempt in good faith to resolve it; and
- 24.2.3 if the people referred to in clause 24.2.2 are for any reason unable to resolve the Dispute within twenty (20) days of it being referred to them, the Parties may attempt to settle it by mediation in accordance with the CEDR model mediation procedure. Unless otherwise agreed between the Parties, the mediator must be nominated by CEDR Solve. To initiate the mediation, a Party must serve notice in writing ('Alternative Dispute Resolution' (**ADR**) **notice**) to the other Party to the Dispute, requesting a mediation. A copy of the ADR notice should be sent to CEDR Solve. The mediation will start not later than ten (10) days after the date of the ADR notice.
- 24.3 If the Dispute is not resolved within thirty (30) days after service of the ADR notice, or either Party fails to participate or to continue to participate in the mediation before the expiration of the period of thirty (30) days, or the mediation terminates before the expiration of the period of thirty (30) days, the Dispute must be referred to the Secretary of State, who shall resolve the matter and whose decision shall be binding upon the Parties.

#### 25. VARIATIONS

- 25.1 The Parties acknowledge that the scope of the Delegated Functions may be reviewed and amended from time to time including by revoking this Agreement and making alternative arrangements.
- 25.2 NHS England may notify the ICB of a Variation Proposal in respect of this Agreement.
- 25.3 The Variation Proposal will set out the variation proposed and the date on which NHS England requires the variation to take effect.
- 25.4 The ICB must respond to a Variation Proposal within thirty (30) Operational Days following the date that it is issued by serving notice on NHS England confirming either:
  - 25.4.1 that it accepts the Variation Proposal; or
  - 25.4.2 that it refuses to accept the Variation Proposal, and sets out reasonable grounds for that refusal.
- 25.5 If the ICB accepts the Variation Proposal, the ICB agrees (without delay) to take all necessary steps (including executing a variation agreement) in order to give effect to any variation by the date on which the proposed variation will take effect as set out in the Variation Proposal.
- 25.6 If the ICB refuses to accept the Variation Proposal or to take such steps as are required to give effect to the variation, NHS England may terminate this Agreement in respect of some or all of the Delegated Functions.
- 25.7 The provisions of this clause 25 are without prejudice to the ability of NHS England to issue Contractual Notices which have the effect of varying this Agreement.

#### 26. **TERMINATION**

26.1 The ICB may:

- 26.1.1 notify NHS England that it requires NHS England to revoke the Delegation; and
- 26.1.2 terminate this Agreement;

with effect from the end of 31 March in any calendar year, provided that:

- 26.1.3 on or before 30 September of the previous calendar year, the ICB sends written notice to NHS England of its requirement that NHS England revoke the Delegation and intention to terminate this Agreement; and
- 26.1.4 the ICB meets with NHS England within ten (10) Operational Days of NHS England receiving the notice set out at clause 26.1.3 above to discuss arrangements for termination and transition of the Delegated Functions to a successor commissioner,

in which case NHS England shall revoke the Delegation and this Agreement shall terminate with effect from the end of 31 March in the next calendar year.

- 26.2 NHS England may revoke the Delegation in whole or in part with effect from 23.59 hours on 31 March in any year, provided that it gives notice to the ICB of its intention to terminate the Delegation on or before 30 September in the year prior to the year in which the Delegation will terminate, and in which case clause 26.4 will apply.
- 26.3 The Delegation may be revoked in whole or in part, and this Agreement may be terminated by NHS England at any time, including in (but not limited to) the following circumstances:
  - 26.3.1 the ICB acts outside of the scope of its delegated authority;
  - 26.3.2 the ICB fails to perform any material obligation of the ICB owed to NHS England under this Agreement;
  - 26.3.3 the ICB persistently commits non-material breaches of this Agreement;
  - 26.3.4 NHS England is satisfied that its intervention powers under section 14Z61 of the NHS Act apply;
  - 26.3.5 to give effect to legislative changes, including conferral of any of the Delegated or Reserved Functions on the ICB;
  - 26.3.6 failure to agree to a variation in accordance with clause 25 (Variations);
  - 26.3.7 NHS England and the ICB agree in writing that the Delegation shall be revoked and this Agreement shall terminate on such date as is agreed; and/or
  - 26.3.8 the ICB merges with another ICB or other body.
- 26.4 This Agreement will terminate upon revocation or termination of the full Delegation (including revocation and termination in accordance with this clause 26 (*Termination*)) except that the provisions referred to at clause 28 (*Provisions Surviving Termination*) will continue in full force and effect.
- 26.5 Without prejudice to clause 13.3 and to avoid doubt, NHS England may waive any right to terminate this Agreement under this clause 26.5 *(Termination)*. Any such waiver is only effective if given in writing and shall not be deemed a waiver of any subsequent right or remedy.
- 26.6 As an alternative to termination of the Agreement in respect of all the Delegated Functions, NHS England may alternatively terminate the Agreement in respect of

specified Delegated Functions (or aspects of such Delegated Functions) only, in which case this Agreement shall otherwise remain in effect.

#### 27. CONSEQUENCE OF TERMINATION

- 27.1 Termination of this Agreement, or termination of the ICB's exercise of any of the Delegated Functions, will not affect any rights or liabilities of the Parties that have accrued before the date of that termination or which later accrue.
- 27.2 Subject to clause 27.4, on or pending termination of this Agreement or termination of the ICB's exercise of any of the Delegated Functions, NHS England, the ICB and if appropriate any successor delegate will:
  - 27.2.1 agree a plan for the transition of the Delegated Functions from the ICB to the successor delegate, including details of the transition, the Parties' responsibilities in relation to the transition, the Parties' arrangements in respect of those staff engaged in the Delegated Functions and the date on which the successor delegate will take responsibility for the Delegated Functions;
  - 27.2.2 implement and comply with their respective obligations under the plan for transition agreed in accordance with clause 27.2.1 above; and
  - 27.2.3 act with a view to minimising any inconvenience or disruption to the commissioning of healthcare in the Area.
- 27.3 For a reasonable period before and after termination of this Agreement or termination of the ICB's exercise of any of the Delegated Functions, the ICB must:
  - 27.3.1 co-operate with NHS England and any successor delegate in order to ensure continuity and a smooth transfer of the Delegated Functions; and
  - 27.3.2 at the reasonable request of NHS England:
    - (a) promptly provide all reasonable assistance and information to the extent necessary to effect an orderly assumption of the Delegated Functions by a successor delegate;
    - (b) deliver to NHS England all materials and documents used by the ICB in the exercise of any of the Delegated Functions; and
  - 27.3.3 use all reasonable efforts to obtain the consent of third parties to the assignment, novation or termination of existing contracts between the ICB and any third party which relate to or are associated with the Delegated Functions.
- 27.4 Where any or all of the Delegated Functions or Reserved Functions are to be directly conferred on the ICB, the Parties will co-operate with a view to ensuring continuity and a smooth transfer to the ICB.

#### 28. **PROVISIONS SURVIVING TERMINATION**

- 28.1 Any rights, duties or obligations of any of the Parties which are expressed to survive, including those referred to in clause 28.2, or which otherwise by necessary implication survive the termination for any reason of this Agreement, together with all indemnities, will continue after termination, subject to any limitations of time expressed in this Agreement.
- 28.2 The surviving provisions include the following clauses together with such other provisions as are required to interpret and give effect to them:

- 28.2.1 Clause 9 (Finance);
- 28.2.2 Clause 12 (Staffing and Workforce);
- 28.2.3 Clause 15 (Liability and Indemnity);
- 28.2.4 Clause 16 (Claims and Litigation);
- 28.2.5 Clause 17 (Data Protection, Freedom of Information and Transparency);
- 28.2.6 Clause 24 (*Disputes*);
- 28.2.7 Clause 26 (Termination);
- 28.2.8
- 28.2.9 SCHEDULE 3

#### **Reserved Functions**

#### 13. Introduction

- 13.1 In accordance with clause 8.4 of this Agreement, all functions of NHS England other than those defined as Delegated Functions are Reserved Functions.
- 13.2 This **Error! Reference source not found.** (*Reserved Functions*) sets out further provision regarding the carrying out of the Reserved Functions.
- 13.3 The ICB will work collaboratively with NHS England and will support and assist NHS England to carry out the Reserved Functions.

#### 14. Management of the national performers list

- 14.1 Subject to Paragraph 2.2, NHS England will continue to perform its functions under the National Health Service (Performers Lists) (England) Regulations 2013.
- 14.2 The ICB will carry out administrative tasks in respect of the Performers Lists as described at:
  - 14.2.1 Paragraph 9 of Part 2, Schedule 2A;
  - 14.2.2 Paragraph 9 of Part 2, Schedule 2B; and
  - 14.2.3 Paragraph 6 of Part 2, Schedule 2C.
- 14.3 NHS England's functions in relation to the management of the national performers list include:
  - 14.3.1 considering applications and decision-making in relation to inclusion on the national performers list, inclusion with conditions and refusals;
  - 14.3.2 identifying, managing and supporting primary care performers where concerns arise; and
  - 14.3.3 managing suspension, imposition of conditions and removal from the national performers list.
- 14.4 NHS England may hold local Performance Advisory Group ("PAG") meetings to consider all complaints or concerns that are reported to NHS England in relation to a named performer and NHS England will determine whether an initial investigation is to be carried out.

- 14.5 NHS England may notify the ICB of all relevant PAG meetings at least seven (7) days in advance of such meetings. NHS England may require a representative of the ICB to attend such meetings to discuss any performer concerns and/or quality issues that may impact on individual performer cases.
- 14.6 The ICB must develop a mechanism to ensure that all complaints regarding any named performer are escalated to the Local NHS England Team for review. The ICB will comply with any Mandated Guidance issued by NHS England in relation to the escalation of complaints about a named performer.

#### 15. Management of the revalidation and appraisal process

- 15.1 NHS England will continue to perform its functions under the Medical Profession (Responsible Officers) Regulations 2010 (as amended by the Medical Profession (Responsible Officers) (Amendment) Regulations 2013).
- 15.2 All functions in relation to GP appraisal and revalidation will remain the responsibility of NHS England, including:
  - 15.2.1 the funding of GP appraisers;
  - 15.2.2 quality assurance of the GP appraisal process; and
  - 15.2.3 the responsible officer network.
- 15.3 Funding to support the GP appraisal is incorporated within the global sum payment to Primary Medical Services Provider.
- 15.4 The ICB must not remove or restrict the payments made to Primary Medical Services Provider in respect of GP appraisal.
- 15.5 Appraisal arrangements in respect of all other primary care practitioner groups shall also be Reserved Functions.

#### 16. Administration of payments and related performers list management activities

- 16.1 NHS England reserves its functions in relation to the administration of payments to individual performers and related performers list management activities under the National Health Service (Performers Lists) (England) Regulations 2013 and other relevant legislation.
- 16.2 NHS England may continue to pay practitioners who are suspended from the national performers list in accordance with relevant determinations made by the Secretary of State.
- 16.3 For the avoidance of doubt, the ICB is responsible for any ad hoc or discretionary payments to Primary Medical Services Providers (including those under section 96 of the NHS Act) in accordance with **Error! Reference source not found.**A (*Delegated Functions*) Part 2 paragraphs 5.1 and 5.1 of this Agreement, including where such payments may be considered a consequence of actions taken under the National Health Service (Performers Lists) (England) Regulations 2013.

#### 17. Section 7A and Capital Expenditure Functions

- 17.1 NHS England retains the Section 7A Functions and will be responsible for taking decisions in relation to the Section 7A Functions.
- 17.2 In accordance with Schedule 10 Part 2, the ICB will provide certain management and/or administrative services to NHS England in relation to the Section 7A Functions.

- 17.3 NHS England retains the Capital Expenditure Functions and will be responsible for taking decisions in relation to the Capital Expenditure Functions.
- 17.4 In accordance with Schedule 10 Part 1, the ICB will provide certain management and/or administrative services to NHS England in relation to the Capital Expenditure Functions.

# 18. Such other ancillary activities that are necessary in order to exercise the Reserved Functions

- 18.1 NHS England will continue to comply with its obligations under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 18.2 The ICB must assist NHS England's controlled drug accountable officer ("CDAO") to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 18.3 The ICB must nominate a relevant senior individual within the ICB (the "ICB CD Lead") to liaise with and assist NHS England to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 18.4 The ICB CD Lead must, in relation to the Delegated Functions:
  - 18.4.1 on request provide NHS England's CDAO with all reasonable assistance in any investigation involving the Delegated Functions;
  - 18.4.2 report all complaints involving controlled drugs to NHS England's CDAO;
  - 18.4.3 report all incidents or other concerns involving the safe use and management of controlled drugs to NHS England's CDAO;
  - 18.4.4 analyse the controlled drug prescribing data available; and
  - 18.4.5 on request supply (or ensure organisations from whom the ICB commissions services involving the regular use of controlled drugs supply) periodic self-declaration and/or self-assessments to NHS England's CDAO.

#### **19.** Reserved Functions – Primary Medical Services

- 19.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Medical Services Functions"):
  - 19.1.1 determining the outcomes expected from Primary Medical Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes and the Department of Health and Social Care mandate;
  - 19.1.2 designing and delivering national transformation programmes in support of national priorities;
  - 19.1.3 the negotiation and agreement of matters concerning General Medical Services contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary medical services nationally;
  - 19.1.4 the development of national standard Primary Medical Service contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Medical Services providers;
  - 19.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
- 19.1.6 the provision of nationally contracted services delivering digital, logistical and support services for Primary Medical Services in England (including but not limited to):
  - 19.1.6.1 Payments;
  - 19.1.6.2 Pensions;
  - 19.1.6.3 Patient Registration;
  - 19.1.6.4 Medical Records;
  - 19.1.6.5 Performer List;
  - 19.1.6.6 Supplies;
  - 19.1.6.7 Call and Recall for Cervical screening (CSAS); and
  - 19.1.6.8 Pharmacy Market Management.
- 19.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

### 20. Reserved Functions – Primary Dental Services

- 20.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 20.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 20.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 20.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
  - 20.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 20.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
    - 20.1.5.1 Payments;
    - 20.1.5.2 Pensions;
    - 20.1.5.3 Performer List; and
    - 20.1.5.4 Market Management.
- 20.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 21. Reserved Functions – Primary Ophthalmic Services

- 21.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Ophthalmic Functions"):
  - 21.1.1 the Primary Ophthalmic Services Contracts policy and associated documentation;
  - 21.1.2 the negotiation and agreement of matters concerning Primary Ophthalmic Services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Ophthalmic Services nationally; and
  - 21.1.3 the provision of nationally contracted services delivering digital, logistical and support services for Primary Ophthalmic Services in England (including but not limited to):
    - 21.1.3.1 Payments;
    - 21.1.3.2 Performers List;
    - 21.1.3.3 Market Management/Entry; and
    - 21.1.3.4 Contract management, assurance and post-payment verification.
- 21.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 22. Reserved Functions – Pharmaceutical Services and Local Pharmaceutical Services

- 22.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Pharmaceutical Functions"):
  - 22.1.1 publication of Pharmaceutical Lists;
  - 22.1.2 functions of NHS England as a determining authority in relation to pharmaceutical remuneration under Part 12 of the Pharmaceutical Regulations;
  - 22.1.3 functions in respect of lists of performers of pharmaceutical services and assistants, noting that as at the date of this Agreement regulations for the purposes of these functions have not been made;
  - 22.1.4 the negotiation and agreement of matters concerning NHS pharmaceutical services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Pharmaceutical Services nationally;
  - 22.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 22.1.6 administration of the pharmacist pre-registration training grant scheme.

#### 23. Reserved Functions – Primary Dental Services

- 23.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 23.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national

transformation programmes in line with any applicable commissioning policies and guidance;

- 23.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
- 23.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
- 23.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
- 23.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
  - 23.1.5.1 Payments
  - 23.1.5.2 Pensions
  - 23.1.5.3 Performer List
  - 23.1.5.4 Market Management.
- 23.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 24. Reserved Functions - Prescribed Dental Services

- 24.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Prescribed Dental Services Functions"):
  - 24.1.1 determining the outcomes expected from Prescribed Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 24.1.2 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act the negotiation and agreement of matters concerning those contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 24.1.3 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the development of standard contracts and national contract variations and guidance;
  - 24.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 24.1.5 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the provision of nationally contracted services delivering digital, logistical and support services in England (including but not limited to):
    - 24.1.5.1 Payments

- 24.1.5.2 Pensions
- 24.1.5.3 Performer List
- 24.1.5.4 Market Management.
- 24.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 28.2.10 SCHEDULE 4 (Further Information Governance and Sharing Provisions).

# 29. **COSTS**

29.1 Each Party is responsible for paying its own costs and expenses incurred in connection with the negotiation, preparation and execution of this Agreement.

#### 30. SEVERABILITY

30.1 If any provision or part of any provision of this Agreement is declared invalid or otherwise unenforceable, that provision or part of the provision as applicable will be severed from this Agreement. This will not affect the validity and/or enforceability of the remaining part of that provision or of other provisions.

# 31. GENERAL

- 31.1 Nothing in this Agreement will create a partnership or joint venture or relationship of principal and agent between NHS England and the ICB.
- 31.2 A delay or failure to exercise any right or remedy in whole or in part shall not waive that or any other right or remedy, nor shall it prevent or restrict the further exercise of that or any other right or remedy.
- 31.3 This Agreement does not give rise to any rights under the Contracts (Rights of Third Parties) Act 1999 to enforce any term of this Agreement.

# SCHEDULE 1

#### **Definitions and Interpretation**

- 1. The headings in this Agreement will not affect its interpretation.
- 2. Reference to any statute or statutory provision, Law, Guidance, Mandated Guidance or Data Guidance, includes a reference to that statute or statutory provision, Law, Guidance, Mandated Guidance or Data Guidance as from time to time updated, amended, extended, supplemented, re-enacted or replaced in whole or in part.
- 3. Reference to a statutory provision includes any subordinate legislation made from time to time under that provision.
- 4. References to clauses and schedules are to the clauses and schedules of this Agreement, unless expressly stated otherwise.
- 5. References to any body, organisation or office include reference to its applicable successor from time to time.
- 6. Any references to this Agreement or any other documents or resources includes reference to this Agreement or those other documents or resources as varied, amended, supplemented, extended, restated and/or replaced from time to time and any reference to a website address for a resource includes reference to any replacement website address for that resource.
- 7. Use of the singular includes the plural and vice versa.
- 8. Use of the masculine includes the feminine and all other genders.
- 9. Use of the term "including" or "includes" will be interpreted as being without limitation.
- 10. The following words and phrases have the following meanings:

Additional Pharmaceutical Services	Services provided in accordance with a direction under section 127 of the NHS Act (also referred to as advanced services and enhanced services in the Pharmaceutical Regulations);
Agreement	means this agreement between NHS England and the ICB comprising the Particulars, the Terms and Conditions, the Schedules and the Mandated Guidance;
Agreement Representatives	means the ICB Representative and the NHS England Representative as set out in the Particulars;
Annual Allocation	means the funds allocated to the ICB annually under section 223G of the NHS Act
APMS Contract	means an agreement or contract for the provision of primary medical services made under section 83(2) of the NHS Act (including any arrangements which are made in reliance on a combination of that section and other powers to arrange for primary medical services);
Area	means the area described in the Particulars;

- Assigned Staff means those NHS England staff as agreed between NHS England and the ICB from time to time;
- **Best Practice** means any methodologies, pathway designs and processes relating to this Agreement or the Delegated Functions developed by the ICB or its Staff for the purposes of delivering the Delegated Functions and which are capable of wider use in the delivery of healthcare services for the purposes of the NHS, but not including inventions that are capable of patent protection and for which patent protection is being sought or has been obtained, registered designs, or copyright in software;
- **Caldicott Principles** means the patient confidentiality principles set out in the report of the Caldicott Committee (December 1997 as amended by the 2013 Report, The Information Governance Review "*To Share or Not to Share?*") and now included in the NHS Confidentiality Code of Practice, as may be amended from time to time;
- Capitalshall have the meaning set out in the Capital Investment<br/>Guidance or such other replacement Mandated Guidance as<br/>issued by NHS England from time to time;
- Capital Expendituremeans those functions of NHS England in relation to the use<br/>and expenditure of Capital funds (but excluding the Premises<br/>Costs Directions Functions);
- Capital Investmentmeans any Mandated Guidance issued by NHS England from<br/>time to time in relation to the development, assurance and<br/>approvals process for proposals in relation to:
  - the expenditure of Capital, or investment in property, infrastructure or information and technology; and
  - the revenue consequences for commissioners or third parties making such investment;
- CEDR means the Centre for Effective Dispute Resolution;
- Claims means, for or in relation to the Delegated Functions (a) any litigation or administrative, mediation, arbitration or other proceedings, or any claims, actions or hearings before any court, tribunal or the Secretary of State, any governmental, regulatory or similar body, or any department, board or agency or (b) any dispute with, or any investigation, inquiry or enforcement proceedings by, any governmental, regulatory or similar body or agency;
- Claim Losses means all Losses arising in relation to any Claim;
- **Combined Authority** means a body of that name established under the provisions of the Local Democracy, Economic Development and Construction Act 2009;
- **Community Dental Services** means specialised dental services commissioned for patients who are unable to access treatment from Primary Dental

Services due to a disability or medical condition, being a form of Prescribed Dental Service;

- **Community Pharmacy Contractual Framework** means the Community Pharmacy Contractual Framework as published by the Department of Health and Social Care from time to time;
- **Complaints Regulations** means the Local Authority Social Services and National Health Service Complaints (England) Regulations 2009/309;
- **Confidential Information** means any information or data in whatever form disclosed, which by its nature is confidential or which the disclosing Party acting reasonably states in writing to the receiving Party is to be regarded as confidential, or which the disclosing Party acting reasonably has marked 'confidential' (including, financial information, strategy documents, tenders, employee confidential information, development or workforce plans and information, and information relating to services) but which is not information which is disclosed in response to a FOIA request, or information which is published as a result of NHS England or government policy in relation to transparency;
- **Contracts** Means any Prescribed Dental Services Contract, Primary Care Contract or Arrangement or other contract or arrangement in respect of the commissioning of any other Delegated Services;
- **Contractual Notice** means a contractual notice issued by NHS England to the ICB, or some or all ICBs (as the case may be), from time to time and relating to the allocation of contracts for the purposes of the Delegated Functions and/or the manner in which the Delegated Functions should be exercised by the ICB;
- **CQC** means the Care Quality Commission;
- **Data Controller** shall have the same meaning as set out in the UK GDPR;
- **Data Guidance** means any applicable guidance, guidelines, direction or determination, framework, code of practice, standard or requirement regarding information governance, confidentiality, privacy or compliance with Data Protection Legislation to the extent published and publicly available or their existence or contents have been notified to the ICB by NHS England and/or any relevant Regulatory or Supervisory Body. This includes but is not limited to guidance issued by NHS Digital, the National Data Guardian for Health & Care, the Department of Health and Social Care, NHS England, the Health Research Authority, the UK Health Security Agency and the Information Commissioner;
- **Data Processor** shall have the same meaning as set out in the UK GDPR;
- **Data Protection Legislation** means the UK GDPR, the Data Protection Act 2018 and all applicable Law concerning privacy, confidentiality or the processing of personal data including but not limited to the Human Rights Act 1998, the Health and Social Care (Safety

and Quality) Act 2015, the common law duty of confidentiality and the Privacy and Electronic Communications (EC Directive) Regulations 2003;

- **Data Sharing Agreement** means a data sharing agreement which should be in substantially the same form as the Data Sharing Agreement template shared by NHS England in respect of this Agreement;
- **Data Subject** shall have the same meaning as set out in the UK GDPR;
- **Delegated Functions** means the statutory functions delegated by NHS England to the ICB under the Delegation and as set out in detail in this Agreement;
- **Delegated Funds** means the funds defined in Clause 9.2;
- **Delegated Services** Means the services commissioned in exercise of the Delegated Functions;
- Delegationmeans the delegation of the Delegated Functions from NHS<br/>England to the ICB as described at clause 6.1;
- Dental Care Services means:
  - (i) Primary Dental Services; and
  - (ii) the Prescribed Dental Services;

**Dental Services Contract** means:

- (i) a GDS Contract;
- (ii) a PDS Agreement (except for any Community Dental Services PDS Agreement, which constitutes a Prescribed Dental Services Contract); and
- (iii) any other contract for the provision of health services made pursuant to NHS England's functions under Part 5 of the NHS Act;

in each case as amended or replaced from time to time and including all ancillary or related agreements directly relating to the subject matter of such agreements, contracts or arrangements;

- **Dental Services Provider** means a natural or legal person who holds a Dental Services Contract;
- Direct Commissioning
   means the webpage maintained by NHS England at

   Guidance Webpage
   https://www.england.nhs.uk/commissioning/howcommissioning-is-changing/;
- Disputea dispute, conflict or other disagreement between the Parties<br/>arising out of or in connection with this Agreement;
- **Effective Date of Delegation** means the Effective Date of Delegation as set out in the Particulars;

EIR	means the Environmental Information Regulations 2004					
Enhanced Services	means the nationally defined enhanced services, as set out in such directions made by the Secretary of State pursuant to his powers contained in sections 98A, 114A, 125A and 168A of the NHS Act as are in force from time to time, or which may be prescribed by NHS England under its Reserved Functions, and any other enhanced services schemes locally developed by the ICB in the exercise of its Delegated Functions (and excluding, for the avoidance of doubt, any enhanced services arranged or provided pursuant to the Section 7A Functions);					
Escalation Rights	means the escalation rights as defined in clause 14 <i>(Escalation Rights)</i> ;					
Financial Year	shall bear the same meaning as in section 275 of the NHS Act;					
FOIA	the Freedom of Information Act 2000;					
Further Arrangements	means arrangements for the exercise of Delegated Functions as defined at clause 11.2;					
GDS Contract	means a General Dental Services contract made under section 100 of the NHS Act;					
GMS Contract	means a General Medical Services contract made under section 84(1) of the NHS Act;					
Good Practice	means using standards, practices, methods and procedures conforming to the law, reflecting up-to-date published evidence and exercising that degree of skill and care, diligence, prudence and foresight which would reasonably and ordinarily be expected from a skilled, efficient and experienced commissioner;					
Guidance	means any applicable guidance, guidelines, direction or determination, framework, code of practice, standard or requirement to which the ICB has a duty to have regard (and whether specifically mentioned in this Agreement or not), to the extent that the same are published and publicly available or the existence or contents of them have been notified to the ICB by any relevant Regulatory or Supervisory Body but excluding Mandated Guidance;					
HSCA	means the Health and Social Care Act 2012;					
ICB	means an Integrated Care Board established pursuant to section 14Z25 of the NHS Act and named in the Particulars;					
ICB Deliverables	all documents, products and materials developed by the ICB or its Staff in relation to this Agreement and the Delegated Functions in any form and required to be submitted to NHS England under this Agreement, including data, reports, policies, plans and specifications;					

- IG Guidance for Serious Incidents IG Guidance for Serious Incidents NHS Digital's Checklist Guidance for Information Governance Serious Incidents Requiring Investigation June 2013, available at: https://digital.nhs.uk/data-and-information/looking-afterinformation/data-security-and-informationgovernance/datasecurity-and-protection-toolkit
- **Indemnity Arrangement** means either: (i) a policy of insurance; (ii) an arrangement made for the purposes of indemnifying a person or organisation; or (iii) a combination of (i) and (ii);
- Information Law the UK GDPR, the Data Protection Act 2018, regulations and guidance made under section 13S and section 251 of the NHS Act; guidance made or given under sections 263 and 265 of the HSCA; the Freedom of Information Act 2000; the common law duty of confidentiality; the Human Rights Act 1998 and all other applicable laws and regulations relating to processing of Personal Data and privacy;
- IPR means inventions, copyright, patents, database right, trademarks, designs and confidential know-how and any similar rights anywhere in the world whether registered or not, including applications and the right to apply for any such rights;
- Law means any applicable law, statute, rule, bye-law, regulation, direction, order, regulatory policy, guidance or code, rule of court or directives or requirements of any regulatory body, delegated or subordinate legislation or notice of any regulatory body (including any Regulatory or Supervisory Body);
- Local Authority means a county council in England, a Combined Authority, a district council in England, a London borough council, the Common Council of the City of London or the Council of the Isles of Scilly;
- **Local Incentive Schemes** means an incentive scheme developed by the ICB in the exercise of its Delegated Functions to extend the range or quality of essential and additional services provided under a Primary Medical Services Contract and support national frameworks in order to meet differing local population needs;
- Local Pharmaceutical
   means

   Services Contract
   a contract entered into pursuant to section 134 of the NHS Act; or
  - a contract entered into pursuant to Paragraph 1 of Schedule 12 to the NHS Act;
- Local Terms means the terms set out in SCHEDULE 7 (Local Terms) and/or such other Schedule or part thereof as designated as Local Terms;
- Losses means all damages, loss, liabilities, claims, actions, costs, expenses (including the cost of legal and/or professional

services) proceedings, demands and charges whether arising under statute, contract or common law;

- Managing Conflicts ofthe NHS publication by that name available at:Interest in the NHShttps://www.england.nhs.uk/about/board-<br/>meetings/committees/coi/
- Mandated Guidance means any protocol, policy, guidance, guidelines, framework or manual relating to the exercise of the Delegated Functions and issued by NHS England to the ICB as Mandated Guidance from time to time, in accordance with clause 7.2 which at the Effective Date of Delegation shall include the Mandated Guidance set out in the Schedules;
- **National Moderation Panel** Means the NHS England panel in respect of the relevant Delegated Function that will have the delegated authority to approve the ICB arrangements in respect of a Delegated Function;
- Need to Knowhas the meaning set out in paragraph 6.2 of Error!Reference source not found.(Further Information<br/>Governance and Sharing Provisions);
- **NHS Act**means the National Health Service Act 2006 (as amended by<br/>the Health and Social Care Act 2012 and the Health and Care<br/>Act 2022 or other legislation from time to time);
- NHS Business Servicesmeans the Special Health Authority established under the<br/>NHS Business Services Authority (Establishment and<br/>Constitution Order) 2005 SI 2005/2414;
- NHS Counter Fraudmeans the Special Health Authority established by and in<br/>accordance with the NHS Counter Fraud Authority<br/>(Establishment, Constitution, and Staff and Other Transfer<br/>Provisions) Order 2017/958;
- NHS England means the body established by section 1H of the NHS Act;
- **NHS England Deliverables** means all documents, products and materials NHS England in which NHS England holds IPRs which are relevant to this Agreement, the Delegated Functions or the Reserved Functions in any form and made available by NHS England to the ICB under this Agreement, including data, reports, policies, plans and specifications;
- **NHS England Functions** means all functions of NHS England as set out in Legislation excluding any functions that have been expressly delegated;
- **Non-Personal Data** means data which is not Personal Data;
- Out of Hours Contract means a primary medical services contract for the provision of primary medical services solely during the out of hours period (6.30pm Monday to Thursday until 8am the next day, 6.30pm Friday to 8am Monday, Christmas Day, Good Friday and bank holidays);

Operational Days	a day other than a Saturday, Sunday, Christmas Day, Good Friday or a bank holiday in England;					
Particulars	means the Particulars of this Agreement as set out in clause 1 <i>(Particulars)</i> ;					
Party/Parties	means a party or both parties to this Agreement;					
PDS Agreement	means a Personal Dental Services Agreement made under section 107 of the NHS Act;					
Performers Lists	The lists of healthcare professionals maintained by NHS England pursuant to the National Health Service (Performers Lists) (England) Regulations 2013;					
Personal Data	shall have the same meaning as set out in the UK GDPR and shall include references to Special Category Personal Data where appropriate;					
Pharmaceutical List	means a list of persons who undertake to provide pharmaceutical services pursuant to regulation 10 of the Pharmaceutical Regulations;					
Pharmaceutical Regulations	means the National Health Service (Pharmaceutical and Local Pharmaceutical Services) Regulations 2013/349;					
Pharmaceutical Services	means:-					
	<ul> <li>services provided pursuant to arrangements under section 126 of the NHS Act; and</li> </ul>					
Pharmaceutical Services Arrangement	section 126 of the NHS Act; and					
	section 126 of the NHS Act; and (ii) Additional Pharmaceutical Services means an arrangement for the provision of Pharmaceutical					
Arrangement Pharmaceutical Services	section 126 of the NHS Act; and (ii) Additional Pharmaceutical Services means an arrangement for the provision of Pharmaceutical Services, including inclusion in a Pharmaceutical List; means a natural or legal person who is party to a Pharmaceutical Services Arrangement or Local					
Arrangement Pharmaceutical Services Provider	section 126 of the NHS Act; and (ii) Additional Pharmaceutical Services means an arrangement for the provision of Pharmaceutical Services, including inclusion in a Pharmaceutical List; means a natural or legal person who is party to a Pharmaceutical Services Arrangement or Local Pharmaceutical Services Contract; means an agreement made in accordance with section 92 of					
Arrangement Pharmaceutical Services Provider PMS Agreement	<ul> <li>section 126 of the NHS Act; and</li> <li>(ii) Additional Pharmaceutical Services</li> <li>means an arrangement for the provision of Pharmaceutical Services, including inclusion in a Pharmaceutical List;</li> <li>means a natural or legal person who is party to a Pharmaceutical Services Arrangement or Local Pharmaceutical Services Contract;</li> <li>means an agreement made in accordance with section 92 of the NHS Act;</li> <li>means the individuals for whom the ICB is responsible for</li> </ul>					
Arrangement Pharmaceutical Services Provider PMS Agreement Population	<ul> <li>section 126 of the NHS Act; and</li> <li>(ii) Additional Pharmaceutical Services</li> <li>means an arrangement for the provision of Pharmaceutical Services, including inclusion in a Pharmaceutical List;</li> <li>means a natural or legal person who is party to a Pharmaceutical Services Arrangement or Local Pharmaceutical Services Contract;</li> <li>means an agreement made in accordance with section 92 of the NHS Act;</li> <li>means the individuals for whom the ICB is responsible for commissioning health services;</li> <li>means tenancies, leases and other arrangements in relation to the occupation of land for the delivery of services under the</li> </ul>					

Prescribed Dental Services	means the dental services prescribed by such regulations made pursuant to section 3B(1)(a) of the NHS Act as are in force from time to time (including, for the avoidance of doubt, services commonly known as secondary care dental services and Community Dental Services);				
Prescribed Dental Services Contract	means any contract for the provision of Prescribed Dental Services;				
Primary Care Contract or Arrangement (PCCA)	means:				
Analigement (FCCA)	(i) a Primary Medical Services Contract;				
	(ii) a Dental Services Contract;				
	(iii) a Primary Ophthalmic Services Contract;				
	(iv) a Local Pharmaceutical Services Contract; and				
	(v) a Pharmaceutical Services Arrangement.				
Primary Care Functions	means:-				
	<ul> <li>the statutory functions conferred on NHS England under Parts 4, 5, 6 and 7 of the NHS Act and secondary legislation made under those Parts; and</li> </ul>				
	<ul> <li>(ii) the other statutory functions conferred on NHS England by either primary legislation, secondary legislation or by arrangement with another person in so far as they are applicable to the discharge of those functions set out at (i) above;</li> </ul>				
Primary Care Provider	means a natural or legal person who holds a Primary Care Contract, or is a Pharmaceutical Services Provider;				
Primary Care Provider Personnel	means all persons (whether clinical or non-clinical) employed or engaged by a Primary Care Provider or by any Sub- Contractor (including volunteers, agency, locums, casual or seconded personnel) in the provision of Services or any activity related to or connected with the provision of the Services;				
Primary Care Services	means the services in respect of which NHS England has a duty or power to make arrangements pursuant to the Primary Care Functions;				
Primary Dental Services	means primary dental care services provided under arrangements made pursuant to Part 5 of the NHS Act, and in accordance with a Dental Services Contract;				
Primary Medical Services	means primary medical services provided under arrangements made pursuant to Part 4 of the NHS Act, and in accordance with a Primary Medical Services Contract;				
Primary Medical Services Contract	means:				

(	ï	)	а	PMS	hΑ	reer	nent;
		/	а		лy	1661	nent,

- (ii) a GMS Contract;
- (iii) an APMS Contract; and
- (iv) any other contract for the provision of health services made pursuant to NHS England's functions under Part 4 of the NHS Act;

in each case as amended or replaced from time to time and including all ancillary or related agreements directly relating to the subject matter of such agreements, contracts or arrangements but excluding any Premises Agreements and excluding any Out of Hours Contracts<sup>2</sup>;

Primary Medical Servicesmeans a natural or legal person who holds a Primary MedicalProviderServices Contract;

means:

Primary Ophthalmic Services Contract

- (i) a General Ophthalmic Services Contract; and
- (ii) any other contract for the provision of health services made pursuant to NHS England's functions under Part 6 of the NHS Act;

in each case as amended or replaced from time to time and including all ancillary or related agreements directly relating to the subject matter of such agreements, contracts or arrangements;

Primary Ophthalmicmeans a natural or legal person who holds a PrimaryServices ProviderOphthalmic Services Contract;

# **Principles of Best Practice** means the Mandated Guidance in relation to property and investment which is to be published either before or after the date of this Agreement;

Prohibited Act the ICB:

- (i) offering, giving, or agreeing to give NHS England (or any of their officers, employees or agents) any gift or consideration of any kind as an inducement or reward for doing or not doing or for having done or not having done any act in relation to the obtaining of performance of this Agreement, the Reserved Functions, the Delegation or any other arrangement with the ICB, or for showing or not showing favour or disfavour to any person in relation to this Agreement or any other arrangement with the ICB; and
- (ii) in connection with this Agreement, paying or agreeing to pay any commission, other than a payment, particulars of which (including the terms

<sup>&</sup>lt;sup>2</sup> Arrangements for Out of Hours Contracts are dealt with under separate Directions outside of this Agreement and do not form part of any Delegated Functions.

and conditions of the agreement for its payment) have been disclosed in writing to NHS England; or

(iii) committing an offence under the Bribery Act 2010;

#### **QOF** means the quality and outcomes framework;

Regulatory or Supervisory Body

means any statutory or other body having authority to issue guidance, standards or recommendations with which the relevant Party and/or Staff must comply or to which it or they must have regard, including:

- (i) CQC;
- (ii) NHS England;
- (iii) the Department of Health and Social Care;
- (iv) the National Institute for Health and Care Excellence;
- (v) Healthwatch England and Local Healthwatch;
- (vi) the General Medical Council;
- (vii) the General Dental Council;
- (viii) the General Optical Council;
- (ix) the General Pharmaceutical Council;
- (x) the Healthcare Safety Investigation Branch; and
- (xi) the Information Commissioner;
- **Relevant Information** means the Personal Data and Non-Personal Data processed under the Delegation and this Agreement, and includes, where appropriate, "confidential patient information" (as defined under section 251 of the NHS Act), and "patient confidential information" as defined in the 2013 Report, The Information Governance Review "To Share or Not to Share?");
- **Reserved Functions** means statutory functions of NHS England that it has not delegated to the ICB including but not limited to those set out in the Schedules to this Agreement;
- Secretary of State means the Secretary of State for Health and Social Care from time to time;
- Section 7A Functions means those functions of NHS England exercised pursuant to section 7A of the NHS Act and relating to Primary Care Services;
- Section 7A Funds shall have the meaning in Schedule 10 Part 2;

```
Special Category Personal shall have the same meaning as in UK GDPR; Data
```

- **Specified Purpose** means the purpose for which the Relevant Information is shared and processed, being to facilitate the exercise of the ICB's Delegated Functions and NHS England's Reserved Functions as specified in paragraph 2.1 of **Error! Reference source not found.** (*Further Information Governance and Sharing Provisions*) to this Agreement;
- Staff or Staffing means the Parties' employees, officers, elected members, directors, voluntary staff, consultants, and other contractors and sub-contractors acting on behalf of either Party (whether or not the arrangements with such contractors and sub-contractors are subject to legally binding contracts) and such contractors' and their sub-contractors' personnel;
- Staffing Modelmeans the employment model for the exercise of the<br/>Delegated Functions including those as defined in Appendix<br/>2 of the NHS England and NHS Improvement operating<br/>models: HR Framework for developing Integrated Care as<br/>may be amended or replaced from time to time;
- Statement of Financial<br/>Entitlements Directionsmeans the General Medical Services Statement of Financial<br/>Entitlements Directions 2021, as amended or updated from<br/>time to time;
- **Sub-Delegate** shall have the meaning in clause 11.2;
- **Transfer Regulations** means the Transfer of Undertakings (Protection of Employment) Regulations 2006, as amended;

Triple Aimmeans the duty to have regard to the wider effects of<br/>decisions, which is placed on each of the Parties under<br/>section 13NA (as regards NHS England) and section 14Z43<br/>(as regards the ICB) of the NHS Act;

- **UK GDPR** means Regulation (EU) 2016/679 of the European Parliament and of the Council of 27th April 2016 on the protection of natural persons with regard to the processing of personal data and on the free movement of such data (General Data Protection Regulation) as it forms part of the law of England and Wales, Scotland and Northern Ireland by virtue of section 3 of the European Union (Withdrawal) Act 2018;
- **Variation Proposal** means a written proposal for a variation to the Agreement, which complies with the requirements of clause 25.3.

# SCHEDULE 2

# **Delegated Functions**

# Schedule 2A: Primary Medical Services

#### Part 1: General Obligations

#### 1. Introduction

- 1.1 This Part 1 of Schedule 2A (*Primary Medical Services*) sets out further provision regarding the carrying out of those Delegated Functions relating to Primary Medical Services, being in summary:
  - 1.1.1 decisions in relation to the commissioning and management of Primary Medical Services;
  - 1.1.2 planning Primary Medical Services in the Area, including carrying out needs assessments;
  - 1.1.3 undertaking reviews of Primary Medical Services in respect of the Area;
  - 1.1.4 management of the Delegated Funds in the Area;
  - 1.1.5 co-ordinating a common approach to the commissioning and delivery of Primary Medical Services with other health and social care bodies in respect of the Area where appropriate; and
  - 1.1.6 such other ancillary activities that are necessary in order to exercise the Delegated Functions.

#### 2. General Obligations

- 2.1 The ICB is responsible for planning the commissioning of primary medical services.
- 2.2 The role of the ICB includes:
  - 2.2.1 carrying out needs assessments, and regular reviews of such assessments, to determine the needs of the population in the Area; and
  - 2.2.2 identifying and implementing changes to meet any unmet needs which may be met through the delivery of Primary Medical Services.
- 2.3 In respect of integrated working, the ICB must:
  - 2.3.1 take an integrated approach to working and co-ordinating with stakeholders including NHS England, Local Authorities, Healthwatch, acute and community providers, the Local Medical Committee, and other stakeholders;
  - 2.3.2 work with NHS England and other ICBs to co-ordinate a common approach to the commissioning of Primary Medical Services generally; and
  - 2.3.3 work with NHS England to coordinate the exercise of their respective performance management functions.
- 2.4 In relation to the Delegated Functions, the ICB agrees to perform the following general obligations:
  - 2.4.1 to manage the Primary Medical Services Contracts and perform all of NHS England's obligations under each of the Primary Medical Services Contracts

in accordance with the terms of the Primary Medical Services Contracts as if it were named in the contract in place of NHS England;

- 2.4.2 actively manage the performance of the Primary Medical Services Provider in order to secure the needs of people who use the services, improve the quality of services and improve efficiency in the provision of the services including by taking timely action to enforce contractual breaches, serve notices or provide discretionary support;
- 2.4.3 ensure that it obtains value for money on behalf of NHS England, and avoids making any double payments under any Primary Medical Services Contracts;
- 2.4.4 notify NHS England immediately (or in any event within two (2) Operational Days) of any breach by the ICB of its obligations to perform any of NHS England's obligations under the Primary Medical Services Contracts;
- 2.4.5 undertake any investigations relating (among other things) to whistleblowing claims, infection control and patient complaints;
- 2.4.6 keep a record of all of the Primary Medical Services Contracts that the ICB manages setting out the following details in relation to each Primary Medical Services Contract:
  - 2.4.6.1 name of the Primary Medical Services Provider;
  - 2.4.6.2 the name by which the Primary Medical Services Provider is known (if different to the name recorded under paragraph 2.4.6.1);
  - 2.4.6.3 location of provision of services; and
  - 2.4.6.4 amounts payable under the Primary Medical Services Contract (if a contract sum is payable) or amount payable in respect of each patient (if there is no contract sum).
- 2.5 Without prejudice to clause 9 (*Finance*) of the Agreement or paragraph 2.4 above, the ICB must actively manage each of the relevant Primary Medical Services Contracts including by:
  - 2.5.1 reviewing the performance of the relevant Primary Medical Services Contract, including in respect of quality standards, incentives and the QOF, observance of service specifications, and monitoring of activity and finance;
  - 2.5.2 assessing quality and outcomes (including clinical effectiveness, patient experience, patient safety and addressing inequalities);
  - 2.5.3 managing variations to the relevant Primary Medical Services Contract or services in accordance with national policy, service user needs and clinical developments;
  - 2.5.4 agreeing information and reporting requirements and managing information breaches (which will include use of the NHS Digital Data Security and Protection Toolkit);
  - 2.5.5 agreeing local prices, managing agreements or proposals for local variations and local modifications;
  - 2.5.6 conducting review meetings and undertaking contract management including the issuing of contract queries and agreeing any remedial action plan or related contract management processes; and

- 2.5.7 complying with and implementing any relevant Mandated Guidance issued from time to time.
- 2.6 This paragraph is without prejudice to clause 10 (*Information, Planning and Reporting*) or any other provision in this Agreement. The ICB must provide NHS England with:
  - 2.6.1 such information relating to individual Primary Medical Services Providers in the Area as NHS England may reasonably request, to ensure that NHS England is able to continue to gather national data regarding the commissioning or performance of Primary Medical Services Providers;
  - 2.6.2 such data/data sets as required by NHS England to ensure population of any national dashboards;
  - 2.6.3 any other data/data sets as required by NHS England; and
  - 2.6.4 the ICB shall procure that providers accurately record and report information so as to allow NHS England and other agencies to discharge their functions.
- 2.7 It should be noted that while the ICB is also required to exercise functions in respect of dispensing doctors, arrangements in respect of these functions are described in Schedule 2D (Pharmaceutical Services).

# Part 2: Specific Obligations

### 1. Introduction

This Part 2 of Schedule 2A (Delegated Functions – Primary Medical Services) sets out further provision regarding the carrying out of each of the Delegated Functions.

# 2. Primary Medical Services Contract Management

The ICB must comply with any future national Mandated Guidance on equitable funding as may apply from time to time.

#### 3. Enhanced Services

- 3.1 The ICB must manage the design (where applicable) and commissioning of any Enhanced Services, including re-commissioning these services annually where appropriate.
- 3.2 The ICB may consider any local enhanced services entered into with Primary Medical Services Providers in its Area using NHS Standard Contracts. Where these would continue to be beneficial to the Area, the ICB may manage the ongoing design and commissioning (including re-commissioning) of these services via a Local Incentives Scheme.
- 3.3 The ICB must ensure that it complies with any Mandated Guidance in relation to the design and commissioning of Enhanced Services.
- 3.4 When commissioning newly designed Enhanced Services the ICB must:
  - 3.4.1 consider the needs of the local population in the Area;
  - 3.4.2 develop the necessary specifications and templates for the Enhanced Services, as required to meet the needs of the local population in the Area;
  - 3.4.3 when developing the necessary specifications and templates for the Enhanced Services, ensure that value for money will be obtained;

- 3.4.4 consult with Local Medical Committees and other stakeholders and comply with the duty of public involvement and consultation under section 14Z45 of the NHS Act;
- 3.4.5 liaise with system providers and representative bodies to ensure that the system in relation to the Directed Enhanced Services, NHS England Enhanced Services and Local Enhanced Services will be functional and secure;
- 3.4.6 support Data Controllers in providing 'fair processing' information as required by the UK GDPR; and
- 3.4.7 support Primary Medical Services Providers in entering into data processing agreements with data processors in the terms required by the UK GDPR.

#### 4. Design of Local Incentive Schemes

- 4.1 The ICB may design and offer Local Incentive Schemes for Primary Medical Services Providers, sensitive to the differing needs of their particular communities. This includes in addition to or as an alternative to the national contractual frameworks (including as an alternative to QOF or Enhanced Services), provided that such schemes are voluntary, and the ICB continues to offer the national schemes.
- 4.2 There is no formal approvals process that the ICB must follow to develop a Local Incentive Scheme, although when designing and implementing any proposed new Local Incentive Scheme the ICB must:
  - 4.2.1 consider the needs of the local population in the Area;
  - 4.2.2 develop the specifications and templates for the Local Incentive Scheme;
  - 4.2.3 consult with Local Medical Committees and other stakeholders and comply with the duty of public involvement and consultation under section 14Z45 of the NHS Act;
  - 4.2.4 liaise with system providers and representative bodies to ensure that the system in relation to the Local Incentive Schemes will be functional and secure;
  - 4.2.5 support Data Controllers in providing privacy information as required by the UK GDPR; and
  - 4.2.6 support Primary Medical Services Providers in entering into data processing agreements with data processors in terms required by the UK GDPR.
- 4.3 The ICB must be able to:
  - 4.3.1 demonstrate improved outcomes, reduced inequalities and value for money;
  - 4.3.2 support ongoing national reporting requirements (where applicable); and
  - 4.3.3 must reflect the changes agreed as part of the national PMS reviews ( <u>https://www.england.nhs.uk/commissioning/wp-</u> content/uploads/sites/12/2016/05/implement-pms-fund-changes.pdf ).
- 4.4 The ongoing assurance of any new Local Incentive Schemes will form part of the ICB's assurance process under any applicable assurance framework.
- 4.5 Any new Local Incentive Scheme must be implemented without prejudice to the right of Primary Medical Services Providers operating under a GMS Contract to obtain their entitlements which are negotiated and set nationally.

4.6 NHS England will continue to set national standing rules, to be reviewed annually, and the ICB must comply with these rules which shall for the purposes of this Agreement be Mandated Guidance.

#### 5. Making Decisions on Discretionary Payments or Support

- 5.1 The ICB must manage and make decisions in relation to any discretionary payments or discretionary support to be made to Primary Medical Services Providers in a consistent, open and transparent way.
- 5.2 The ICB must exercise its discretion to determine the level of payment or type of support to Primary Medical Services Providers, in accordance with any relevant Mandated Guidance.

## 6. Making Decisions about Commissioning Urgent Care for Out of Area Registered Patients

- 6.1 The ICB must manage the design and commissioning of urgent care services (including home visits as required) for its patients registered out of area (including recommissioning these services annually where appropriate).
- 6.2 The ICB must ensure that it complies with any Mandated Guidance in relation to the design and commissioning of these services.
- 6.3 For the purposes of paragraph 6.1, urgent care means the provision of primary medical services on an urgent basis.

# 7. Transparency and freedom of information

- 7.1 The ICB must:
  - 7.1.1 Respond to requests for information from members of the public and the media, including requests made pursuant to the FOIA, whose subject-matter relates to the performance of the Delegated Functions in the ICB's Area; and
  - 7.1.2 Provide information and assistance as required to support NHS England in the preparation of responses to parliamentary questions in connection with the Delegated Functions.

#### 8. Planning the Provider Landscape

- 8.1 The ICB must plan the primary medical services provider landscape in the Area, including considering and taking decisions in relation to:
  - 8.1.1 establishing new Primary Medical Services Providers in the Area;
  - 8.1.2 managing Primary Medical Services Providers providing inadequate standards of patient care;
  - 8.1.3 the procurement or award of new Primary Medical Services Contracts (in accordance with any procurement protocol or Guidance issued by NHS England from time to time);
  - 8.1.4 closure of practices and branch surgeries;
  - 8.1.5 dispersing the patient lists of Primary Medical Services Providers; and
  - 8.1.6 agreeing variations to the boundaries of Primary Medical Services Providers.
- 8.2 In relation to any new Primary Medical Services Contract to be entered into, the ICB must, without prejudice to any obligation in paragraph 14 (*Procurement and New Contracts*) below, and paragraph 2.5 of Part 1 of this Schedule 2A:

- 8.2.1 consider and use the form of Primary Medical Services Contract that will ensure compliance with NHS England's obligations under Law taking into account the persons to whom such Primary Medical Services Contracts may be awarded;
- 8.2.2 provide to NHS England confirmation as required from time to time that it has considered and complied with its obligations under this Agreement and the Law; and
- 8.2.3 for the avoidance of doubt, Schedule 5 (Financial Provisions and Decision Making Limits) deals with the sign off requirements for Primary Medical Services Contracts.

# 9. Primary Care Networks

- 9.1 In managing the design and commissioning of the Network Contract Directed Enhanced Services, including re-commissioning these services annually where appropriate, the ICB must plan and manage the Primary Care Networks in the Area, complying with published specifications and Mandated Guidance, including to:
  - 9.1.1 maintain or establish identified Network Areas to support the local population in the Area;
  - 9.1.2 review any waived PCN list size requirements wherever possible and appropriate to best support the local population in the Area;
  - 9.1.3 ensure that each PCN has at all times an accountable Clinical Director;
  - 9.1.4 align each PCN with an ICB that would best support delivery of services to the local population in the Area; and
  - 9.1.5 collaborate and work with other ICBs as appropriate to agree which ICB will be the lead ICB for the PCN.

#### 10. Approving Primary Medical Services Provider Mergers and Closures

- 10.1 The ICB is responsible for approving Primary Medical Services Provider mergers and Primary Medical Services Provider closures in the Area.
- 10.2 The ICB must undertake all necessary consultation when taking any decision in relation to Primary Medical Services Provider mergers or Primary Medical Services Provider closures in the Area, including those set out under section 14Z45 of the NHS Act (duty for public involvement and consultation). The consultation undertaken must be appropriate and proportionate in the circumstances and should include consulting with the Local Medical Committee.
- 10.3 Prior to making any decision in accordance with this paragraph 10 (*Approving Primary Medical Services Provider Mergers and Closures*), the ICB must be able to clearly demonstrate the grounds for such a decision and must have fully considered any impact on the Primary Medical Services Provider's registered population and that of surrounding practices. The ICB must be able to clearly demonstrate that it has considered other options and has entered into dialogue with the Primary Medical Services Provider or merger will be managed.
- 10.4 In making any decisions pursuant to this paragraph 10 (*Approving Primary Medical Services Provider Mergers and Closures*), the ICB shall act in accordance with relevant Mandated Guidance and also take account of its obligations as set out in paragraph 14 (*Procurement and New Contracts*), below, where applicable.

# 11. Making Decisions in relation to Management of Poorly Performing Primary Medical Services Providers

- 11.1 The ICB must make decisions in relation to the management of poorly performing Primary Medical Services Provider including, without limitation, decisions and liaison with the CQC where the CQC has reported non-compliance with standards (but excluding any decisions in relation to the Performers List).
- 11.2 In accordance with paragraph 11.1 above, the ICB must:
  - 11.2.1 ensure regular and effective collaboration with the CQC to ensure that information on general practice is shared and discussed in an appropriate and timely manner;
  - 11.2.2 ensure that any risks identified are managed and escalated where necessary;
  - 11.2.3 respond to CQC assessments of Primary Medical Services Providers where improvement is required;
  - 11.2.4 where a Primary Medical Services Provider is placed into special measures, lead a quality summit to ensure the development and monitoring of an appropriate improvement plan (including a communications plan and actions to manage primary care resilience in the locality); and
  - 11.2.5 take appropriate contractual action, including (without limitation) in response to CQC findings.

#### 12. **Premises Costs Directions Functions**

- 12.1 The ICB must comply with the Premises Costs Directions and will be responsible for making decisions in relation to the Premises Costs Directions Functions.
- 12.2 In particular, but without limiting paragraph 12.1, the ICB shall make decisions concerning:
  - 12.2.1 applications for new payments under the Premises Costs Directions (whether such payments are to be made by way of grants or in respect of recurring premises costs); and
  - 12.2.2 revisions to existing payments being made under the Premises Costs Directions.
- 12.3 The ICB must comply with any decision-making limits set out in SCHEDULE 5 (*Financial Provisions and Decision Making Limits*) when taking decisions in relation to the Premises Costs Directions Functions.
- 12.4 The ICB will comply with any Guidance issued by the Secretary of State or NHS England in relation to the Premises Costs Directions, including the Principles of Best Practice, and any other Mandated Guidance in relation to the Premises Costs Directions.
- 12.5 The ICB must work to ensure that the premises estate is properly managed and maintained, including by ensuring strategic estates planning is in place, and work cooperatively with other ICBs as appropriate.
- 12.6 The ICB must ensure it maintains comprehensive records of the primary care estate and any changes to it.
- 12.7 The ICB must liaise where appropriate with NHS Property Services Limited and Community Health Partnerships Limited in relation to the Premises Costs Directions Functions.
- 12.8 The ICB must prioritise the following measures in respect of management of the primary care estate in the Area:

- 12.8.1 working collaboratively with landlords and tenants to maximise the use of existing estate;
- 12.8.2 effective asset management practices including (without limitation) regularisation of the occupation of the estate, lease events, rent reviews and up-to-date documentation management; and
- 12.8.3 seeking the resolution of premises disputes in a timely manner.

#### 13. Maintaining the Performers List

On receiving a notice from a practitioner (who is party to a Primary Medical Services Contract) of an amendment to information recorded about them in the Performers List, pursuant to regulation 9(1) of the National Health Service (Performers Lists) (England) Regulations 2013, the ICB must support NHS England's amendment of the Performers List as soon as possible after receiving the notice using the Primary Care Support services provided by NHS England, insofar as that amendment relates to a change in contractor details.

#### 14. **Procurement and New Contracts**

- 14.1 Until any new arrangements for awarding Primary Medical Services Contracts comes into force, the ICB will make procurement decisions relevant to the exercise of the Delegated Functions and in accordance with the detailed arrangements regarding procurement set out in the procurement protocol issued and updated by NHS England from time to time.
- 14.2 In discharging its responsibilities set out in this Schedule 2A, the ICB must comply at all times with Law and any relevant Guidance (including any applicable procurement law and/or guidance on the selection of, and award of contracts to, providers of healthcare services).
- 14.3 On the coming into force of new arrangements for awarding Primary Medical Services Contracts, the ICB will make decisions on awarding new contracts relevant to the exercise of the Delegated Functions.
- 14.4 When the ICB makes decisions in connection with the awarding of Primary Medical Services Contracts it should ensure that it is able to demonstrate compliance with requirements for the award of Primary Medical Services Contracts, including that the decision was:
  - 14.4.1 made in the best interest of patients, taxpayers and the population;
  - 14.4.2 robust and defensible, with conflicts of interests appropriately managed;
  - 14.4.3 made transparently; and
  - 14.4.4 compliant with the rules of the regime as set out in NHS England guidance.
- 14.5 Where the ICB wishes to develop and offer a locally designed contract, it must ensure that it has consulted with the relevant Local Medical Committees in relation to the proposal and that it can demonstrate that the scheme will:
  - 14.5.1 improve outcomes for patients;
  - 14.5.2 reduce inequalities in the population; and
  - 14.5.3 provide value for money.
- 15. Complaints

15.1 The ICB will handle complaints made in respect of Primary Medical Services in accordance with the Complaints Regulations.

## 16. Commissioning ancillary support services

- 16.1 The ICB must procure, and undertake the management and monitoring of contracts for the provision of, such ancillary support services as are required to support the ICB in the effective discharge of the Delegated Functions, including, but not limited to the following:
  - 16.1.1 collection and disposal of clinical waste;
  - 16.1.2 provision of translation and interpretation services;
  - 16.1.3 occupational health services.

# 17. Finance

Further requirements in respect of finance will be specified in Mandated Guidance.

#### 18. Workforce

- 18.1 The arrangements for the provision and maintenance of sufficient and appropriately qualified, trained and experienced Staff in order for the ICB to fulfil its responsibilities for each of the Delegated Functions ("the Staffing Model"), will be communicated formally to the ICB by NHS England following recommendations made by the National Moderation Panel.
- 18.2 The ICB is not permitted to vary the Staffing Model agreed with NHS England as part of its application for delegation of the said functions however a variation can be applied for by the ICB and considered by the National Moderation Panel at any time.

#### Schedule 2B: Dental Care Services

The provisions of this Schedule 2B form part of this Agreement only where indicated in the Particulars.

#### Part 1A: General Obligations – Primary Dental Services

# 1. Introduction

- 1.1 This Part 1A of Schedule 2B (*Dental Care Services*) sets out general provisions regarding the carrying out of those Delegated Functions relating to Primary Dental Services, being in summary:
  - 1.1.1 decisions in relation to the commissioning and management of Primary Dental Services;
  - 1.1.2 planning Primary Dental Services in the Area, including carrying out needs assessments;
  - 1.1.3 undertaking reviews of Primary Dental Services in the Area;
  - 1.1.4 management of the Delegated Funds in the Area;
  - 1.1.5 co-ordinating a common approach to the commissioning and delivery of Primary Dental Services with other health and social care bodies in respect of the Area where appropriate; and
  - 1.1.6 such other ancillary activities that are necessary in order to exercise the Delegated Functions.

#### 2. General Obligations

- 2.1 The ICB is responsible for planning the commissioning of Primary Dental Services.
- 2.2 When planning and commissioning Primary Dental Services, the ICB must comply with Mandated Guidance issued by NHS England.
- 2.3 In respect of integrated working, the ICB must:
  - 2.3.1 take an integrated approach to working and co-ordinating with stakeholders including NHS England, Local Dental Professional Networks, Local Authorities, Healthwatch, acute and community providers, the Local Dental Committee, and other stakeholders;
  - 2.3.2 work with NHS England and other ICBs to co-ordinate a common approach to the commissioning of Primary Dental Services generally; and
  - 2.3.3 work with NHS England to coordinate the exercise of their respective performance management functions.
- 2.4 In relation to the Delegated Functions, the ICB agrees to perform the following general obligations with regard to Dental Services Contracts:
  - 2.4.1 to manage the Dental Services Contracts and perform all of NHS England's obligations under each of the Dental Services Contracts in accordance with the terms of the Dental Services Contracts as if it were named in the contract in place of NHS England;
  - 2.4.2 working with other organisations, including the NHS Business Services Authority and the NHS England specialised commissioning team as appropriate, actively manage the performance of the Dental Services Provider in order to secure the needs of people who use the services,

improve the quality of services and improve efficiency in the provision of the services including by taking timely action to enforce contractual breaches, serve notices or provide discretionary support;

- 2.4.3 ensure that it obtains value for money on behalf of NHS England, including by avoiding making any double payments under any Dental Services Contracts and reducing the number of contracts which are under-delivering so that funds can be reallocated to meet local oral health needs;
- 2.4.4 notify NHS England immediately (or in any event within two (2) Operational Days) of any breach by the ICB of its obligations to perform any of NHS England's obligations under the Dental Services Contracts;
- 2.4.5 undertake any investigations relating (among other things) to whistleblowing claims, infection control and patient complaints;
- 2.4.6 keep a record of all of the Dental Services Contracts that the ICB manages on behalf of NHS England setting out the following details in relation to each Dental Services Contract:
  - 2.4.6.1 name of Dental Services Provider;
  - 2.4.6.2 any practice or trading name by which the Dental Services Provider is known (if different to the name recorded under paragraph 2.4.6.1);
  - 2.4.6.3 location of provision of services; and
  - 2.4.6.4 amounts payable under the contract (if a contract sum is payable) or amount payable in respect of each patient (if there is no contract sum).
- 2.5 Without prejudice to clause 9 *(Finance)* or paragraph 2.4 above, the ICB must actively manage each of the relevant Dental Services Contracts including by:
  - 2.5.1 reviewing and monitoring spending on services provided pursuant to Dental Services Contracts in the Area;
  - 2.5.2 reviewing and monitoring spending on Primary Dental Services commissioned in the Area;
  - 2.5.3 creating purchase orders, coding invoices and making appropriate amendments within the Compass contractor payments system;
  - 2.5.4 managing the relevant Dental Services Contract, including in respect of quality standards, incentives, observance of service specifications, and monitoring of activity and finance;
  - 2.5.5 assessing quality and outcomes (including clinical effectiveness, patient experience and patient safety);
  - 2.5.6 managing variations to the relevant Dental Services Contract or services in accordance with national policy, service user needs and clinical developments;
  - 2.5.7 agreeing information and reporting requirements and managing information breaches (which will include use of the NHS Digital Data Security and Protection Toolkit);

- 2.5.8 undertaking annual contract activity negotiations, including agreeing local prices, managing agreements or proposals for local variations and local modifications;
- 2.5.9 conducting review meetings and undertaking contract management including the issuing of contract queries and agreeing any remedial action plan or related contract management processes;
- 2.5.10 allocating sufficient resources for undertaking contract mediation; and
- 2.5.11 complying with and implementing any relevant Mandated Guidance issued from time to time.
- 2.6 This paragraph is without prejudice to clause 10 (*Information, Planning and Reporting*) or any other provision in this Agreement. The ICB must provide NHS England with:
  - 2.6.1 such information relating to individual providers of Primary Dental Services in the Area as NHS England may reasonably request, to ensure that NHS England is able to continue to gather national data regarding the commissioning or performances of providers of Primary Dental Services;
  - 2.6.2 such data/data sets as required by NHS England to ensure population of any national dashboards;
  - 2.6.3 any other data/data sets as required by NHS England; and
  - 2.6.4 the ICB shall procure that providers accurately record and report information so as to allow NHS England and other agencies to discharge their functions.

#### Part 1B: Specific Obligations – Primary Dental Services only

#### 1. Introduction

1.1 This Part 1B of Schedule 2B (*Dental Care Services*) sets out further provision regarding the carrying out of each of the Delegated Functions in relation to Primary Dental Services.

## 2. Dental Services Contract Management

- 2.1 The ICB must:
  - 2.1.1 comply with all current and future relevant national Mandated Guidance regarding contract reviews;
  - 2.1.2 monitor contract performance and primary care dental spending, with a view in particular to achieving a reduction in the number of contract holders who are under-delivering, and the reallocation of unused resources to meet the oral health needs of the Area; and
  - 2.1.3 in cooperation with the NHS Business Services Authority, monitor contract performance with a view in particular to addressing patient safety concerns and promoting patient safety.
- 2.2 The ICB must undertake the annual reconciliation of monies claimed by providers against the services provided under any contract for the provision of Dental Care Services made pursuant to NHS England's functions under Part 5 of the NHS Act procuring such ancillary support services as are required for the performance of this function.

# 3. Transparency and freedom of information

- 3.1 The ICB must:
  - 3.1.1 respond to requests for information from members of the public and the media, including requests made pursuant to the FOIA, whose subject-matter relates to the performance of the Delegated Functions in the ICB's Area; and
  - 3.1.2 provide information and assistance as required to support NHS England in the preparation of responses to parliamentary questions in connection with the Delegated Functions.

#### 4. Planning the Provider Landscape

- 4.1 The ICB must plan the provider landscape in the Area, including considering and taking decisions in relation to:
  - 4.1.1 establishing new Dental Services Providers in the Area;
  - 4.1.2 managing Dental Services Providers providing inadequate standards of patient care;
  - 4.1.3 the procurement or award of new Dental Services Contracts (in accordance with any procurement protocol or Guidance issued by NHS England from time to time); and
  - 4.1.4 closure of practices.
- 4.2 In relation to any new Dental Services Contract to be entered into, the ICB must, without prejudice to any obligation in paragraph 10 (*Procurement and New Contracts*), below:
  - 4.2.1 consider and use the form of Dental Services Contract that will ensure compliance with NHS England's obligations under Law taking into account the persons to whom such Dental Services Contracts may be awarded;
  - 4.2.2 provide to NHS England confirmation as required from time to time that it has considered and complied with its obligations under this Agreement and the Law; and
  - 4.2.3 for the avoidance of doubt, SCHEDULE 5 (*Financial Provisions and Decision Making Limits*) deals with the sign off requirements for Dental Services Contracts.

# 5. Finance

5.1 Further requirements in respect of finance will be specified in Mandated Guidance.

#### 6. Staffing and Workforce

- 6.1 Subject to the terms of this Agreement, the Delegated Functions will be carried out by NHS England Staff in accordance with decisions concerning the Delegated Functions made by the ICB unless the Staff carrying out the Delegated Functions have transferred to the ICB (and/or the ICB has engaged or employed Staff for that purpose).
- 6.2 The arrangements for the provision and maintenance of sufficient and appropriately qualified, trained and experienced Staff in order for the ICB to fulfil its responsibilities for each of the Delegated Functions ("the Staffing Model"), will be communicated formally to the ICB by NHS England following recommendations made by the National Moderation Panel. Further requirements in respect of workforce will be specified in Mandated Guidance.

6.3 The ICB is not permitted to vary the Staffing Model agreed with NHS England as part of its application for delegation of the said functions however a variation can be applied for by the ICB and considered by the National Moderation Panel at any time.

#### 7. Integrating dentistry into communities at Primary Care Network level

7.1 The ICB must exercise the Delegated Functions with a view to achieving greater integration of dentists into the Integrated Care System at the Primary Care Network level.

#### 8. Making Decisions in relation to Management of Poorly Performing Dental Services Providers

- 8.1 The ICB must make decisions in relation to the management of poorly performing Dental Services Provider including, without limitation, decisions and liaison with the CQC where the CQC has reported non-compliance with standards (but excluding any decisions in relation to the Performers List).
- 8.2 In accordance with paragraph 8.1 above, the ICB must:
  - 8.2.1 ensure regular and effective collaboration with the CQC to ensure that information is shared and discussed in an appropriate and timely manner;
  - 8.2.2 ensure that any risks identified are managed and escalated where necessary;
  - 8.2.3 respond to CQC assessments of Dental Services Providers where improvement is required;
  - 8.2.4 where a Dental Services Provider is placed into special measures, lead a quality summit to ensure the development and monitoring of an appropriate improvement plan (including a communications plan and actions to manage primary care resilience in the locality); and
  - 8.2.5 take appropriate contractual action including (without limitation) in response to CQC findings.

# 9. Maintaining the Performers List

On receiving a notice from a practitioner (who is party to a Dental Services Contract) of an amendment to information recorded about them in the Performers List, pursuant to regulation 9(1) of the National Health Service (Performers Lists) (England) Regulations 2013, the ICB must support NHS England's amendment of the Performers List as soon as possible after receiving the notice using the Primary Care Support services provided by NHS England, insofar as that amendment relates to a change in contractor details.

## 10. Procurement and New Contracts

- 10.1 Until any new arrangements for awarding Dental Services Contracts come into force, the ICB will make procurement decisions relevant to the exercise of the Delegated Functions and in accordance with the detailed arrangements regarding procurement set out in the procurement protocol issued and updated by NHS England from time to time.
- 10.2 In discharging its responsibilities set out in this Schedule 2B, the ICB must comply at all times with Law and all relevant Guidance (including any applicable procurement law and/or guidance on the selection of, and award of contracts to, providers of healthcare services).

- 10.3 On the coming into force of new arrangements for awarding Dental Services Contracts, the ICB will make decisions on awarding new contracts relevant to the exercise of the Delegated Functions.
- 10.4 When the ICB makes decisions in connection with the awarding of Dental Services Contracts it should ensure that it is able to demonstrate compliance with requirements for the award of Dental Services Contracts, including that the decision was:
  - 10.4.1 made in the best interest of patients, taxpayers and the population;
  - 10.4.2 robust and defensible, with conflicts of interests appropriately managed;
  - 10.4.3 made transparently, and
  - 10.4.4 compliant with the rules of the regime as set out in NHS England guidance.

#### 11. Complaints

11.1 The ICB will handle all complaints made in respect of Primary Dental Services in accordance with the Complaints Regulations.

#### 12. Commissioning Ancillary Support Services

- 12.1 The ICB must procure, and undertake the management and monitoring of contracts for the provision of, such ancillary support services as are required to support the ICB in the effective discharge of the Delegated Functions, including, but not limited to the following:
  - 12.1.1 provision of translation and interpretation services; and
  - 12.1.2 occupational health services.

#### Part 2A: General Obligations – Prescribed Dental Services

#### 1. Introduction

- 1.1 This Part 2A of Schedule 2B (*Dental Care Services*) sets out general provisions regarding the carrying out of those Delegated Functions relating to Prescribed Dental Services. Prescribed Dental Services constitute Community Dental Services and Secondary Care Dental Services. These include:
  - 1.1.1 decisions in relation to the commissioning and management of Prescribed Dental Services;
  - 1.1.2 planning Prescribed Dental Services in the Area, including carrying out needs assessments;
  - 1.1.3 undertaking reviews of Prescribed Dental Services in the Area;
  - 1.1.4 management of the Delegated Funds in respect of Prescribed Dental Services in the Area;
  - 1.1.5 co-ordinating a common approach to the commissioning and delivery of Prescribed Dental Services with other health and social care bodies where appropriate; and
  - 1.1.6 such other ancillary activities that are necessary in order to exercise the Delegated Functions.

1.2 For the purposes of this Schedule 2B, "Secondary Care Dental Services" refers to Prescribed Dental Services which are not Community Dental Services.

# 2. General Obligations

- 2.1 The ICB is responsible for commissioning Prescribed Dental Services for its Population which for the purpose of this Part 2A of Schedule 2B (*Dental Care Services*), shall refer to a group of people for whom the ICB has core responsibility, as established under the rules published by NHS England under section 14Z31 of the Act.
- 2.2 In respect of integrated working, the ICB must take an integrated approach to working and co-ordinating with stakeholders including NHS England, Local Dental Professional Networks, Local Authorities, Healthwatch, acute and community providers, the Local Dental Committee, managed clinical networks and other stakeholders.
- 2.3 When planning and commissioning Prescribed Dental Services, the ICB must comply with Mandated Guidance issued by NHS England.
- 2.4 The provisions of Paragraph 2.4, 2.5 and 2.6 of Part 1A of this Schedule 2B shall apply in respect of Prescribed Dental Services as if "Dental Services Contract" includes all contracts for Prescribed Dental Services and "Primary Dental Services" include Prescribed Dental Services.
- 2.5 In awarding any new contract for Prescribed Dental Services, the ICB must:
  - 2.5.1 comply with Law and all relevant Guidance (including any applicable procurement law and/or guidance on the selection of, and award of contracts to, providers of healthcare services);
  - 2.5.2 use the current NHS Standard Contract published by NHS England from time to time; or an appropriate contract for the provision of Dental Care Services made pursuant to NHS England's functions under Part 5 of the NHS Act; and
  - 2.5.3 where the NHS Standard Contract is used, pay for the Services in accordance with the NHS Payment Scheme (as defined in the Health and Social Care Act 2012).

# Part 2B: Specific Obligations – Prescribed Dental Services

# 1. Introduction

1.1 This Part 2B of Schedule 2B (*Prescribed Dental Care Services*) sets out further provision regarding the carrying out of each of the Delegated Functions in relation to Prescribed Dental Services.

# 2. Community Dental Services Commissioning Obligations

- 2.1 Community Dental Services may currently be contracted for by way of either an NHS Standard Contract or a PDS Agreement, as appropriate to the particular service. Accordingly:
  - 2.1.1 where Community Dental Services are commissioned on PDS Agreement terms (or it is appropriate to commission a new agreement for Community Dental Services on a PDS Agreement or other agreement made pursuant to NHS England's functions under Part 5 of the NHS Act), those contracts must be managed in accordance with the relevant provisions of Part 1A and Part 1B of this Schedule 2B as if they were Primary Dental Services for the purposes of that Part. The provisions of this Part 2A of Schedule 2B also apply; and

2.1.2 where Community Dental Services are commissioned on NHS Standard Contract terms, the provisions of this Part 2A of Schedule 2B apply in full.

# 3. Secondary Care Dental Services Commissioning Obligations

- 3.1 For the first financial year following delegation of Secondary Care Dental Services to the ICB (the "Initial Year of Delegation"), the Secondary Care Dental Services shall be commissioned through wider NHS Standard Contracts made between NHS England and the relevant providers that a) cover the whole population of England; and b) typically also cover other services. Accordingly, unless otherwise stated within a Contractual Notice, for the Initial Year of Delegation ONLY the following shall apply:
  - 3.1.1 The commissioning responsibility for the Secondary Care Dental Service elements of the relevant NHS Standard Contracts is delegated to the ICB to the extent that they relate to its Population;
  - 3.1.2 NHS England is, and will remain, the "co-ordinating commissioner" (as defined in the NHS Standard Contract) for those contracts, meaning that NHS England retains core contract management responsibility;
  - 3.1.3 Delegation of commissioning responsibility for the Secondary Care Dental service elements of the relevant NHS Standard Contracts is permitted by clause GC12 of those contracts. NHS England has confirmed these delegation arrangements by letter to each affected provider so that they are aware of the ICB's role as Secondary Care Dental Services commissioner.
  - 3.1.4 whilst the ICB is commissioner of the Secondary Care Dental Service elements of the contract that relate to its Population, it does not have any direct contract management role and must work with NHS England as coordinating commissioner, raising any contractual issues with NHS England for consideration and any appropriate action;
  - 3.1.5 The ICB shall ensure that contractual payments are made to providers for the provision of Secondary Care Dental Services in respect of the ICB's Population, as required by the terms of those contracts. This may represent only a proportion of the overall payment due to the provider for Secondary Care Dental Services delivered more widely under that contract.
- 3.2 For all subsequent financial years following the Initial Year of Delegation the ICB will be responsible for ensuring that appropriate contractual arrangements are in place to ensure continuity of Secondary Care Dental Services for its Population.

#### 4. Prescribed Dental Services Contract Management

- 4.1 Subject to Paragraph 4.2 of this Part 2B of Schedule 2B, the ICB must:
  - 4.1.1 comply with all current and future relevant national Mandated Guidance regarding contract reviews;
  - 4.1.2 monitor contract performance and prescribed care dental spending, with a view in particular to ensuring the delivery of agreed contract activity, and the reallocation of unused resources to meet the oral health needs of the Area;
  - 4.1.3 monitor contract performance with a view in particular to addressing patient safety concerns and promoting patient safety; and
  - 4.1.4 ensure appropriate oversight of the Prescribed Dental Services, including, where appropriate, procuring such ancillary support services as are required for the performance of this function.

4.2 For the Initial Year of Delegation in respect of Secondary Care Dental Services the requirements set out in paragraph 4.1 of this Part 2B of Schedule 2B do not apply and the terms of the relevant Contractual Notice shall apply.

# 5. Transparency and freedom of information

- 5.1 The ICB must:
  - 5.1.1 respond to requests for information from members of the public and the media, including requests made pursuant to the FOIA, whose subject-matter relates to the performance of the Delegated Functions in the ICB's Area; and
  - 5.1.2 provide information and assistance as required to support NHS England in the preparation of responses to parliamentary questions in connection with the Delegated Functions.

#### 6. Planning the Provider Landscape

- 6.1 The ICB must plan the provider landscape in the Area, including considering and taking decisions in relation to:
  - 6.1.1 establishing new providers of Prescribed Dental Services in the Area;
  - 6.1.2 managing providers of Prescribed Dental Services providing inadequate standards of patient care; and
  - 6.1.3 the procurement or award of new contracts for Prescribed Dental Services (in accordance with any procurement protocol or Guidance issued by NHS England from time to time).
- 6.2 In relation to any new contracts for Prescribed Dental Services to be entered into, the ICB must, without prejudice to any obligation in paragraph 12 (*Procurement and New Contracts*):
  - 6.2.1 consider and use the form of contract that will ensure compliance with NHS England's obligations under Law taking into account the persons to whom such contracts may be awarded;
  - 6.2.2 provide to NHS England confirmation as required from time to time that it has considered and complied with its obligations under this Agreement and the Law.

# 7. Staffing and Workforce

7.1 The provisions of paragraph 6 of Part 1B of this Schedule 2B shall apply.

#### 8. Finance

8.1 The ICB must ensure the financial delivery of the Prescribed Dental Services in accordance with any Mandated Guidance provided by NHS England.

#### 9. Integrating dentistry into communities at Primary Care Network level

9.1 The ICB must exercise the Delegated Functions with a view to achieving greater integration of dentists into the Integrated Care System at the Primary Care Network level.

#### 10. Making Decisions in relation to Management of Poorly Performing Dental Services Providers

- 10.1 The ICB must make decisions in relation to the management of poorly performing providers of Prescribed Dental Services and including, without limitation, decisions and liaison with the CQC where the CQC has reported non-compliance with standards.
- 10.2 In accordance with paragraph **Error! Reference source not found.** above, the ICB must:
  - 10.2.1 ensure regular and effective collaboration with the CQC to ensure that information is shared and discussed in an appropriate and timely manner;
  - 10.2.2 ensure that any risks identified are managed and escalated where necessary;
  - 10.2.3 respond to CQC assessments of providers of Prescribed Dental Services where improvement is required;
  - 10.2.4 where a providers of Prescribed Dental Services is placed into special measures, lead a quality summit to ensure the development and monitoring of an appropriate improvement plan (including a communications plan and actions to manage primary care resilience in the locality); and
  - 10.2.5 take appropriate contractual action in response to CQC findings.

#### 11. Maintaining the Performers List

On receiving a notice from a practitioner (who is party to a contract for Prescribed Dental Services) of an amendment to information recorded about them in the Performers List, pursuant to regulation 9(1) of the National Health Service (Performers Lists) (England) Regulations 2013, the ICB must support NHS England's amendment of the Performers List as soon as possible after receiving the notice using the Primary Care Support services provided by NHS England, insofar as that amendment relates to a change in contractor details.

#### 12. Procurement and New Contracts

- 12.1 Until any new arrangements for awarding contracts for Prescribed Dental Services come into force, the ICB will make procurement decisions relevant to the exercise of the Delegated Functions and in accordance with the detailed arrangements regarding procurement set out in the procurement protocol issued and updated by NHS England from time to time.
- 12.2 In discharging its responsibilities set out in this Schedule 2B, the ICB must comply at all times with Law and all relevant Guidance (including any applicable procurement law and/or guidance on the selection of, and award of contracts to, providers of healthcare services).
- 12.3 On the coming into force of new arrangements for awarding contracts for Prescribed Dental Services, the ICB will make decisions on awarding new contracts relevant to the exercise of the Delegated Functions.
- 12.4 When the ICB makes decisions in connection with the awarding of contracts for Prescribed Dental Services it should ensure that it is able to demonstrate compliance with requirements for the award of contracts for Prescribed Dental Services, including that the decision was:
  - 12.4.1 made in the best interest of patients, taxpayers and the population;
  - 12.4.2 robust and defensible, with conflicts of interests appropriately managed;
  - 12.4.3 made transparently, and
12.4.4 compliant with the rules of the regime as set out in NHS England guidance.

## 13. Commissioning Ancillary Support Services

- 13.1 The ICB must procure, and undertake the management and monitoring of contracts for the provision of, such ancillary support services as are required to support the ICB in the effective discharge of the Delegated Functions, including, but not limited to the following:
  - 13.1.1 provision of translation and interpretation services; and
  - 13.1.2 occupational health services.

#### 14. Complaints

14.1 The ICB shall be responsible for handling complaints made in respect of Prescribed Dental Services.

## Schedule 2C: Primary Ophthalmic Services

The provisions of this Schedule 2C form part of this Agreement only where indicated in the Particulars.

#### Part 1: General Obligations

## 1. Introduction

- 1.1 This Part 1 of Schedule 2C (*Primary Ophthalmic Services*) sets out general provisions regarding the carrying out of the Delegated Functions, being, in summary:
  - 1.1.1 decisions in relation to the management of Primary Ophthalmic Services;
  - 1.1.2 undertaking reviews of Primary Ophthalmic Services in the Area;
  - 1.1.3 management of the Delegated Funds in the Area;
  - 1.1.4 co-ordinating a common approach to the commissioning of Primary Ophthalmic Services with other commissioners in the Area where appropriate; and
  - 1.1.5 such other ancillary activities that are necessary in order to exercise the Delegated Functions.

#### 2. General Obligations

- 2.1 The ICB is responsible for managing the provision of Primary Ophthalmic Services.
- 2.2 When carrying out Delegated Functions in respect of Primary Ophthalmic Services, the ICB must comply with all Mandated Guidance issued by NHS England.
- 2.3 The role of the ICB includes identifying and seeking to address any unmet needs which may be met through the delivery of Primary Ophthalmic Services.
- 2.4 In respect of integrated working, the ICB must:
  - 2.4.1 take an integrated approach to working and co-ordinating with stakeholders including NHS England, Local Eye Health Networks, Local Authorities, Healthwatch, acute and community providers, Local Optical Committees, and other stakeholders;
  - 2.4.2 work with NHS England and other ICBs to co-ordinate a common approach to the commissioning of Primary Ophthalmic Services generally; and
  - 2.4.3 work with NHS England to coordinate the exercise of their respective performance management functions.
- 2.5 In relation to the Delegated Functions, the ICB agrees to perform the following general obligations:
  - 2.5.1 to manage the Primary Ophthalmic Services Contracts on behalf of NHS England and perform all of NHS England's obligations under each of the Primary Ophthalmic Services Contracts in accordance with the terms of the Primary Care Contracts as if it were named in the contract in place of NHS England;
  - 2.5.2 working with other organisations, including the NHS Business Services Authority and NHS England as appropriate, actively manage the performance of the Primary Ophthalmic Services Provider in order to secure the needs of people who use the services, improve the quality of services and improve efficiency in the provision of the services including by taking

timely action to enforce contractual breaches, serve notices or provide discretionary support;

- 2.5.3 ensure that it obtains value for money on behalf of NHS England and avoids making any double payments under any Primary Ophthalmic Services Contracts;
- 2.5.4 notify NHS England immediately (or in any event within two (2) Operational Days) of any breach by the ICB of its obligations to perform any of NHS England's obligations under the Primary Ophthalmic Services Contracts;
- 2.5.5 undertake any investigations relating (among other things) to whistleblowing claims, infection control and patient complaints;
- 2.5.6 keep a record of all of the Primary Ophthalmic Services Contracts that the ICB manages on behalf of NHS England setting out the following details in relation to each Primary Ophthalmic Services Contract:
  - 2.5.6.1 name of the Primary Ophthalmic Services Provider;
  - 2.5.6.2 any practice or trading name by which the Primary Ophthalmic Services Provider is known (if different to the name recorded under paragraph **Error! Reference source not found.**);
  - 2.5.6.3 location of provision of services; and
  - 2.5.6.4 amounts payable under the Primary Ophthalmic Services Contract (if a contract sum is payable) or amount payable in respect of each patient (if there is no contract sum).
- 2.6 Without prejudice to clause 9 (*Finance*) or paragraph **Error! Reference source not found.** above, the ICB must actively manage each of the relevant Primary Ophthalmic Services Contracts including by:
  - 2.6.1 managing the relevant Primary Ophthalmic Services Contract, including in respect of quality standards, incentives, observance of service specifications, and monitoring of activity and finance;
  - 2.6.2 assessing quality and outcomes (including clinical effectiveness, patient experience and patient safety);
  - 2.6.3 managing variations to the relevant Primary Ophthalmic Services Contract or services in accordance with national policy, service user needs and clinical developments;
  - 2.6.4 agreeing information and reporting requirements and managing information breaches (which will include use of the NHS Digital Data Security and Protection Toolkit);
  - 2.6.5 conducting review meetings and undertaking contract management including the issuing of contract queries and agreeing any remedial action plan or related contract management processes; and
  - 2.6.6 complying with and implementing any relevant Mandated Guidance issued from time to time.
- 2.7 This paragraph is without prejudice to clause 10 (*Information, Planning and Reporting*) or any other provision in this Agreement. The ICB must provide NHS England with:

- 2.7.1 such information relating to individual providers of Primary Ophthalmic Services in the Area as NHS England may reasonably request, to ensure that NHS England is able to continue to gather national data regarding the commissioning or performances of providers of Primary Ophthalmic Services;
- 2.7.2 such data/data sets as required by NHS England to ensure population of any national dashboards;
- 2.7.3 any other data/data sets as required by NHS England; and
- 2.7.4 the ICB shall procure that providers accurately record and report information so as to allow NHS England and other agencies to discharge their functions.

## Part 2: Specific Obligations

#### 1. Introduction

1.1 This Part 2 of Schedule 2C *(Primary Ophthalmic Services)* sets out further provision regarding the carrying out of each of the Delegated Functions.

## 2. Primary Ophthalmic Services Contract Management

- 2.1 The ICB must:
  - 2.1.1 comply with all current and future relevant national Mandated Guidance regarding General Ophthalmic Contract reviews and any other contract reviews;
  - 2.1.2 take on the responsibility for existing services provided pursuant to a Primary Ophthalmic Services Contract, and for commissioning new services;
  - 2.1.3 assume the responsibility for the award of new Primary Ophthalmic Services Contracts; and
  - 2.1.4 monitor contract performance with a view to achieving assurance and improvement in the delivery of services in the context of the ICB;

in each case acknowledging that the NHS Business Services Authority provides endto-end support services in relation to these functions, as referred to in SCHEDULE 6 (*Mandated Assistance and Support*). The ICB accordingly agrees to co-operate with the NHS Business Services Authority in the delivery of these functions.

#### 3. Transparency and freedom of information

- 3.1 The ICB must:
  - 3.1.1 Respond to requests for information from members and the public and the media, including requests made pursuant to the FOIA, whose subject-matter relates to the performance of the Delegated Functions in the ICB's Area; and
  - 3.1.2 Provide information and assistance as required to support NHS England in the preparation of responses to parliamentary questions in connection with the Delegated Functions.

#### 4. Maintaining the Performers List

4.1 On receiving a notice from a practitioner (who is party to a Primary Ophthalmic Services Contract) of an amendment to information recorded about them in the Performers List,

pursuant to regulation 9(1) of the National Health Service (Performers Lists) (England) Regulations 2013, the ICB must support NHS England's amendment of the performers list as soon as possible after receiving the notice using the Primary Care Support services provided by NHS England, insofar as that amendment relates to a change in contractor details.

## 5. Finance

5.1 Further requirements in respect of finance will be specified in Mandated Guidance.

#### 6. Workforce

- 6.1 The arrangements for the provision and maintenance of sufficient and appropriately qualified, trained and experienced Staff in order for the ICB to fulfil its responsibilities for each of the Delegated Functions ("the Staffing Model"), will be communicated formally to the ICB by NHS England following recommendations made by the National Moderation Panel. Further requirements in respect of workforce will be specified in Mandated Guidance.
- 6.2 The ICB is not permitted to vary the Staffing Model agreed with NHS England as part of its application for delegation of the said functions however a variation can be applied for by the ICB and considered by the National Moderation Panel at any time.

## 7. Integrating optometry into communities at Primary Care Network level

7.1 The ICB must exercise the Delegated Functions with a view to achieving greater integration of optometrists into the Integrated Care System at the Primary Care Network level.

## 8. Complaints

8.1 The ICB will handle complaints made in respect of primary ophthalmic services in accordance with the Complaints Regulations.

#### 9. Commissioning ancillary support services

- 9.1 The ICB must procure, and undertake the management and monitoring of contracts for the provision of, such ancillary support services as are required to support the ICB in the effective discharge of the Delegated Functions, including, but not limited to the following:
  - 9.1.1 provision of translation and interpretation services; and
  - 9.1.2 occupational health services.

## Schedule 2D: Delegated Functions – Pharmaceutical Services

The provisions of this Schedule 2D form part of this Agreement only where indicated in the Particulars.

1. In this Schedule, the following additional definitions shall apply:

Advanced Services	has the meaning given to that term by the Pharmaceutical Regulations;
Conditions of Inclusion	means those conditions set out at Part 9 of the Pharmaceutical Regulations;
Delegated Pharmaceutical Functions	the functions set out at paragraph <b>Error!</b> <b>Reference source not found.</b> of this Schedule;
Designated Commissioner	has the meaning given to that term at paragraph <b>Error! Reference source not found.</b> of this Schedule;
Dispensing Doctor	has the meaning given to that term by the Pharmaceutical Regulations;
Dispensing Doctor Decisions	means decisions made under Part 8 of the Pharmaceutical Regulations;
Dispensing Doctor Lists	has the meaning given to that term by the Pharmaceutical Regulations;
Drug Tariff	has the meaning given to that term by the Pharmaceutical Regulations;
Electronic Prescription Service	has the meaning given to that term by the Pharmaceutical Regulations;
Enhanced Services	has the meaning given to that term by the Pharmaceutical Regulations;
Essential Services	is to be construed in accordance with paragraph 3 of Schedule 4 to the Pharmaceutical Regulations;
Fitness to Practise Functions	has the meaning given to that term at paragraph <b>Error! Reference source not found.</b> of this Schedule;
Locally Commissioned Services	means services which are not Essential Services, Advanced Services, Enhanced Services or services commissioned under an LPS Scheme;
LPS Chemist	has the meaning given to that term by the Pharmaceutical Regulations;
LPS Scheme	has the meaning given to that term by Paragraph 1(2) of Schedule 12 to the NHS Act;
NHS Chemist	has the meaning given to that term by the Pharmaceutical Regulations;
Pharmaceutical Lists	has the meaning given to that term at paragraph <b>Error! Reference source not found.</b> . of this Schedule and any reference to a Pharmaceutical List should be construed accordingly;

Pharmaceutical Regulations	means the National Health Service (Pharmaceutical and Local Pharmaceutical Services) Regulations 2013 and reference to a Regulation refers to a provision of the Pharmaceutical Regulations, unless otherwise stated;
Rurality Decisions	means decisions made under Part 7 of the Pharmaceutical Regulations;
Terms of Service	means the terms upon which, by virtue of the Pharmaceutical Regulations, a person undertakes to provide Pharmaceutical Services;

#### Delegated Pharmaceutical Functions

- 2. Except in so far as they fall within the scope of the Reserved Functions, and subject to paragraphs **Error! Reference source not found.**, **Error! Reference source not found.**, 4 and 5, the ICB agrees to perform the following functions of NHS England in respect of the Area (the "Delegated Pharmaceutical Functions"), in all cases in accordance with relevant Law, Mandated Guidance and other Guidance:
  - 2.1 preparing, maintaining and submitting for publication by NHS England lists of persons, other than medical practitioners or dental practitioners, who have undertaken to provide pharmaceutical services from premises situated within the Area<sup>3</sup>, specifically:
    - 2.1.1 lists of persons who have undertaken to provide pharmaceutical services in particular by way of the provision of drugs;
    - 2.1.2 lists of persons who have undertaken to provide pharmaceutical services only by way of the provision of appliances; and
    - 2.1.3 lists of persons participating in the Electronic Prescription Service<sup>4</sup>

collectively referred to in this Schedule as the "Pharmaceutical Lists". In doing so, it is sufficient for the lists referred to at paragraphs **Error! Reference source not found.** and **Error! Reference source not found.** to include a marker showing which persons are also participating in the Electronic Prescription Service, rather than preparing a separate list for the purposes of paragraph **Error! Reference source not found.** 

- 2.1.4 managing and determining applications by persons for inclusion in a Pharmaceutical List<sup>5</sup>;
- 2.1.5 managing and determining applications by persons included in a Pharmaceutical List;
- 2.1.6 responsibilities for financial resources related to the Delegated Pharmaceutical Functions as described in Mandated Guidance issued by NHS England;
- 2.1.7 overseeing the compliance of those included in the Pharmaceutical Lists with:

<sup>&</sup>lt;sup>3</sup> Including (without limitation) updates to those lists following any removal under regulation 115 of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>4</sup> Regulation 10 of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>5</sup> Schedule 2 of the Pharmaceutical Regulations

- 2.1.7.1 their Terms of Service and identifying and investigating breaches, including possible breaches, of those terms;
- 2.1.7.2 relevant Conditions of Inclusion; and
- 2.1.7.3 requirements of the Community Pharmacy Contractual Framework.
- 2.1.8 exercising powers in respect of Performance Related Sanctions and Market Exit<sup>6</sup>;
- 2.1.9 exercising all other rights, and complying with all other obligations, of NHS England in respect of the Terms of Service and Conditions of Inclusion of those included in the Pharmaceutical Lists;
- 2.1.10 communicating to those included in the Pharmaceutical Lists any announcement made by NHS England modifying Terms of Service of any person included in the Pharmaceutical Lists as a consequence of a disease being, or in anticipation of a disease being imminently:
- 2.1.11 pandemic; and
- 2.1.12 a serious risk or potentially a serious risk to human health<sup>7</sup>;
- 2.1.13 communicating to those included in the Pharmaceutical Lists any other matters which NHS England may require the ICB to communicate from time to time;
- 2.1.14 performing functions in respect of the disqualification of practitioners, and related measures concerning a practitioners inclusion in the Pharmaceutical Lists, set out in Chapter 6 of Part 7 to the NHS Act and the provisions of the Pharmaceutical Regulations made under that Chapter ("the Fitness to Practise Functions");
- 2.1.15 performing functions in respect of enforcement, reviews and appeals relating to the Fitness to Practise Functions<sup>8</sup>;
- 2.1.16 making LPS Schemes<sup>9</sup>, subject to the requirements of paragraph 5;
- 2.1.17 overseeing the compliance of those who are party to Local Pharmaceutical Services Contracts with the terms of those contracts and identifying and investigating breaches, including possible breaches, of the terms of those contracts;
- 2.1.18 exercising all rights, and complying with all obligations, of NHS England under Local Pharmaceutical Services Contracts;
- 2.1.19 determining LPS matters<sup>10</sup> in respect of LPS Schemes;
- 2.1.20 determining Rurality Decisions and other rurality matters<sup>11</sup>;
- 2.1.21 determining Dispensing Doctor Decisions<sup>12</sup>;

<sup>&</sup>lt;sup>6</sup> Part 10 of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>7</sup> Regulation 11(3) of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>8</sup> Part 11 of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>9</sup> Section 134 NHS Act and Part 13 of the Pharmaceutical Regulations.

<sup>&</sup>lt;sup>10</sup> Part 13 of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>11</sup> Part 7 of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>12</sup> Part 8 of the Pharmaceutical Regulations

- 2.1.22 preparing and maintaining Dispensing Doctor Lists<sup>13</sup>;
- 2.1.23 making arrangements for the provision of adequate pharmaceutical service delivery across the ICB area;
- 2.1.24 making arrangements for the delivery of Essential Services, Advanced Services and Enhanced Services;
- 2.1.25 supporting implementation and delivery of all elements of the Community Pharmacy Contractual Framework;
- 2.1.26 consulting with patients, the public and other stakeholders to the extent required by the duty of public involvement and consultation under section 14Z45 of the NHS Act;
- 2.1.27 responding to Appeals to the Secretary of State and First Tier Tribunal in respect of the Delegated Pharmaceutical Functions<sup>14</sup>;
- 2.1.28 responding to Claims in respect of the Delegated Pharmaceutical Functions;
- 2.1.29 recovering overpayments from NHS Chemists, LPS Chemists, Dispensing Doctors and Primary Medical Services Providers<sup>15</sup>;
- 2.1.30 bringing any legal proceedings in respect of the Delegated Pharmaceutical Functions;
- 2.1.31 making any notifications to, and consulting with, third parties in respect of the Delegated Pharmaceutical Functions;
- 2.1.32 recognising one or more Local Pharmaceutical Committees which it considers are representative of Pharmaceutical Services Providers in the ICB's Area and liaising with and consulting such Local Pharmaceutical Committees as required by the Pharmaceutical Regulations;
- 2.1.33 commissioning the provision of NHS Smartcards to Pharmaceutical Services Providers and their staff by registration authorities;
- 2.1.34 making any payments due to NHS Chemists suspended from a Pharmaceutical List in accordance with the determination made by the Secretary of State in respect of such payments; and
- 2.1.35 undertaking any investigations relating (among other things) to whistleblowing claims (relating to a superintendent pharmacist, a director or the operation of a pharmacy contractor), infection control and patient complaints.
- 2.2 Where the Area comprises the areas of two or more Health and Wellbeing Boards in their entirety:
  - 2.2.1 the Delegated Pharmaceutical Functions shall be exercised so as to maintain separately in respect of each Health and Wellbeing Board area:
    - 2.2.1.1 Pharmaceutical Lists in respect of premises in that Health and Wellbeing Board area;

<sup>&</sup>lt;sup>13</sup> Regulation 46 of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>14</sup> Schedule 3 of the Pharmaceutical Regulations

<sup>&</sup>lt;sup>15</sup> Regulation 94 of the Pharmaceutical Regulations

- 2.2.1.2 a list of LPS Chemists providing local pharmaceutical services at or from premises in that Health and Wellbeing Board area<sup>16</sup>; and
- 2.2.1.3 a Dispensing Doctor List (together the "Relevant Lists"); and
- 2.2.1.4 the ICB shall comply with such Contractual Notices as NHS England may issue from time to time concerning the arrangements for the exercise of the Delegated Pharmaceutical Functions across two or more Health and Wellbeing Board areas.
- 2.3 Where the Area comprises part of the area of a Health and Wellbeing Board (the "Relevant Health and Wellbeing Board"):
  - 2.3.1 NHS England shall by Contractual Notice designate:
    - 2.3.1.1 the ICB;
    - 2.3.1.2 another ICB whose area comprises in part the area of the Relevant Health and Wellbeing Board; or
    - 2.3.1.3 NHS England;

as the body responsible for maintaining the Relevant Lists (as defined in paragraph **Error! Reference source not found.** of this Schedule 2D) in respect of the Relevant Health and Wellbeing Board ("the Designated Commissioner");

- 2.3.2 the ICB shall exercise the Delegated Pharmaceutical Functions in respect of that part of the Relevant Health and Wellbeing Board's area that falls within the Area but in doing so shall liaise with any Designated Commissioner for the purposes of maintaining the accuracy of the Relevant Lists (as defined in paragraph **Error! Reference source not found.** of this Schedule 2D) in respect of the Relevant Health and Wellbeing Board; and
- 2.3.3 the ICB shall comply with all Contractual Notices issued by NHS England for the purposes of determining responsibilities in the circumstances described in this paragraph 2.3.

#### Prescribed Support

- 3. Notwithstanding the inclusion of the following within the Delegated Functions, the ICB shall discharge the functions set out at:
  - 3.1 Paragraph 2.1.1 (maintaining Pharmaceutical Lists)
  - 3.2 Paragraph 2.1.2 (managing applications for inclusion)
  - 3.3 Paragraph 2.1.3 (managing applications from those included in a list)
  - 3.4 Paragraph 2.1.5 (overseeing compliance with Terms of Service and Conditions of Inclusion)
  - 3.5 Paragraph 2.1.10 (Fitness to Practise)
  - 3.6 Paragraph 2.1.18 (maintaining and publishing Dispensing Doctors Lists)
  - 3.7 Paragraph 2.1.25 (recovery of overpayments)

<sup>&</sup>lt;sup>16</sup> Regulation 114 of the Pharmaceutical Regulations

with the assistance and support of the NHS Business Services Authority, Primary Care Support England or such other person as NHS England shall designate by Contractual Notice for these purposes from time to time and in accordance with the allocation of operational responsibilities described by NHS England in Mandated Guidance.

#### LPS Schemes

4. The ICB shall not without the prior written consent of NHS England make any new LPS Schemes.

## Barred Persons

5. The ICB must ensure that persons barred from involvement in specific elements of the Delegated Functions are excluded from such involvement in accordance with the Pharmaceutical Regulations.

#### Other Services

6. The provisions of this schedule are without prejudice to the ability of the ICB to make arrangements for the provision of Locally Commissioned Services for the purposes of the NHS in accordance with its own commissioning functions and using its own financial resources.

#### Payments

- 7. In exercising the Delegated Pharmaceutical Functions, the ICB must ensure that:
  - 7.1 all payments to which the Drug Tariff applies are made solely in accordance with the Drug Tariff; and
  - 7.2 any other payments for services (including without limitation those relating to LPS Schemes and Enhanced Services) are made in accordance with recognised contractual mechanisms intended to apply to those services.

#### Flu vaccinations

- 8. The Parties acknowledge and agree that:
  - 8.1 responsibility for arranging any national scheme for flu vaccinations remains with NHS England as part of its Section 7A Functions; and
  - 8.2 where any such national scheme is arranged by NHS England, the ICB is required to commission flu vaccines as Advanced Services. For the purposes of this Agreement, this forms part of the ICB's responsibilities under this Agreement.

#### Integration

- 9. In respect of integrated working, the ICB must:
  - 9.1 take an integrated approach to working and co-ordinating with stakeholders including NHS England, Local Authorities, Healthwatch, acute and community providers, professional representative groups, contractor representative groups and other stakeholders;
  - 9.2 work with NHS England and other ICBs to co-ordinate a common approach to the commissioning of Pharmaceutical Services generally; and
  - 9.3 work with NHS England to coordinate the exercise of their respective performance management functions.

Integrating pharmacy into communities at Primary Care Network level

10. The ICB must exercise the Delegated Functions with a view to achieving greater integration of community pharmacy into the Integrated Care System at the Primary Care Network level including participation in network governance arrangements.

## Complaints

*11.* The ICB will handle complaints made in respect of Pharmaceutical Services and Local Pharmaceutical Services in accordance with the Complaints Regulations.

#### Commissioning ancillary support services

- 12. The ICB must procure, and undertake the management and monitoring of contracts for the provision of, such ancillary support services as are required to support the ICB in the effective discharge of the Delegated Functions, including, but not limited to the following:
  - 12.1 collection and disposal of clinical waste; and
  - 12.2 provision of translation and interpretation services; and
  - 12.3 occupational health services.

## Finance

13. Further requirements in respect of finance will be specified in Mandated Guidance.

#### Workforce

14. Further requirements in respect of workforce will be specified in Mandated Guidance.

## **SCHEDULE 3**

#### **Reserved Functions**

## 25. Introduction

- 25.1 In accordance with clause 8.4 of this Agreement, all functions of NHS England other than those defined as Delegated Functions are Reserved Functions.
- 25.2 This **Error! Reference source not found.** (*Reserved Functions*) sets out further provision regarding the carrying out of the Reserved Functions.
- 25.3 The ICB will work collaboratively with NHS England and will support and assist NHS England to carry out the Reserved Functions.

#### 26. Management of the national performers list

- 26.1 Subject to Paragraph 2.2, NHS England will continue to perform its functions under the National Health Service (Performers Lists) (England) Regulations 2013.
- 26.2 The ICB will carry out administrative tasks in respect of the Performers Lists as described at:
  - 26.2.1 Paragraph 9 of Part 2, Schedule 2A;
  - 26.2.2 Paragraph 9 of Part 2, Schedule 2B; and
  - 26.2.3 Paragraph 6 of Part 2, Schedule 2C.
- 26.3 NHS England's functions in relation to the management of the national performers list include:
  - 26.3.1 considering applications and decision-making in relation to inclusion on the national performers list, inclusion with conditions and refusals;
  - 26.3.2 identifying, managing and supporting primary care performers where concerns arise; and
  - 26.3.3 managing suspension, imposition of conditions and removal from the national performers list.
- 26.4 NHS England may hold local Performance Advisory Group ("PAG") meetings to consider all complaints or concerns that are reported to NHS England in relation to a named performer and NHS England will determine whether an initial investigation is to be carried out.
- 26.5 NHS England may notify the ICB of all relevant PAG meetings at least seven (7) days in advance of such meetings. NHS England may require a representative of the ICB to attend such meetings to discuss any performer concerns and/or quality issues that may impact on individual performer cases.
- 26.6 The ICB must develop a mechanism to ensure that all complaints regarding any named performer are escalated to the Local NHS England Team for review. The ICB will comply with any Mandated Guidance issued by NHS England in relation to the escalation of complaints about a named performer.

#### 27. Management of the revalidation and appraisal process

27.1 NHS England will continue to perform its functions under the Medical Profession (Responsible Officers) Regulations 2010 (as amended by the Medical Profession (Responsible Officers) (Amendment) Regulations 2013).

- 27.2 All functions in relation to GP appraisal and revalidation will remain the responsibility of NHS England, including:
  - 27.2.1 the funding of GP appraisers;
  - 27.2.2 quality assurance of the GP appraisal process; and
  - 27.2.3 the responsible officer network.
- 27.3 Funding to support the GP appraisal is incorporated within the global sum payment to Primary Medical Services Provider.
- 27.4 The ICB must not remove or restrict the payments made to Primary Medical Services Provider in respect of GP appraisal.
- 27.5 Appraisal arrangements in respect of all other primary care practitioner groups shall also be Reserved Functions.

#### 28. Administration of payments and related performers list management activities

- 28.1 NHS England reserves its functions in relation to the administration of payments to individual performers and related performers list management activities under the National Health Service (Performers Lists) (England) Regulations 2013 and other relevant legislation.
- 28.2 NHS England may continue to pay practitioners who are suspended from the national performers list in accordance with relevant determinations made by the Secretary of State.
- 28.3 For the avoidance of doubt, the ICB is responsible for any ad hoc or discretionary payments to Primary Medical Services Providers (including those under section 96 of the NHS Act) in accordance with **Error! Reference source not found.**A (*Delegated Functions*) Part 2 paragraphs 5.1 and 5.1 of this Agreement, including where such payments may be considered a consequence of actions taken under the National Health Service (Performers Lists) (England) Regulations 2013.

#### 29. Section 7A and Capital Expenditure Functions

- 29.1 NHS England retains the Section 7A Functions and will be responsible for taking decisions in relation to the Section 7A Functions.
- 29.2 In accordance with Schedule 10 Part 2, the ICB will provide certain management and/or administrative services to NHS England in relation to the Section 7A Functions.
- 29.3 NHS England retains the Capital Expenditure Functions and will be responsible for taking decisions in relation to the Capital Expenditure Functions.
- 29.4 In accordance with Schedule 10 Part 1, the ICB will provide certain management and/or administrative services to NHS England in relation to the Capital Expenditure Functions.

## 30. Such other ancillary activities that are necessary in order to exercise the Reserved Functions

- 30.1 NHS England will continue to comply with its obligations under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 30.2 The ICB must assist NHS England's controlled drug accountable officer ("CDAO") to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.

- 30.3 The ICB must nominate a relevant senior individual within the ICB (the "ICB CD Lead") to liaise with and assist NHS England to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 30.4 The ICB CD Lead must, in relation to the Delegated Functions:
  - 30.4.1 on request provide NHS England's CDAO with all reasonable assistance in any investigation involving the Delegated Functions;
  - 30.4.2 report all complaints involving controlled drugs to NHS England's CDAO;
  - 30.4.3 report all incidents or other concerns involving the safe use and management of controlled drugs to NHS England's CDAO;
  - 30.4.4 analyse the controlled drug prescribing data available; and
  - 30.4.5 on request supply (or ensure organisations from whom the ICB commissions services involving the regular use of controlled drugs supply) periodic self-declaration and/or self-assessments to NHS England's CDAO.

#### 31. Reserved Functions – Primary Medical Services

- 31.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Medical Services Functions"):
  - 31.1.1 determining the outcomes expected from Primary Medical Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes and the Department of Health and Social Care mandate;
  - 31.1.2 designing and delivering national transformation programmes in support of national priorities;
  - 31.1.3 the negotiation and agreement of matters concerning General Medical Services contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary medical services nationally;
  - 31.1.4 the development of national standard Primary Medical Service contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Medical Services providers;
  - 31.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 31.1.6 the provision of nationally contracted services delivering digital, logistical and support services for Primary Medical Services in England (including but not limited to):
    - 31.1.6.1 Payments;
    - 31.1.6.2 Pensions;
    - 31.1.6.3 Patient Registration;
    - 31.1.6.4 Medical Records;
    - 31.1.6.5 Performer List;
    - 31.1.6.6 Supplies;

- 31.1.6.7 Call and Recall for Cervical screening (CSAS); and
- 31.1.6.8 Pharmacy Market Management.
- 31.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

### 32. Reserved Functions – Primary Dental Services

- 32.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 32.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 32.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 32.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
  - 32.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 32.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
    - 32.1.5.1 Payments;
    - 32.1.5.2 Pensions;
    - 32.1.5.3 Performer List; and
    - 32.1.5.4 Market Management.
- 32.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 33. Reserved Functions – Primary Ophthalmic Services

- 33.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Ophthalmic Functions"):
  - 33.1.1 the Primary Ophthalmic Services Contracts policy and associated documentation;
  - 33.1.2 the negotiation and agreement of matters concerning Primary Ophthalmic Services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Ophthalmic Services nationally; and
  - 33.1.3 the provision of nationally contracted services delivering digital, logistical and support services for Primary Ophthalmic Services in England (including but not limited to):

- 33.1.3.1 Payments;
- 33.1.3.2 Performers List;
- 33.1.3.3 Market Management/Entry; and
- 33.1.3.4 Contract management, assurance and post-payment verification.
- 33.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

## 34. Reserved Functions – Pharmaceutical Services and Local Pharmaceutical Services

- 34.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Pharmaceutical Functions"):
  - 34.1.1 publication of Pharmaceutical Lists;
  - 34.1.2 functions of NHS England as a determining authority in relation to pharmaceutical remuneration under Part 12 of the Pharmaceutical Regulations;
  - 34.1.3 functions in respect of lists of performers of pharmaceutical services and assistants, noting that as at the date of this Agreement regulations for the purposes of these functions have not been made<sup>17</sup>;
  - 34.1.4 the negotiation and agreement of matters concerning NHS pharmaceutical services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Pharmaceutical Services nationally;
  - 34.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 34.1.6 administration of the pharmacist pre-registration training grant scheme.

## 35. Reserved Functions – Primary Dental Services

- 35.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 35.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 35.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 35.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
  - 35.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;

<sup>&</sup>lt;sup>17</sup> Part 7, Chapter 4A of the NHS Act (not currently in force)

- 35.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
  - 35.1.5.1 Payments
  - 35.1.5.2 Pensions
  - 35.1.5.3 Performer List
  - 35.1.5.4 Market Management.
- 35.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 36. Reserved Functions - Prescribed Dental Services

- 36.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Prescribed Dental Services Functions"):
  - 36.1.1 determining the outcomes expected from Prescribed Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 36.1.2 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act the negotiation and agreement of matters concerning those contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 36.1.3 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the development of standard contracts and national contract variations and guidance;
  - 36.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 36.1.5 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the provision of nationally contracted services delivering digital, logistical and support services in England (including but not limited to):
    - 36.1.5.1 Payments
    - 36.1.5.2 Pensions
    - 36.1.5.3 Performer List
    - 36.1.5.4 Market Management.
- 36.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

## SCHEDULE 4

#### Further Information Governance and Sharing Provisions

#### 1. Introduction

- 1.1 The purpose of this
- 1.2 SCHEDULE 3

#### **Reserved Functions**

#### 37. Introduction

- 37.1 In accordance with clause 8.4 of this Agreement, all functions of NHS England other than those defined as Delegated Functions are Reserved Functions.
- 37.2 This **Error! Reference source not found.** (*Reserved Functions*) sets out further provision regarding the carrying out of the Reserved Functions.
- 37.3 The ICB will work collaboratively with NHS England and will support and assist NHS England to carry out the Reserved Functions.

#### 38. Management of the national performers list

- 38.1 Subject to Paragraph 2.2, NHS England will continue to perform its functions under the National Health Service (Performers Lists) (England) Regulations 2013.
- 38.2 The ICB will carry out administrative tasks in respect of the Performers Lists as described at:
  - 38.2.1 Paragraph 9 of Part 2, Schedule 2A;
  - 38.2.2 Paragraph 9 of Part 2, Schedule 2B; and
  - 38.2.3 Paragraph 6 of Part 2, Schedule 2C.
- 38.3 NHS England's functions in relation to the management of the national performers list include:
  - 38.3.1 considering applications and decision-making in relation to inclusion on the national performers list, inclusion with conditions and refusals;
  - 38.3.2 identifying, managing and supporting primary care performers where concerns arise; and
  - 38.3.3 managing suspension, imposition of conditions and removal from the national performers list.
- 38.4 NHS England may hold local Performance Advisory Group ("PAG") meetings to consider all complaints or concerns that are reported to NHS England in relation to a named performer and NHS England will determine whether an initial investigation is to be carried out.
- 38.5 NHS England may notify the ICB of all relevant PAG meetings at least seven (7) days in advance of such meetings. NHS England may require a representative of the ICB to attend such meetings to discuss any performer concerns and/or quality issues that may impact on individual performer cases.

38.6 The ICB must develop a mechanism to ensure that all complaints regarding any named performer are escalated to the Local NHS England Team for review. The ICB will comply with any Mandated Guidance issued by NHS England in relation to the escalation of complaints about a named performer.

#### **39.** Management of the revalidation and appraisal process

- 39.1 NHS England will continue to perform its functions under the Medical Profession (Responsible Officers) Regulations 2010 (as amended by the Medical Profession (Responsible Officers) (Amendment) Regulations 2013).
- 39.2 All functions in relation to GP appraisal and revalidation will remain the responsibility of NHS England, including:
  - 39.2.1 the funding of GP appraisers;
  - 39.2.2 quality assurance of the GP appraisal process; and
  - 39.2.3 the responsible officer network.
- 39.3 Funding to support the GP appraisal is incorporated within the global sum payment to Primary Medical Services Provider.
- 39.4 The ICB must not remove or restrict the payments made to Primary Medical Services Provider in respect of GP appraisal.
- 39.5 Appraisal arrangements in respect of all other primary care practitioner groups shall also be Reserved Functions.

#### 40. Administration of payments and related performers list management activities

- 40.1 NHS England reserves its functions in relation to the administration of payments to individual performers and related performers list management activities under the National Health Service (Performers Lists) (England) Regulations 2013 and other relevant legislation.
- 40.2 NHS England may continue to pay practitioners who are suspended from the national performers list in accordance with relevant determinations made by the Secretary of State.
- 40.3 For the avoidance of doubt, the ICB is responsible for any ad hoc or discretionary payments to Primary Medical Services Providers (including those under section 96 of the NHS Act) in accordance with **Error! Reference source not found.**A (*Delegated Functions*) Part 2 paragraphs 5.1 and 5.1 of this Agreement, including where such payments may be considered a consequence of actions taken under the National Health Service (Performers Lists) (England) Regulations 2013.

#### 41. Section 7A and Capital Expenditure Functions

- 41.1 NHS England retains the Section 7A Functions and will be responsible for taking decisions in relation to the Section 7A Functions.
- 41.2 In accordance with Schedule 10 Part 2, the ICB will provide certain management and/or administrative services to NHS England in relation to the Section 7A Functions.
- 41.3 NHS England retains the Capital Expenditure Functions and will be responsible for taking decisions in relation to the Capital Expenditure Functions.
- 41.4 In accordance with Schedule 10 Part 1, the ICB will provide certain management and/or administrative services to NHS England in relation to the Capital Expenditure Functions.

# 42. Such other ancillary activities that are necessary in order to exercise the Reserved Functions

- 42.1 NHS England will continue to comply with its obligations under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 42.2 The ICB must assist NHS England's controlled drug accountable officer ("CDAO") to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 42.3 The ICB must nominate a relevant senior individual within the ICB (the "ICB CD Lead") to liaise with and assist NHS England to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 42.4 The ICB CD Lead must, in relation to the Delegated Functions:
  - 42.4.1 on request provide NHS England's CDAO with all reasonable assistance in any investigation involving the Delegated Functions;
  - 42.4.2 report all complaints involving controlled drugs to NHS England's CDAO;
  - 42.4.3 report all incidents or other concerns involving the safe use and management of controlled drugs to NHS England's CDAO;
  - 42.4.4 analyse the controlled drug prescribing data available; and
  - 42.4.5 on request supply (or ensure organisations from whom the ICB commissions services involving the regular use of controlled drugs supply) periodic self-declaration and/or self-assessments to NHS England's CDAO.

#### 43. Reserved Functions – Primary Medical Services

- 43.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Medical Services Functions"):
  - 43.1.1 determining the outcomes expected from Primary Medical Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes and the Department of Health and Social Care mandate;
  - 43.1.2 designing and delivering national transformation programmes in support of national priorities;
  - 43.1.3 the negotiation and agreement of matters concerning General Medical Services contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary medical services nationally;
  - 43.1.4 the development of national standard Primary Medical Service contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Medical Services providers;
  - 43.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 43.1.6 the provision of nationally contracted services delivering digital, logistical and support services for Primary Medical Services in England (including but not limited to):
    - 43.1.6.1 Payments;

43.1.6.2	Pensions;
43.1.6.3	Patient Registration;
43.1.6.4	Medical Records;
43.1.6.5	Performer List;
43.1.6.6	Supplies;

- 43.1.6.7 Call and Recall for Cervical screening (CSAS); and
- 43.1.6.8 Pharmacy Market Management.
- 43.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 44. Reserved Functions – Primary Dental Services

- 44.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 44.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 44.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 44.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
  - 44.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 44.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
    - 44.1.5.1 Payments;
    - 44.1.5.2 Pensions;
    - 44.1.5.3 Performer List; and
    - 44.1.5.4 Market Management.
- 44.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 45. Reserved Functions – Primary Ophthalmic Services

- 45.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Ophthalmic Functions"):
  - 45.1.1 the Primary Ophthalmic Services Contracts policy and associated documentation;

- 45.1.2 the negotiation and agreement of matters concerning Primary Ophthalmic Services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Ophthalmic Services nationally; and
- 45.1.3 the provision of nationally contracted services delivering digital, logistical and support services for Primary Ophthalmic Services in England (including but not limited to):
  - 45.1.3.1 Payments;
  - 45.1.3.2 Performers List;
  - 45.1.3.3 Market Management/Entry; and
  - 45.1.3.4 Contract management, assurance and post-payment verification.
- 45.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

## 46. Reserved Functions – Pharmaceutical Services and Local Pharmaceutical Services

- 46.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Pharmaceutical Functions"):
  - 46.1.1 publication of Pharmaceutical Lists;
  - 46.1.2 functions of NHS England as a determining authority in relation to pharmaceutical remuneration under Part 12 of the Pharmaceutical Regulations;
  - 46.1.3 functions in respect of lists of performers of pharmaceutical services and assistants, noting that as at the date of this Agreement regulations for the purposes of these functions have not been made;
  - 46.1.4 the negotiation and agreement of matters concerning NHS pharmaceutical services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Pharmaceutical Services nationally;
  - 46.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 46.1.6 administration of the pharmacist pre-registration training grant scheme.

#### 47. Reserved Functions – Primary Dental Services

- 47.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 47.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 47.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;

- 47.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
- 47.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
- 47.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
  - 47.1.5.1 Payments
  - 47.1.5.2 Pensions
  - 47.1.5.3 Performer List
  - 47.1.5.4 Market Management.
- 47.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 48. Reserved Functions - Prescribed Dental Services

- 48.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Prescribed Dental Services Functions"):
  - 48.1.1 determining the outcomes expected from Prescribed Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 48.1.2 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act the negotiation and agreement of matters concerning those contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 48.1.3 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the development of standard contracts and national contract variations and guidance;
  - 48.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 48.1.5 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the provision of nationally contracted services delivering digital, logistical and support services in England (including but not limited to):
    - 48.1.5.1 Payments
    - 48.1.5.2 Pensions
    - 48.1.5.3 Performer List
    - 48.1.5.4 Market Management.
- 48.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

- 1.3 **SCHEDULE 4** (*Further Information Governance and Sharing Provisions* is to set out the scope for the secure and confidential sharing of information between the Parties on a **Need To Know** basis, in order to enable the Parties to exercise their functions in pursuance of this Agreement.
- 1.4 References in this
- 1.5 SCHEDULE 3

#### **Reserved Functions**

#### 49. Introduction

- 49.1 In accordance with clause 8.4 of this Agreement, all functions of NHS England other than those defined as Delegated Functions are Reserved Functions.
- 49.2 This **Error! Reference source not found.** (*Reserved Functions*) sets out further provision regarding the carrying out of the Reserved Functions.
- 49.3 The ICB will work collaboratively with NHS England and will support and assist NHS England to carry out the Reserved Functions.

## 50. Management of the national performers list

- 50.1 Subject to Paragraph 2.2, NHS England will continue to perform its functions under the National Health Service (Performers Lists) (England) Regulations 2013.
- 50.2 The ICB will carry out administrative tasks in respect of the Performers Lists as described at:
  - 50.2.1 Paragraph 9 of Part 2, Schedule 2A;
  - 50.2.2 Paragraph 9 of Part 2, Schedule 2B; and
  - 50.2.3 Paragraph 6 of Part 2, Schedule 2C.
- 50.3 NHS England's functions in relation to the management of the national performers list include:
  - 50.3.1 considering applications and decision-making in relation to inclusion on the national performers list, inclusion with conditions and refusals;
  - 50.3.2 identifying, managing and supporting primary care performers where concerns arise; and
  - 50.3.3 managing suspension, imposition of conditions and removal from the national performers list.
- 50.4 NHS England may hold local Performance Advisory Group ("PAG") meetings to consider all complaints or concerns that are reported to NHS England in relation to a named performer and NHS England will determine whether an initial investigation is to be carried out.
- 50.5 NHS England may notify the ICB of all relevant PAG meetings at least seven (7) days in advance of such meetings. NHS England may require a representative of the ICB to attend such meetings to discuss any performer concerns and/or quality issues that may impact on individual performer cases.

50.6 The ICB must develop a mechanism to ensure that all complaints regarding any named performer are escalated to the Local NHS England Team for review. The ICB will comply with any Mandated Guidance issued by NHS England in relation to the escalation of complaints about a named performer.

#### 51. Management of the revalidation and appraisal process

- 51.1 NHS England will continue to perform its functions under the Medical Profession (Responsible Officers) Regulations 2010 (as amended by the Medical Profession (Responsible Officers) (Amendment) Regulations 2013).
- 51.2 All functions in relation to GP appraisal and revalidation will remain the responsibility of NHS England, including:
  - 51.2.1 the funding of GP appraisers;
  - 51.2.2 quality assurance of the GP appraisal process; and
  - 51.2.3 the responsible officer network.
- 51.3 Funding to support the GP appraisal is incorporated within the global sum payment to Primary Medical Services Provider.
- 51.4 The ICB must not remove or restrict the payments made to Primary Medical Services Provider in respect of GP appraisal.
- 51.5 Appraisal arrangements in respect of all other primary care practitioner groups shall also be Reserved Functions.

#### 52. Administration of payments and related performers list management activities

- 52.1 NHS England reserves its functions in relation to the administration of payments to individual performers and related performers list management activities under the National Health Service (Performers Lists) (England) Regulations 2013 and other relevant legislation.
- 52.2 NHS England may continue to pay practitioners who are suspended from the national performers list in accordance with relevant determinations made by the Secretary of State.
- 52.3 For the avoidance of doubt, the ICB is responsible for any ad hoc or discretionary payments to Primary Medical Services Providers (including those under section 96 of the NHS Act) in accordance with **Error! Reference source not found.**A (*Delegated Functions*) Part 2 paragraphs 5.1 and 5.1 of this Agreement, including where such payments may be considered a consequence of actions taken under the National Health Service (Performers Lists) (England) Regulations 2013.

#### 53. Section 7A and Capital Expenditure Functions

- 53.1 NHS England retains the Section 7A Functions and will be responsible for taking decisions in relation to the Section 7A Functions.
- 53.2 In accordance with Schedule 10 Part 2, the ICB will provide certain management and/or administrative services to NHS England in relation to the Section 7A Functions.
- 53.3 NHS England retains the Capital Expenditure Functions and will be responsible for taking decisions in relation to the Capital Expenditure Functions.
- 53.4 In accordance with Schedule 10 Part 1, the ICB will provide certain management and/or administrative services to NHS England in relation to the Capital Expenditure Functions.

# 54. Such other ancillary activities that are necessary in order to exercise the Reserved Functions

- 54.1 NHS England will continue to comply with its obligations under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 54.2 The ICB must assist NHS England's controlled drug accountable officer ("CDAO") to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 54.3 The ICB must nominate a relevant senior individual within the ICB (the "ICB CD Lead") to liaise with and assist NHS England to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 54.4 The ICB CD Lead must, in relation to the Delegated Functions:
  - 54.4.1 on request provide NHS England's CDAO with all reasonable assistance in any investigation involving the Delegated Functions;
  - 54.4.2 report all complaints involving controlled drugs to NHS England's CDAO;
  - 54.4.3 report all incidents or other concerns involving the safe use and management of controlled drugs to NHS England's CDAO;
  - 54.4.4 analyse the controlled drug prescribing data available; and
  - 54.4.5 on request supply (or ensure organisations from whom the ICB commissions services involving the regular use of controlled drugs supply) periodic self-declaration and/or self-assessments to NHS England's CDAO.

#### 55. Reserved Functions – Primary Medical Services

- 55.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Medical Services Functions"):
  - 55.1.1 determining the outcomes expected from Primary Medical Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes and the Department of Health and Social Care mandate;
  - 55.1.2 designing and delivering national transformation programmes in support of national priorities;
  - 55.1.3 the negotiation and agreement of matters concerning General Medical Services contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary medical services nationally;
  - 55.1.4 the development of national standard Primary Medical Service contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Medical Services providers;
  - 55.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 55.1.6 the provision of nationally contracted services delivering digital, logistical and support services for Primary Medical Services in England (including but not limited to):
    - 55.1.6.1 Payments;

55.1.6.2	Pensions;
55.1.6.3	Patient Registration;
55.1.6.4	Medical Records;
55.1.6.5	Performer List;
55.1.6.6	Supplies;

- 55.1.6.7 Call and Recall for Cervical screening (CSAS); and
- 55.1.6.8 Pharmacy Market Management.
- 55.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 56. Reserved Functions – Primary Dental Services

- 56.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 56.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 56.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 56.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
  - 56.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 56.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
    - 56.1.5.1 Payments;
    - 56.1.5.2 Pensions;
    - 56.1.5.3 Performer List; and
    - 56.1.5.4 Market Management.
- 56.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 57. Reserved Functions – Primary Ophthalmic Services

- 57.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Ophthalmic Functions"):
  - 57.1.1 the Primary Ophthalmic Services Contracts policy and associated documentation;

- 57.1.2 the negotiation and agreement of matters concerning Primary Ophthalmic Services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Ophthalmic Services nationally; and
- 57.1.3 the provision of nationally contracted services delivering digital, logistical and support services for Primary Ophthalmic Services in England (including but not limited to):
  - 57.1.3.1 Payments;
  - 57.1.3.2 Performers List;
  - 57.1.3.3 Market Management/Entry; and
  - 57.1.3.4 Contract management, assurance and post-payment verification.
- 57.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

## 58. Reserved Functions – Pharmaceutical Services and Local Pharmaceutical Services

- 58.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Pharmaceutical Functions"):
  - 58.1.1 publication of Pharmaceutical Lists;
  - 58.1.2 functions of NHS England as a determining authority in relation to pharmaceutical remuneration under Part 12 of the Pharmaceutical Regulations;
  - 58.1.3 functions in respect of lists of performers of pharmaceutical services and assistants, noting that as at the date of this Agreement regulations for the purposes of these functions have not been made;
  - 58.1.4 the negotiation and agreement of matters concerning NHS pharmaceutical services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Pharmaceutical Services nationally;
  - 58.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
  - 58.1.6 administration of the pharmacist pre-registration training grant scheme.

#### 59. Reserved Functions – Primary Dental Services

- 59.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 59.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 59.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;

- 59.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
- 59.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
- 59.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
  - 59.1.5.1 Payments
  - 59.1.5.2 Pensions
  - 59.1.5.3 Performer List
  - 59.1.5.4 Market Management.
- 59.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

## 60. Reserved Functions - Prescribed Dental Services

- 60.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Prescribed Dental Services Functions"):
  - 60.1.1 determining the outcomes expected from Prescribed Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 60.1.2 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act the negotiation and agreement of matters concerning those contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 60.1.3 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the development of standard contracts and national contract variations and guidance;
  - 60.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 60.1.5 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the provision of nationally contracted services delivering digital, logistical and support services in England (including but not limited to):
    - 60.1.5.1 Payments
    - 60.1.5.2 Pensions
    - 60.1.5.3 Performer List
    - 60.1.5.4 Market Management.
- 60.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

- 1.6 **SCHEDULE 4** (*Further Information Governance and Sharing Provisions*) to the **Need to Know** basis or requirement (as the context requires) should be taken to mean that the Data Controllers' personnel will only have access to Personal Data or Special Category Personal Data if it is lawful for such personnel to have access to such data for the Specified Purpose in paragraph 2.1 and the function they are required to fulfil at that particular time, in relation to the Specified Purpose, cannot be achieved without access to the Personal Data or Special Category Personal Data specified.
- 1.7 This Schedule and the Data Sharing Agreements entered into under this Schedule are designed to:
  - 1.7.1 provide information about the reasons why Relevant Information may need to be shared and how this will be managed and controlled by the Parties;
  - 1.7.2 describe the purposes for which the Parties have agreed to share Relevant Information;
  - 1.7.3 set out the lawful basis for the sharing of information between the Parties, and the principles that underpin the exchange of Relevant Information;
  - 1.7.4 describe roles and structures to support the exchange of Relevant Information between the Parties;
  - 1.7.5 apply to the sharing of Relevant Information relating to Delegated Functions in respect of
    - 1.7.5.1 Primary Care Providers and Primary Care Provider Personnel; and
    - 1.7.5.2 Dental Services Providers and their personnel;
    - 1.7.5.3 All other providers of Delegated Functions.
  - 1.7.6 apply to the sharing of Relevant Information whatever the medium in which it is held and however it is transmitted;
  - 1.7.7 ensure that Data Subjects are, where appropriate, informed of the reasons why Personal Data about them may need to be shared and how this sharing will be managed;
  - 1.7.8 apply to the activities of the Parties' personnel; and
  - 1.7.9 describe how complaints relating to Personal Data sharing between the Parties will be investigated and resolved, and how the information sharing will be monitored and reviewed.

## 2. Purpose

- 2.1 The Specified Purpose of the data sharing is to facilitate the exercise of the ICB's Delegated Functions and NHS England's Reserved Functions as described in this Agreement.
- 2.2 ICBs must ensure that they have in place appropriate Data Sharing Agreements to enable data to be received by it from NHS Digital (or the successor to the relevant statutory functions of NHS Digital) and any other third party organisations from which the ICB must obtain data for the purpose of exercising the Delegated Functions. Specific and detailed purposes must be set out the Data sharing Agreement that complies with all relevant Legislation and Guidance.

## 3. Benefits of information sharing

3.1 The benefits of sharing information are the achievement of the Specified Purpose set out above, with benefits for service users and other stakeholders in terms of the improved delivery of the NHS services to which this Agreement relates.

## 4. Lawful basis for Sharing

- 4.1 Each Party shall comply with all relevant Information Law requirements and good practice in relation to the processing of Relevant Information shared further to this Agreement.
- 4.2 The ICB shall ensure that there is a Data Protection Impact Assessment ("DPIA") that covers all Delegated Functions. The ICB shall identify the lawful basis for sharing Relevant Information for each purpose and data flow, and where appropriate, enter into a Data Sharing Agreement.

## 5. Relevant Information to be shared

5.1 The Relevant Information to be shared shall be set out in a Data Sharing Agreement.

## 6. Restrictions on use of the Shared Information

- 6.1 Each Party shall only process the Relevant Information as is necessary to achieve the Specified Purpose, and, in particular, shall not use or process Relevant Information for any other purpose unless agreed in writing by the Data Controller that released the information to the other. There shall be no other use or onward transmission of the Relevant Information to any third party without a lawful basis first being determined, and the originating Data Controller being notified.
- 6.2 Access to, and processing of, the Relevant Information provided by a Party must be the minimum necessary to achieve the Specified Purpose. Information and Special Category Personal Data will be handled at all times on a restricted basis, in compliance with Information Law requirements, and the parties' personnel should only have access to Personal Data on a justifiable **Need to Know** basis.
- 6.3 Neither the provisions of this
- 6.4 SCHEDULE 3

## **Reserved Functions**

## 61. Introduction

- 61.1 In accordance with clause 8.4 of this Agreement, all functions of NHS England other than those defined as Delegated Functions are Reserved Functions.
- 61.2 This **Error! Reference source not found.** (*Reserved Functions*) sets out further provision regarding the carrying out of the Reserved Functions.
- 61.3 The ICB will work collaboratively with NHS England and will support and assist NHS England to carry out the Reserved Functions.

## 62. Management of the national performers list

- 62.1 Subject to Paragraph 2.2, NHS England will continue to perform its functions under the National Health Service (Performers Lists) (England) Regulations 2013.
- 62.2 The ICB will carry out administrative tasks in respect of the Performers Lists as described at:

- 62.2.1 Paragraph 9 of Part 2, Schedule 2A;
- 62.2.2 Paragraph 9 of Part 2, Schedule 2B; and
- 62.2.3 Paragraph 6 of Part 2, Schedule 2C.
- 62.3 NHS England's functions in relation to the management of the national performers list include:
  - 62.3.1 considering applications and decision-making in relation to inclusion on the national performers list, inclusion with conditions and refusals;
  - 62.3.2 identifying, managing and supporting primary care performers where concerns arise; and
  - 62.3.3 managing suspension, imposition of conditions and removal from the national performers list.
- 62.4 NHS England may hold local Performance Advisory Group ("PAG") meetings to consider all complaints or concerns that are reported to NHS England in relation to a named performer and NHS England will determine whether an initial investigation is to be carried out.
- 62.5 NHS England may notify the ICB of all relevant PAG meetings at least seven (7) days in advance of such meetings. NHS England may require a representative of the ICB to attend such meetings to discuss any performer concerns and/or quality issues that may impact on individual performer cases.
- 62.6 The ICB must develop a mechanism to ensure that all complaints regarding any named performer are escalated to the Local NHS England Team for review. The ICB will comply with any Mandated Guidance issued by NHS England in relation to the escalation of complaints about a named performer.

## 63. Management of the revalidation and appraisal process

- 63.1 NHS England will continue to perform its functions under the Medical Profession (Responsible Officers) Regulations 2010 (as amended by the Medical Profession (Responsible Officers) (Amendment) Regulations 2013).
- 63.2 All functions in relation to GP appraisal and revalidation will remain the responsibility of NHS England, including:
  - 63.2.1 the funding of GP appraisers;
  - 63.2.2 quality assurance of the GP appraisal process; and
  - 63.2.3 the responsible officer network.
- 63.3 Funding to support the GP appraisal is incorporated within the global sum payment to Primary Medical Services Provider.
- 63.4 The ICB must not remove or restrict the payments made to Primary Medical Services Provider in respect of GP appraisal.
- 63.5 Appraisal arrangements in respect of all other primary care practitioner groups shall also be Reserved Functions.

#### 64. Administration of payments and related performers list management activities

64.1 NHS England reserves its functions in relation to the administration of payments to individual performers and related performers list management activities under the

National Health Service (Performers Lists) (England) Regulations 2013 and other relevant legislation.

- 64.2 NHS England may continue to pay practitioners who are suspended from the national performers list in accordance with relevant determinations made by the Secretary of State.
- 64.3 For the avoidance of doubt, the ICB is responsible for any ad hoc or discretionary payments to Primary Medical Services Providers (including those under section 96 of the NHS Act) in accordance with **Error! Reference source not found.**A (*Delegated Functions*) Part 2 paragraphs 5.1 and 5.1 of this Agreement, including where such payments may be considered a consequence of actions taken under the National Health Service (Performers Lists) (England) Regulations 2013.

#### 65. Section 7A and Capital Expenditure Functions

- 65.1 NHS England retains the Section 7A Functions and will be responsible for taking decisions in relation to the Section 7A Functions.
- 65.2 In accordance with Schedule 10 Part 2, the ICB will provide certain management and/or administrative services to NHS England in relation to the Section 7A Functions.
- 65.3 NHS England retains the Capital Expenditure Functions and will be responsible for taking decisions in relation to the Capital Expenditure Functions.
- 65.4 In accordance with Schedule 10 Part 1, the ICB will provide certain management and/or administrative services to NHS England in relation to the Capital Expenditure Functions.

## 66. Such other ancillary activities that are necessary in order to exercise the Reserved Functions

- 66.1 NHS England will continue to comply with its obligations under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 66.2 The ICB must assist NHS England's controlled drug accountable officer ("CDAO") to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 66.3 The ICB must nominate a relevant senior individual within the ICB (the "ICB CD Lead") to liaise with and assist NHS England to carry out its functions under the Controlled Drugs (Supervision of Management and Use) Regulations 2013.
- 66.4 The ICB CD Lead must, in relation to the Delegated Functions:
  - 66.4.1 on request provide NHS England's CDAO with all reasonable assistance in any investigation involving the Delegated Functions;
  - 66.4.2 report all complaints involving controlled drugs to NHS England's CDAO;
  - 66.4.3 report all incidents or other concerns involving the safe use and management of controlled drugs to NHS England's CDAO;
  - analyse the controlled drug prescribing data available; and
  - 66.4.5 on request supply (or ensure organisations from whom the ICB commissions services involving the regular use of controlled drugs supply) periodic self-declaration and/or self-assessments to NHS England's CDAO.

#### 67. Reserved Functions – Primary Medical Services
- 67.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Medical Services Functions"):
  - 67.1.1 determining the outcomes expected from Primary Medical Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes and the Department of Health and Social Care mandate;
  - 67.1.2 designing and delivering national transformation programmes in support of national priorities;
  - 67.1.3 the negotiation and agreement of matters concerning General Medical Services contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary medical services nationally;
  - 67.1.4 the development of national standard Primary Medical Service contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Medical Services providers;
  - 67.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 67.1.6 the provision of nationally contracted services delivering digital, logistical and support services for Primary Medical Services in England (including but not limited to):
    - 67.1.6.1 Payments;
    - 67.1.6.2 Pensions;
    - 67.1.6.3 Patient Registration;
    - 67.1.6.4 Medical Records;
    - 67.1.6.5 Performer List;
    - 67.1.6.6 Supplies;
    - 67.1.6.7 Call and Recall for Cervical screening (CSAS); and
    - 67.1.6.8 Pharmacy Market Management.
- 67.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 68. Reserved Functions – Primary Dental Services

- 68.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 68.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 68.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the

Department of Health and Social Care and bodies representing providers of primary dental services nationally;

- 68.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
- 68.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
- 68.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
  - 68.1.5.1 Payments;
  - 68.1.5.2 Pensions;
  - 68.1.5.3 Performer List; and
  - 68.1.5.4 Market Management.
- 68.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

#### 69. Reserved Functions – Primary Ophthalmic Services

- 69.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Ophthalmic Functions"):
  - 69.1.1 the Primary Ophthalmic Services Contracts policy and associated documentation;
  - 69.1.2 the negotiation and agreement of matters concerning Primary Ophthalmic Services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Ophthalmic Services nationally; and
  - 69.1.3 the provision of nationally contracted services delivering digital, logistical and support services for Primary Ophthalmic Services in England (including but not limited to):
    - 69.1.3.1 Payments;
    - 69.1.3.2 Performers List;
    - 69.1.3.3 Market Management/Entry; and
    - 69.1.3.4 Contract management, assurance and post-payment verification.
- 69.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

## 70. Reserved Functions – Pharmaceutical Services and Local Pharmaceutical Services

- 70.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Pharmaceutical Functions"):
  - 70.1.1 publication of Pharmaceutical Lists;

- 70.1.2 functions of NHS England as a determining authority in relation to pharmaceutical remuneration under Part 12 of the Pharmaceutical Regulations;
- 70.1.3 functions in respect of lists of performers of pharmaceutical services and assistants, noting that as at the date of this Agreement regulations for the purposes of these functions have not been made;
- 70.1.4 the negotiation and agreement of matters concerning NHS pharmaceutical services with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of Pharmaceutical Services nationally;
- 70.1.5 the provision of commissioning and contracting policy and guidance to support ICBs to meet their delegated duties; and
- 70.1.6 administration of the pharmacist pre-registration training grant scheme.

### 71. Reserved Functions – Primary Dental Services

- 71.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Primary Dental Services Functions"):
  - 71.1.1 determining the outcomes expected from Primary Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
  - 71.1.2 the negotiation and agreement of matters concerning Dental Services Contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
  - 71.1.3 the development of national standard Dental Service Contracts and national contract variations and guidance to ensure an equitable approach to applying nationally agreed changes to all Primary Dental Services providers;
  - 71.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
  - 71.1.5 the provision of nationally contracted services delivering digital, logistical and support services for Primary Dental Services in England (including but not limited to):
    - 71.1.5.1 Payments
    - 71.1.5.2 Pensions
    - 71.1.5.3 Performer List
    - 71.1.5.4 Market Management.
- 71.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

## 72. Reserved Functions - Prescribed Dental Services

72.1 The following functions and related activities shall continue to be exercised by NHS England (the "Reserved Prescribed Dental Services Functions"):

- 72.1.1 determining the outcomes expected from Prescribed Dental Services and the main characteristics of high quality services, taking into account national priorities for improving NHS outcomes; designing and delivering national transformation programmes in line with any applicable commissioning policies and guidance;
- 72.1.2 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act the negotiation and agreement of matters concerning those contracts with national stakeholders including, without limitation, the Department of Health and Social Care and bodies representing providers of primary dental services nationally;
- 72.1.3 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the development of standard contracts and national contract variations and guidance;
- 72.1.4 the provision of all dental commissioning and contracting policy and guidance to support ICBs to meet their delegated duties;
- 72.1.5 in respect of any Prescribed Dental Services contracted pursuant to NHS England's functions under Part 5 of the NHS Act, the provision of nationally contracted services delivering digital, logistical and support services in England (including but not limited to):
  - 72.1.5.1 Payments
  - 72.1.5.2 Pensions
  - 72.1.5.3 Performer List
  - 72.1.5.4 Market Management.
- 72.2 The ICB will work collaboratively with NHS England, and will support and assist those nationally contracted services to carry out their services.

- 6.5 **SCHEDULE 4** (*Further Information Governance and Sharing Provisions*) nor any Data Sharing Agreements entered into in accordance with this Schedule should be taken to permit unrestricted access to data held by any Party. It lays the parameters for the safe and secure sharing and processing of information on a justifiable **Need to Know** basis.
- 6.6 Neither Party shall subcontract any processing of the Relevant Information without the prior written consent of the other Party. Where a Party subcontracts its obligations, it shall do so only by way of a written agreement with the sub-contractor which imposes the same obligations as are imposed on the Data Controllers under this Agreement.
- 6.7 Neither Party shall cause or allow Data to be transferred to any territory outside the United Kingdom without the prior written permission of the responsible Data Controller.
- 6.8 Any particular restrictions on use of certain Relevant Information are included in the Personal Data Agreement annexed to this Schedule.

## 7. Ensuring fairness to the Data Subject

- 7.1 In addition to having a lawful basis for sharing information, the UK GDPR generally requires that the sharing must be fair and transparent. In order to achieve fairness and transparency to the Data Subjects, the Parties will take the following measures:
  - 7.1.1 amendment of internal guidance to improve awareness and understanding among personnel;
  - 7.1.2 amendment of respective privacy notices and policies to reflect the processing of data carried out further to this Agreement, including covering the requirements of articles 13 and 14 UK GDPR and providing these (or making them available to) Data Subjects;
  - 7.1.3 ensuring that information and communications relating to the processing of data is easily accessible and easy to understand, and that clear and plain language be used; and
  - 7.1.4 giving consideration to carrying out activities to promote public understanding of how data is processed where appropriate.
- 7.2 Each Party shall procure that its notification to the Information Commissioner's Office and record of processing maintained for the purposes of Article 30 UK GDPR reflects the flows of information under this Agreement.
- 7.3 Each Party shall reasonably cooperate with the other in undertaking any Data Protection Impact Assessment associated with the processing of data further to this Agreement, and in doing so engage with their respective Data Protection Officers in the performance by them of their duties pursuant to Article 39 UK GDPR.
- 7.4 Further provision in relation to specific data flows should be included in Data Protection Agreements.

### 8. Governance: personnel

- 8.1 Each Party must take reasonable steps to ensure the suitability, reliability, training and competence, of any personnel who have access to the Personal Data (and Special Category Personal Data) including reasonable background checks and evidence of completeness should be available on request by each Party.
- 8.2 The Parties agree to treat all Relevant Information as confidential and imparted in confidence and must safeguard it accordingly. Where any of the Parties' personnel are not healthcare professionals (for the purposes of the Data Protection Act 2018) the

employing Parties must procure that personnel operate under a duty of confidentiality which is equivalent to that which would arise if that person were a healthcare professional.

- 8.3 Each Party shall ensure that all personnel required to access the Personal Data (including Special Category Personal Data) are informed of the confidential nature of the Personal Data and each Party shall include appropriate confidentiality clauses in employment/service contracts of all personnel that have any access whatsoever to the Relevant Information, including details of sanctions for acting in a deliberate or reckless manner that may breach the confidentiality or the non-disclosure provisions of Information Law requirements, or causes damage to or loss of the Relevant Information.
- 8.4 Each Party shall provide evidence (further to any reasonable request) that all personnel that have any access to the Relevant Information whatsoever are adequately and appropriately trained to comply with their responsibilities under Information Law and this Agreement.
- 8.5 Each Party shall ensure that:
  - 8.5.1 only those personnel involved in delivery of the Agreement use or have access to the Relevant Information; and
  - 8.5.2 that such access is granted on a strict **Need to Know** basis and shall implement appropriate access controls to ensure this requirement is satisfied and audited. Evidence of audit should be made freely available on request by the originating Data Controller.; and
  - 8.5.3 specific limitations on the personnel who may have access to the Information are set out in the relevant Data Sharing Agreement

### 9. Governance: Protection of Personal Data

- 9.1 At all times, the Parties shall have regard to the requirements of Information Law and the rights of Data Subjects.
- 9.2 Wherever possible (in descending order of preference), only anonymised information, or strongly or weakly pseudonymised information will be shared and processed by Parties, without the need to share easily identifiable Personal Data. The Parties shall cooperate in exploring alternative strategies to avoid the use of Personal Data in order to achieve the Specified Purpose. However, it is accepted that some Relevant Information shared further to this Agreement may be Personal Data/Special Category Personal Data.
- 9.3 Processing of any Personal Data or Special Category Personal Data shall be to the minimum extent necessary to achieve the Specified Purpose, and on a **Need to Know** basis. If either Party:
  - 9.3.1 becomes aware of any unauthorised or unlawful processing of any Relevant Information or that any Relevant Information is lost or destroyed or has become damaged, corrupted or unusable; or
  - 9.3.2 becomes aware of any security vulnerability or breach,

in respect of the Relevant Information it shall promptly (and within 48 hours) notify the other Party. The Parties shall fully cooperate with one another to remedy the issue as soon as reasonably practicable, and in making information about the incident available to the Information Commissioner and Data Subjects where required by Information Law.

9.4 In processing any Relevant Information further to this Agreement, each Party shall:

- 9.4.1 process the Personal Data (including Special Category Personal Data) only in accordance with the terms of this Agreement and otherwise (to the extent that it acts as a Data Processor for the purposes of Article 27-28 GDPR) only in accordance with written instructions from the originating Data Controller in respect of its Relevant Information;
- 9.4.2 process the Personal Data (including Special Category Personal Data) only to the extent as is necessary for the provision of the Specified Purpose or as is required by law or any regulatory body;
- 9.4.3 process the Personal Data (including Special Category Personal Data) only in accordance with Information Law requirements and shall not perform its obligations under this Agreement in such a way as to cause any other Data Controller to breach any of their applicable obligations under Information Law; and
- 9.4.4 process the Personal Data in accordance with the requirements of Information Law and in particular the principles set out in Article 5(1) and accountability requirements set out in Article 5(2) UK GDPR.
- 9.5 Each Party shall act generally in accordance with Information Law requirements, and in particular shall implement, maintain and keep under review appropriate technical and organisational measures to ensure and to be able to demonstrate that the processing of Personal Data is undertaken in accordance with Information Law, and in particular to protect the Personal Data (and Special Category Personal Data) against unauthorised or unlawful processing and against accidental loss, destruction, damage, alteration or disclosure. These measures shall:
  - 9.5.1 Take account of the nature, scope, context and purposes of processing as well as the risks of varying likelihood and severity for the rights and freedoms of Data Subjects; and
  - 9.5.2 Be appropriate to the harm which might result from any unauthorised or unlawful processing, accidental loss, destruction or damage to the Personal Data (and Special Category Personal Data) and having regard to the nature of the Personal Data (and Special Category Personal Data) which is to be protected.
- 9.6 In particular, each Party shall:
  - 9.6.1 ensure that only personnel authorised under this Agreement have access to the Personal Data (and Special Category Personal Data);
  - 9.6.2 ensure that the Relevant Information is kept secure and in an encrypted form, and shall use all reasonable security practices and systems applicable to the use of the Relevant Information to prevent and to take prompt and proper remedial action against, unauthorised access, copying, modification, storage, reproduction, display or distribution, of the Relevant Information;
  - 9.6.3 obtain prior written consent from the originating Party in order to transfer the Relevant Information to any third party;
  - 9.6.4 permit the other Party or their representatives (subject to reasonable and appropriate confidentiality undertakings), to inspect and audit the data processing activities carried out further to this Agreement (and/or those of its agents, successors or assigns) and comply with all reasonable requests or directions to enable each Party to verify and/or procure that the other is in full compliance with its obligations under this Agreement; and

- 9.6.5 if requested, provide a written description of the technical and organisational methods and security measures employed in processing Personal Data.
- 9.7 Each Party shall adhere to the specific requirements as to information security set out in the Data Sharing Agreements.
- 9.8 Each Party shall use best endeavours to achieve and adhere to the requirements of the NHS Digital Data Security and Protection Toolkit.
- 9.9 The Parties' Single Points of Contact set out in paragraph 14 (*Governance: Single Points of Contact*) below will be the persons who, in the first instance, will have oversight of third party security measures.

## **10.** Governance: Transmission of Information between the Parties

- 10.1 This paragraph supplements paragraph 9 (*Governance: Protection of Personal Data*) of this Schedule.
- 10.2 Transfer of Personal Data between the Parties shall be done through secure mechanisms including use of the N3 network, encryption, and approved secure (NHS.net / gcsx) email.
- 10.3 Wherever possible, Personal Data should be transmitted (and held) in pseudonymised form, with only reference to the NHS number in 'clear' transmissions. Where there are significant consequences for the care of the patient, then additional data items, such as the postcode, date of birth and/or other identifiers should also be transmitted, in accordance with good information governance and clinical safety practice, so as to ensure that the correct patient record/data is identified.
- 10.4 Any other special measures relating to security of transfer should be included in a Data Sharing Agreement.
- 10.5 Each Party shall keep an audit log of Relevant Information transmitted and received in the course of this Agreement.
- 10.6 The Parties' Single Point of Contact notified pursuant to paragraph 14 (*Governance:* Single Points of Contact) will be the persons who, in the first instance, will have oversight of the transmission of information between the Parties.

#### 11. Governance: Quality of Information

- 11.1 The Parties will take steps to ensure the quality of the Relevant Information and to comply with the principles set out in Article 5 UK GDPR.
- 11.2 Special measures relating to ensuring quality are set out in the Personal Data Agreement annexed to this Schedule.

#### 12. Governance: Retention and Disposal of Shared Information

- 12.1 The non-originating Party shall securely destroy or return the Relevant Information once the need to use it has passed or, if later, upon the termination of this Agreement, howsoever determined. Where Relevant Information is held electronically the Relevant Information will be deleted and formal notice of the deletion sent to the Party that shared the Relevant Information. Once paper information is no longer required, paper records will be securely destroyed or securely returned to the Party they came from.
- 12.2 Each Party shall provide an explanation of the processes used to securely destroy or return the information, or verify such destruction or return, if requested by the other Party and shall comply with any request of the Data Controllers to dispose of data in accordance with specified standards or criteria.

- 12.3 If either Party is required by any law, regulation, or government or regulatory body to retain any documents or materials that it would otherwise be required to return or destroy under this paragraph 12 *(Governance: Retention and Disposal of Shared Information)*, it shall notify the other Party in writing of that retention, giving details of the documents or materials that it must retain.
- 12.4 Retention of any data shall comply with the requirements of Article 5(1)(e) GDPR and with all good practice including the Records Management NHS Code of Practice, as updated or amended from time to time.
- 12.5 Any special retention periods should be set out in the Data Sharing Agreements.
- 12.6 Each Party shall ensure that Relevant Information held in paper form is held in secure files, and, when it is no-longer needed, destroyed using a cross cut shredder or subcontracted to a confidential waste company that complies with European Standard EN15713.
- 12.7 Each Party shall ensure that, when no longer required, electronic storage media used to hold or process Personal Data are destroyed or overwritten to current policy requirements.
- 12.8 Electronic records will be considered for deletion once the relevant retention period has ended.
- 12.9 In the event of any bad or unusable sectors of electronic storage media that cannot be overwritten, the Party shall ensure complete and irretrievable destruction of the media itself in accordance with policy requirements.

## 13. Governance: Complaints and Access to Personal Data

- 13.1 Each Party shall assist the other in responding to any request made under Information Law made by persons who wish to access copies of information held about them ("**Subject Access Requests**"), as well as any other purported exercise of a Data Subject's rights under Information Law or complaint to or investigation undertaken by the Information Commissioner.
- 13.2 Complaints about information sharing shall be routed through each Party's own complaints procedure but reported to the Single Points of Contact set out in paragraph 14 (*Governance: Single Points of Contact*) below.
- 13.3 The Parties shall use all reasonable endeavours to work together to resolve any dispute or complaint arising under this Agreement or any data processing carried out further to it.
- 13.4 Basic details of the Agreement shall be included in the appropriate log under each Party's Publication Scheme.

## 14. Governance: Single Points of Contact

14.1 The Parties each shall appoint a Single Point of Contact to whom all queries relating to the particular information sharing should be directed in the first instance.

### 15. Monitoring and review

1. The Parties shall monitor and review on an ongoing basis the sharing of Relevant Information to ensure compliance with Information Law and best practice. Specific monitoring requirements must be set out in the relevant Data Sharing Agreement.

# **Financial Provisions and Decision Making Limits**

Part 1 - Financial Limits and Approvals for Primary Care

- 1. The ICB shall ensure that any decisions in respect of the Delegated Functions and which exceed the financial limits set out below are only taken:
  - 1.1 by the following persons and/or individuals set out in column 2 of Table 1 below; and
  - 1.2 following the approval of NHS England (if any) as set out in column 3 of the Table 1 below.
- 2. NHS England may, from time to time, update Table 1 by sending a notice to the ICB of amendments to Table 1.

Table 1 – Financial Limits		
Decision	Person/Individual	NHS England Approval
General		
Taking any step or action in relation to the settlement of a Claim, where the value of the settlement exceeds £100,000	ICB Chief Executive Officer or Chief Finance Officer or Chair	NHS England Head of Legal Services and Local NHS England Team Director or Director of Finance
Any matter in relation to the Delegated Functions which is novel, contentious or repercussive	ICB Chief Executive Officer or Chief Finance Officer or Chair	Local NHS England Team Director or Director of Finance or NHS England Region Director or Director of Finance or NHS England Chief Executive or Chief Financia Officer
Revenue Contracts		
The entering into of any Primary Care Contract or Arrangement which has or is capable of having a term which exceeds five (5) years	ICB Chief Executive Officer or Chief Finance Officer or Chair	Local NHS England Team Director or Director of Finance
	ill not have delegated or directed responsibility for dec ut the ICB may be required to carry out certain admir and Liability).	

### Mandated Assistance and Support

## 1. Primary Dental Services

- 1.1 NHS Business Services Authority has existing agreements with NHS England to support its delivery of the following services:
  - 1.1.1 Contract management end-to-end administration of contract variations and other regional team/ICB support activities;
  - 1.1.2 Performance management provide mid and end of year administration process to support regional teams and ICBs and undertake risk based assurance reviews PPV can also be instigated by the ICS or Counter Fraud;
  - 1.1.3 Clinical assurance reviews provide clinical assurance of quality of dental services delivered, working in collaboration with regional teams/ICBs to identify and seek to address any concerns;
  - 1.1.4 Provide data reports to teams defining quantity and service delivery at a contractor level.

## 2. Primary Ophthalmic Services

- 2.1 NHS Business Services Authority have existing agreements with NHS England to support its delivery of the following services:
  - 2.1.1 Contract management. End-to-end administration of new contract applications, contract variations and contract terminations.
  - 2.1.2 Contract assurance. Administration of the annual contractor assurance declaration and additional in-depth assurance declaration where appropriate. Provision of assurance reports at ICS and contractor level, supporting further assurance decisions.
  - 2.1.3 GOS complaints. Administration of the annual GOS complaints survey.
  - 2.1.4 Post-Payment Verification (PPV). End-to-end process for identifying and verifying GOS claims as part of the national PPV framework. This includes obtaining and reviewing claims and carrying out a financial recovery where appropriate. PPV can also be instigated by the ICS or Counter Fraud.
  - 2.1.5 GOS 4 pre-authorisation of repair or replacement glasses.

## 3. Pharmaceutical Services and Local Pharmaceutical Services

- 3.1 NHS Business Services Authority has existing agreements with NHS England to support ICBs to discharge their assurance responsibilities by the delivery of the following services to ICBs:
  - 3.1.1 Performance management direct support to commissioners and community pharmacy contractors to implement corrective and preventative intervention;
  - 3.1.2 Contract assurance administration of the annual contractor assurance declaration and additional in-depth assurance declaration where

appropriate, provision of assurance reports at ICS and contractor level supporting further assurance decisions by the ICB;

3.1.3 Post-Payment Verification (PPV) – end-to-end process for identifying and verifying claims as part of the national PPV framework to support ICBs to play their part in ensuring compliance with contractual requirements and delivery of quality and value for money. This includes obtaining and reviewing claims, investigation of outliers and other potential inappropriate claims, along with referrals and investigatory reports to the ICB to consider and decide overpayment recoveries, and carrying out the financial recovery where appropriate. PPV can also be instigated by the ICS or Counter Fraud.

# 4. Support Services directed by DHSC

- 4.1 NHS Business Services Authority is directed by DHSC to undertake specific activities as well as having existing agreements with NHS England to support its delivery of primary care services. These include (without limitation):
  - 4.1.1 The administration of national payment platforms for primary care services to dentists, pharmacy contractors, appliance contractors, oxygen contractors and special school eye care providers;
  - 4.1.2 The calculation of payment for covid-19 and flu vaccinations to PCNs and GP practices as well as payments to Dispensing Doctors and prescribing only doctors for personal administration claims and sharing this information accurately and in a timely manner;
  - 4.1.3 Clinical advisory support;
  - 4.1.4 Administration functions;
  - 4.1.5 Assurance services performance and contract management of primary care providers;
  - 4.1.6 The provision of information to primary care organisations for all contractor groups via standardised reporting (eg. ePACT2, eDEN and eOPS);
  - 4.1.7 Working with NHS England and ICB Counter Fraud Teams to reduce loss across the system.

# Local Terms

A Memorandum of Understanding has been agreed and signed between NHS Somerset Integrated Care Board and NHS England South West Direct Commissioning Directorate and the other 6 ICBs in the South West. The MOU is all binding and all relevant for inclusion in this schedule sets out how via a Collaborative Commissioning Hub, NHSE South West Direct Commissioning Directorate and ICBs will work together to support delivery of delegated commissioning for pharmaceutical services, general ophthalmic services, and dental services (POD). The MOU is an appendix of this Delegation Agreement.

The MOU is effective as of 1<sup>st</sup> April 2023 until 31 March 2024. The MOU is underpinned by;

- a Quality Framework
- Finance Standard Operating Proceedure
- Data Processing Agreement for the Collaborative Commissioning Hub.

The Collaborative Commissioning Hub will be resourced by NHS England South West existing POD teams following an aligned staffing model deployed to facilitate the ICB in undertaking the Delegated Functions as set out in the South West Memorandum of Understanding;

- Primary Medical Services Functions including general practice transformation
- Primary Dental Services and Prescribed Dental Services Functions
- Primary Ophthalmic Services Functions
- Pharmaceutical Services and Local Pharmaceutical Services Functions

The POD Commissioning Team will transfer to a host ICB on 1 July 2023.

# Deployment of NHS England Staff to the ICB

# Note: This schedule relates to the Deployment of Staff who are employed by NHS England only.

## Deployment of NHS England Staff

- 1. NHS England may deploy Staff to the ICB for the purposes of carrying out the Delegated Functions.
- 2. The Parties have agreed that arrangements for the provision of NHS England Staff and the associated employment model envisaged by section 5.9 of the HR Framework <u>https://www.england.nhs.uk/wp-content/uploads/2021/06/B1427-Human-resources-framework-for-developing-integrated-care-boards-version-2-March-2022.pdf</u>) will be determined by the National Moderation Panel convened for this purpose and endorsed by NHS England's Executive Group.
- 3. The Parties agree and acknowledge that the Staffing Models will be developed in conjunction with the ICB and are subject to the decision of the National Moderation Panel and cannot be varied without the express agreement of NHS England.
- 4. A proposal for a variation to any Staffing Model must be made by means of a formal submission to the National Moderation Panel which will determine the proposal, following which the proposal if approved, will be endorsed by NHS England's Executive Group.
- 5. Subject to any variation made in accordance with paragraphs 3 and 4 above, a Staffing Model determined in accordance with paragraph 2 will apply for the duration of this Agreement.

# Availability of NHS England Staff

- 6. In addition to any Staff deployed in any communicated Staffing Model arrangement, NHS England may deploy additional Staff to the ICB to perform administrative and management support services together with such other services specified in Error! Reference source not found. (Local Terms) (the "Services") so as to facilitate the ICB in undertaking the Delegated Functions pursuant to the terms of this Agreement.
- 7. NHS England will take all reasonable steps to ensure that the NHS England Staff deployed for the purposes of carrying out the Delegated Functions shall:
  - a. faithfully and diligently perform duties and exercise such powers as may from time to time be reasonably assigned to or vested in them; and
  - b. perform all duties assigned to them pursuant to this Schedule 8.
- The ICB shall notify NHS England if the ICB becomes aware of any act or omission by any NHS England Staff which may have a material adverse impact on the provision of the Services or constitute a material breach of the terms and conditions of employment of the NHS England Staff.
- 9. NHS England shall use all reasonable efforts to make its Staff available for the purposes of this Schedule 8 whilst the NHS England Staff are absent:
  - a. by reason of industrial action;
  - b. as a result of the suspension or exclusion of employment or secondment of any Staff by NHS England;

- c. in accordance with the NHS England Staff's respective terms and conditions of employment and policies, including, but not limited to, by reason of training, holidays, sickness, injury, trade union duties, paternity leave or maternity or where absence is permitted or required by Law;
- d. if making the NHS England Staff available would breach or contravene any Law;
- e. as a result of the cessation of employment of any individual NHS England Staff; and/or
- f. at such other times as may be agreed between NHS England and the ICB.

## Employment of the NHS England Deployed Staff

- 10. NHS England shall employ their Staff and shall be responsible for the employment of their Staff at all times on whatever terms and conditions as NHS England and their Staff may agree from time to time.
- 11. NHS England shall pay their Staff their salaries and benefits and make any deductions for income tax liability and national insurance or similar contributions it is required to make from the Staff's salaries and other payments.
- 12. NHS England shall not, and shall procure that the NHS England Staff shall not, hold themselves out as employees of the ICB.

## Management of NHS England staff

- 13. NHS England where appropriate, shall in consultation with the ICB, make arrangements to ensure the day-to-day control of the activities of their Staff is shared with the ICB and deal with any relevant management issues concerning their Staff including, without limitation, performance appraisal, discipline and leave requests.
- 14. The ICB agrees to provide all such assistance and co-operation that NHS England may reasonably request from time to time to resolve grievances raised by NHS England Staff and to deal with any disciplinary allegations made against NHS England Staff arising out of or in connection with the provision of the Services which shall include, without limitation, supplying NHS England with all information and the provision of access to all documentation and NHS England Staff as NHS England requires for the purposes of considering and dealing with such issues and participating promptly in any action which may be necessary.

### **Conduct of Claims**

- 15. If the ICB becomes aware of any matter that may give rise to a claim by or against a member of NHS England Staff, notice of that fact shall be given as soon as possible to NHS England. NHS England and the ICB shall co-operate in relation to the investigation and resolution of any such claims or potential claims.
- 16. No admission of liability shall be made by or on behalf of the ICB and any such claim shall not be compromised, disposed of or settled without the consent of NHS England.

### **Confidential Information and Property**

- 17. For the avoidance of doubt, this paragraph 17 (Confidential Information and Property) is without prejudice to any other provision of this Agreement in relation to confidential information.
- 18. It is acknowledged that to enable the NHS England Staff to provide the Services, the Parties may share Confidential Information.
- 19. The Parties agree to adopt all such procedures as the other party may reasonably require and to keep confidential all Confidential Information.

#### Intellectual Property

20. All IPR made, written, designed, discovered or originated by Staff (People Resources) deployed by NHS England, shall be the property of NHS England to the fullest extent permitted by Law and NHS England shall be the absolute beneficial owner of the copyright in any such IPR.

# Mandated Guidance

# Primary Medical Care

- Primary Medical Care Policy and Guidance Manual.
- The 'Principles of Best Practice' and any other guidance relating to *the Premises Cost Directions 2013.*
- Guidance relating to the Minimum Practice Income Guarantee.
- Guidance relating to Primary Medical Care discretionary payments.
- Guidance for Commissioners: Interpreting and Translation Services in Primary Care.
- Framework for Patient and Public Participation in Primary Care Commissioning.
- <u>NHS England National Primary Care Occupational Health Service Specification.</u>
- Guidance relating to list cleansing in relation to Primary Medical Care providers.
- Guidance relating to mergers and closures of GP practices and/or Primary Medical Care providers.
- Guidance relating to Primary Medical Care and POD contract reviews.
- Guidance relating to the escalation of complaints from a named 'performer'.
  Including: <u>Framework for Managing Performer Concerns.</u>

# Pharmaceutical Services and Local Pharmaceutical Services

- Pharmacy Manual.
- <u>NHS England National Primary Care Occupational Health Service Specification</u>.
- The NHS Pharmacy Regulations Guidance 2020<sup>[1]</sup>.
- Guidance for ICSs and STPs on transformation and improvement opportunities to benefit patients through integrated pharmacy and medicines optimisation.

# Primary Ophthalmic Services

- Policy Book for Eye Health.
- NHS England National Primary Care Occupational Health Service Specification.

# Primary and Prescribed Dental Services

- Policy Book for Primary Dental Services.
- Securing Excellence in Commissioning NHS Dental Services.
- Securing Excellence in Commissioning NHS Dental Services: Key facts.
- Securing Excellence in Commissioning NHS Dental Services: FAQs.
- Quick Guide: Best use of unscheduled dental care services.
- How to update NHS Choices for Dental Practices.
- Flowchart for managing patients with a dental problem/pain.
- Guidance on NHS 111 Directory of Services for dental providers.
- <u>Definitions Unscheduled Dental Care</u>.
- Introductory Guide for Commissioning Dental Specialties.
- Guide for Commissioning Dental Specialties: Orthodontics.
- Guide for Commissioning Dental Specialties: Oral Surgery and Oral Medicine.
- Guide for Commissioning Dental Specialties: Special Care Dentistry.
- Guide for Commissioning Service Standards: Conscious Sedation in a Primary Care Setting.
- Commissioning Standard for Dental Specialties: Paediatric Dentistry.
- <u>Commissioning Standard for Urgent Dental Care</u>.
- <u>Commissioning Standard for Restorative Dentistry</u>.

<sup>&</sup>lt;sup>[1]</sup> <u>https://www.england.nhs.uk/primary-care/pharmacy/pharmacy-manual/nhs-pharmacy-regulations-guidance-2020/</u>

- Commissioning Standard for Dental Care for People with Diabetes.
- Accreditation of Performers and Providers of Level 2 Complexity Care.
- NHS England National Primary Care Occupational Health Service Specification.
- Dental Access Controls.

# Finance

- Guidance on NHS System Capital Envelopes.
- Finance and Payments Guidance for Community Pharmacy, Dental and Primary Care Ophthalmology Services Delegated to ICBs from 2022.
- <u>Managing Public Money (HM Treasury)</u>.
- Guidance relating to Personal Service Medical Reviews.
  Including: Implementing Personal Medical Services Reviews.
- Dental Commissioning and Financial Management Guidance.

## Workforce

• Guidance on the Employment Commitment.

## Other Guidance

- National Guidance on System Quality Groups.
- Managing Conflicts of Interest in the NHS.
- Arrangements for Delegation and Joint Exercise of Statutory Functions.
- Guidance relating to procurement and provider selection.
- IG Guidance relating to serious incidents.
- All other applicable IG and Data Protection Guidance.
- Any applicable Freedom of Information protocols.
- Any applicable guidance on Counter Fraud, including from The NHS Counter Fraud Authority.
- Any applicable guidance relating to the use of data and data sets for reporting.
- Any applicable guidance relating to the commissioning and management of clinical waste in primary care e.g.
  - o Including: Management and disposal of healthcare waste.

### Administrative and Management Services

- 1. The ICB shall provide the following administrative and management services to NHS England:
  - 1.1 the administrative and management services in relation to the Capital Expenditure Functions and the Capital Expenditure Funds as more particularly set out in this Part 1 of this SCHEDULE (*Administrative and Management Services*); and
  - 1.2 the administrative and management services in relation to the Section 7A Functions and Section 7A Funds as more particularly set out in Part 2 of this Schedule 10.
  - 1.3 the administrative and management services in relation to other Reserved Functions as more particularly set out in Part 3 of this SCHEDULE (*Administrative and Management Services*).

Part 1: Administrative and/or Management Services and Funds in relation to the Capital Expenditure Functions

- 1. The Parties acknowledge that the Capital Expenditure Functions are a Reserved Function.
- 2. The Parties further acknowledge that:
  - 2.1 accordingly, the Delegated Funds do not include any funds in respect of amounts payable in relation to the Capital Expenditure Functions ("Capital Expenditure Funds"); and
  - 2.2 NHS England remains responsible and accountable for the discharge of the Capital Expenditure Functions and nothing in Part 1 of this SCHEDULE (*Administrative and Management Services*) shall be construed as a divestment or delegation of NHS England's Capital Expenditure Functions.
- 3. Without prejudice to paragraph 3 above, the ICB will comply with any Mandated Guidance issued in relation to the Capital Expenditure Functions and shall (on request from NHS England) provide the following administrative services to NHS England in respect of the Capital Expenditure Funds:
  - 3.1 the administration and payment of sums that NHS England has approved as payable in relation to the Capital Expenditure Functions;
  - 3.2 if requested by NHS England and taking into account (i) any other support or services provided to NHS England by NHS Property Services Limited or otherwise and (ii) any Mandated Guidance issued in respect of the Capital Expenditure Functions, the provision of advice and/or recommendations to NHS England in respect of expenditure to be made under the Capital Expenditure Functions; and
  - 3.3 such other support or administrative assistance to NHS England that NHS England may reasonably request in order to facilitate the discharge by NHS England of its responsibilities under or in respect of the Capital Expenditure Functions.
- 4. NHS England may, at the same time as it allocates the Delegated Funds to the ICB under Clause 9.11, transfer to the ICB such amounts as are necessary to enable the discharge of the ICB's obligations under this Part 1 of SCHEDULE (*Administrative and Management Services*) in respect of the Capital Expenditure Functions.

Part 2 - Administrative and/or Management Services and Funds in relation to Section 7A Functions

- 1. The Parties acknowledge that the Section 7A Functions are part of the Reserved Functions.
- 2. The Parties further acknowledge that:
  - 2.1 accordingly, the Delegated Funds do not include any funds in respect of amounts payable in relation to the Section 7A Functions (whether such arrangements are included in or under Primary Care Contracts or Arrangements or not) ("Section 7A Funds"); and
  - 2.2 NHS England remains responsible and accountable for the discharge of the Section 7A Functions and nothing in this Schedule 10 Part 2 shall be construed as a divestment or delegation of the Section 7A Functions.
- 3. The ICB will provide the following services to NHS England in respect of the Section 7A Funds:
- 4. the administration and payment of sums that NHS England has approved as payable under or in respect of arrangements for the Section 7A Functions; and
- 5. such other support or administrative assistance to NHS England that NHS England may reasonably request in order to facilitate the discharge by NHS England of its responsibilities under or in respect of the Section 7A Funds.
- 6. NHS England shall, at the same time as it allocates the Delegated Funds to the ICB under Clause 9.11, allocate to the ICB such amounts as are necessary to enable the discharge of the ICB's obligations under this Schedule 10 Part 2 in respect of the Section 7A Funds.

## Part 3: Administrative and/or Management Services and Funds in relation to other Reserved Functions

- 1. NHS England may ask the ICB to provide certain management and/or administrative services to NHS England (from a date to be notified by NHS England to the ICB) in relation to the carrying out of any of the Reserved Functions.
- 2. If NHS England makes such a request to the ICB, then the ICB will, but only if the ICB agrees to provide such services, from the date requested by NHS England, comply with:
- 3. provisions equivalent to those set out above in relation to the Capital Expenditure Functions (Part 1 of this Schedule 10) and the Section 7A Functions (Part 2 of this Schedule 10) including in relation to the administration of any funds for such functions but only to the extent that such provisions are relevant to the management or administrative services to be provided; and
- 4. such other provisions in respect of the carrying out of such management and administrative services as agreed between NHS England and the ICB.

#### POD Risk Register Somerset - March 2023

ю I	BICCH Risk Description	Date Raised Date Closed Risk	: Owner	Risk Coordinators	Risk Category	Risk Sub Category	Further Details	Current Likelihood	Current Impact	Current Score	Target Likelihood	arget Impact	Target Date	Mitigated Score	Risk Updates	Causes	Consequences	Controls & Milligation	Linked Risks
10338	ICID There is a risk to be update and contract accesses of the CPOP shared on the Book Netting accesses of the CPOP shared on the Book Netting accesses of the CPOP shared on the Shared Netting 2022.	07/16/2022 News (Reck	i Holmos 1 holmos 1 @eha.net)	From Durreport (Gaverport)(Brit And), Levile Fliggs (Walergoggiften and), Jewry Caline (Walergoggiften and), Jewry Caline (Walergoggiften and), Levile Heimes (Mak) Andreas (Byrka, nd), Michael Toy (Inches bryg)(He and)	Performance (operational and financia)			3	3	9	3	3	31/03/2023	9	Bears General (2023-154) 154-86) (4111/2022 Related to the application dimension frame. Confusing to shale radional data with COL and proper to area.	The activity data was collected through Planmoutcomes a lacent heat by tHIS England with the 31st Match 202 The dauge of process in Argin required the planmary that the second second second second second second second completed comparison of the second second second second completed comparison of the second second second second completed comparison of the second s	Dualskis to describe data in a timety way with the individual project taxes or CES. Substitution of the second second second second second data was to second second second second second second data was to second second second second second second data second to second second second second second data and the project taxes in under the second data second second data data second data second data second d	Brang axiable data problet by 9e National Planmary Taum	There is a risk to the spatial and contraned nations of the COTOPS annual in the float theory weat is also part for CPT Frage of the 2023-2023.
8327	AI ICB. There is a real of selected phoresonable and end- for particle and the select as a spin surger of recorded upplemed clauses within Commonly Pleasancy across Book Next	s 1805/2022 Liz	Cosford (itz.cosford@phs.net)	Fone Deverport (f. deverport)(ghta.ndt), Leake Rigge (Haleringge/ghta.ndt), Burco Greeners and Sone State Amry Collins (gmy collins (ghta.ndt), Shok Helmans (Aski-brand (ghta.ndt), Michaele Toy (Inicitede toy@pha.ndt)	Operational	Business Continuity		4	3	12	3	2	01/10/2022	6	Find Developed (2025-1516) 2120). A realisive guide has based outputs and the second output of the second output of the second output of the second output of the second development of the second output of the second commany of participants, and a second output of the second output of the second output of the second output of the second output of the second output of the second development of the second output of the second and second second output of the second output of the second output of the second output of the second development of the second output of the second output of the second development of the second output of the second output of the second development of the second output of the second outpu	Workfore Pressves for Community Phermatists acros England – which has caused a high number of urginare distants in community phermatists.	<sup>10</sup> Priferiti anabili to actess medicines, itoreasing duman in the previous providing queries to practices an intervening test of providence.	Relevent organizered discare policy with Load Planmacultud Constitues across the South that is even as annulated operands. It stars, 61 August 2022 Plansing communications processors that di Constructly Planmacultud to the plants are compared and annual stars, 61 August 2022 Enauty Constructions with yorks of Constructly Planmacultud and discare and annual stars and annual stars and annual discare and annual stars and annual stars and annual annual by the stars and annual stars and annual stars and by the stars and annual stars and annual stars and by the stars and discare and the the plants and starts and annual the stars and discare and the the plants and starts and annual the stars and discare and the the plants and starts and annual the stars and discare and the the plants and starts and annual stars. 31 May 2022	There is a did of relaxed planneaded landcas for patients as there are a plan reduct of accorded trajectored discusses attitic Contractly Thermacy across Book Mudi. There is a relax of resolution planneaded and uncless for plannead trajectored planneade of resolution planneaded and uncless for plannead trajectore as there are a relax of relaxed planneaded and uncless for plannead planneade of relaxed planneaded and uncless for plannead planneade of relaxed planneaded and uncless for planneady There is a relax of relaxed planneaded and uncless for planneady There is a relax of relaxed planneaded and uncless for planneady There is a relax of relaxed planneaded and uncless for planneady There is a relax of relaxed planneaded and uncless for plannead there is a relaxed relaxed planneaded and uncless for plannead planneade and the relaxed planneaded and uncless for planneade
8330	ICB Ingla nature of Commuty Phormag processing Indicating aggregation of the procession across the form West.	1805/2022 La C	londford (kr.cosflond@inte.net)	Fina Durrepott (Idampott)giften net), Lasle Riggs (Idampott)giften net), Lasle Riggs (Idampott)giften net), Lasle Net (Idampott)giften net), Las Disa- tisten (Idam tigheta net), Las Disa- tistense (Idak Idams) (Igan and Harbels Toy (nichels toy@ris.net) Michels Toy (nichels toy@ris.net)	Operational	Buainess Continuity		3	3	9	3	2	30/08/2022		Fixe Description (2022-15-15 09:15-11) This continues to live an description of the second s	Workforce pressures which have mean't Community Permission modeling to realize the provision of permissionful anniholes across the week.	Plannary Claures - reduced access to plannacedita analoss for patients, particularly later alternoon weaksdays are at the weakand.	Deur ommatodon bil skälebbin ä urbinsken	Ngh nahlar of Carmonly Plannary problem induling supplementary hose problem across the Bodh Weat
11400	AI (C) Market est applications exosulted than Login. Plannacia control for 00 (Mir et al. 1996) In Sandory Bores	0 07/02/2023	NBAS Heamas (nBBA heamas S@rina.net)	Fora Devergort (Edvergofter, and, Steron Greenes (dharongnawas(girta .nd;) Jenny Colins (jenny, collins (girta, nd;)	Operational	Paople	An initial marking paper has been devolved and advance with ICBs to provide an use date, thates papers will be a set of the the set of the set of the set of the each application is being understaten to identify if the Socih Well has also at of the set of the socih Well has also at of the set of the set of the set of the Manager to agree an approach to managing these down Mage setablish have been devolped and alward with ICBs	4	4	16	4	3	31/03/2023	12		Logis's Plasmacy Haud Office has made a strategic decision to clease all of their plasmacias who are based Sainsbury's Brunes	Reduced Community Premising problem across the South West, controlly this will have an inspect on 25 Burger Hammander, this equation to 13-10 to the impact on assertizationancefloading commission impact on assertizationancefloading commission sentence across the Booth View.	Working closely with Libyds Phermany Development Manager	Narhet est applications excelled from Lippis Parameteis arrors the Boch Ward who are loaded in Balankarys Storas Narhet est applications excelled from Lippis Permission arrors the Boch Market and applications excelled from Lippis Permission arrors the Boch Market and applications excelled from Lippis Permission arrors the Boch
Dental 1	Al ICB COVID - There is a risk of deteriorating adult ora health due to the impact of the Covid-19 pandemi on the provision of secondary and community can dental services - adults	C Steve	n Sylvester	Mel Smoker	Patient safety and quality of care			4	4	16	4	4		16		Significant impact of Covid-19 response in dentistry due to actualisecontary care prioritisation and provision alongsiste IPC risks and implications for service delivery/patient flow. Waiting lists impacted		Provision of dental services across acute/community providers. Triage via 111 and Durtal Higheina. Managing patient expectations - Non complex tooth extraction should be undertaken in the primary care dartal chair Orgoing work by No Cal Scapery MCN Addl Cade on ch twe as great an inpact as Paedos on waiting lata makeword or under called dental dhair capacity bit forms additional commissioning in Tarkieword or under called dental dhair capacity bit forms additional commissioning in Tarkieword or under called dental dhair capacity bit forms additional commissioning in Tarkieword or under called dental dhair capacity bit forms additional commissioning in the second second called dental dhair capacity bit forms additional commissioning in the called dental dhair capacity bit forms additional commissioning in the second second dental dhair capacity bit forms additional commissioning in the second dental dhair capacity bit forms additional commissioning in the second dental dhair capacity bit forms additional commissioning in the second dental dhair capacity bit forms additional commissioning in the second dental dhair capacity bit forms additional commissioning in the second data dhair capacity bit forms additional commissioning in the second data dhair capacity bit forms additional commissioning in the second data dhair capacity bit forms additional commissioning in the second data data data data capacity bit forms additional commissioning in the second data data data data data data capacity bit forms additional data data data data data data data da	West who are based in Sainsbury. Stores
Dental 2	AI (CB) COVID There is a nin wir divativating pandatire and having has to have impact of the Coxid-19 pandemic on the provision of secondary and community care detail services - children	s Shew	e Sylvester	Mel Smoker	Patient solicity and quality of care			4	4	16	4	4		16	1711/122 Donal Taxin Risk Meeting Update - Work is engoing to by and menage for waiting lists. Boore to remain the same. 8/02/27: no change	capacity. Acutalisacondary care trust prioritisation during pandemi pariod. Community/Bpacial Care services supporting urgent can provision during pandemic period.	ic e	International constant, and the second secon	
Dental 3	Al ICB COVID There is a risk of destributing pandatic register of the provision of primary care dental sentices.	D Steve	n Sylvester	Mel Sincker	Patient safety and quality of care			4	4	16	4	4		16	17/11/22: Dental Team Risk Meeting Update - No change, risk decre to remain the same.		in pain.	Provision of aerolean through providers with bidge through 111 and danta hubbles. Dialk Finand Jonatons pairs analysis—with 7 and through hubbles. Bapanisma Toshthrakheg processment runderswy to extend the successful Deven plot to 21 29 VICBs in seas of depination. First Dental Bases plot infroductied in the region. All new mandatory detail contracts to have a Starting Well Care 6-2 child focus.	0
Dental 4	Al ICB COVID - There is a risk of deterionating adult or a health due to the impact of the Covid-19 pandemi on the provision of primary care dental services.	c	e Sylvester	Mel Smoker	Patient safety and quality of care			4	4	16	4	4		16	17/11/22: Dental Tearn Risk Meeting Update - No change, risk score to remain the same. 8/02/23: No change		Workforce tatigues as a result of operating services durine carefurne rearied a AccessAltreneased wait times for population resulting in deteriorating on theath, including whereable program in impact re. health inequalities. I increased artificitor personiting and increased use of analgesia for those in pain. Workforce tatigues as a result of operating services during andremic periods.		
Dental 5	All ICB COVID - Risk that dental capacity (routine and urgent) is currently insufficient to meet demand given context of Covid and impact on services. Resulting in patient complaints and services reporting Seeing under pressure.	Steve	n Sylvester	Mel Smoker	Patient safety and quality of care			4	3	12	4	3		12	8/02/23: No change	Impact of Covid on capacity of services.		Commissioning of extitional surgert care capacity in primary and urgent diretal care. Mobilisation of the Stabilisation pathwary South West Dental Reform Programme Development of the Communications and Engagement Strategy for the SW Dental Reform programme	
Dental 6	All ICB There is clinical risk to patients waiting to be reviewed through the IFR process	Steve	e Sylvester	Mel Smoker	Patient safety and quality of care	Performance		3	3	•	з	3		•	8/02/23: No change	The number of referrals exceed the number of cases the can be reviewed in the monthly meetings	at Patients treatments will be delayed if they are not approved by the panel in a timely manner. This may hav a detrimental effect on the overall outcome of the	Review of process taking place and being reviewed regulary. This is an interim errorgment whilst support is considered	
Dental 7	All ICB Delay in decision regarding the specification for th Special Care and Paediatric PDS contracts due t expire 31st March 2024	e Steve	e Sylvester	Mel Smoker	Patient safety and quality of care	Innovation		2	5	10	2	5		10	26/01/23 - Data packs being finalised. Once completed to arrange meetings with ICBs to discuss next steps	Commissioning responsibility being passed to the ICBs from April 2023	Contracts have been extended several times and are no eligible for further extension. Unsettling for service providers with potential impact on	t Data packs being prepared for ICBs to share information as soon as possible to agree actions in advance	
Dental 8	All ICB There is a risk of having a significant dental underspend		n Sylvester	Mel Smoker	Performance			5	4	20	5	4		20	01/11/22. Durfall Team Risk Meeting Update - The team discussed the mitigations we have in place. There is a plan for when paper are presented to 000 which is followed by a Project plan to made. 20/2/23 - Untre discussions nr GB bodyes arranged to emane whole team can support and understand potential spend mode.	Full financial forecast not available until haif way through the financial year constructs. Contracts contracts to be handed back Datys is getting (20 we on consissioning plans and commit to funding. Dates to strongwated Dates to strongwated. Dates to the strongwated plans and applications to appre- prian and agreement on what the payment structure strong to the strong to the strong to the strong to back to be strongwated.	aerice etoxision Funds for details services would not be spent on dentail services and may be redistributed back to primary care og or ICBs, Causing econom annog (CBs newly assuming responsibility for dentail services Risk with dentail access and harm to patients.	making the supports the entoient operation of the CCP and indexang indexement of ICBs	
Dental 9 Dental 10	All ICB End dated contracts - There is a risk that the end dated contracts are not commissioned in sufficie time All ICB Phase 2 Maniatray Dental Services Proviment		e Sylvester	Mel Smoker Mel Smoker	Patient Safety and Quality of Care			3	4	12	3	4		12	01/11/22. Dental Team Risk Meeting Lipdate - Risk remains unchanged 8/02/23: No change in score - continue to work on the information to support ID68 decision	Capacity from both Dental team and systems to agree plan and agreement on what the payment structure should be	The current providers cease to provide the service and patients who rely on care in the home do not have access	There are providers in place who may be able to accept an extension to contract if we are able to fund the additional costs	
Dental 10	All ICB Phase 2 Mandatory Dental Services Procurement All ICB Risk of practices with a low UDA value		e Sylvester e Sylvester	Mel Smoker Mel Smoker	Patient safety and quality of care Financial and vfm			3	4	12	3	4		12		should be Risk that the Phase 2 procurement for activity lost arour the SW does not appeal to practices and activity is not r commissioned Historical IDA rates that are not consistent arross the		Deci22 Agreement secured at DOG to scope costs/procurement requirements of implementing a regional wide service. There are few controls to change the UDA value, as this has a consequence on other	
Denial 11	relinquishing their contracts		-					4	5	20	4	5				Historical UDA rates that are not consistent across the SW or nationally means it is difficult to recruit a workfore where the values are low.		contracts and budgets. Since 1 October 2022 there is a new minimum £23 UDA value	
Dental 12	All ICB Access to theatre space for CDS providers for adult and child GAs	Steve	e Sylvester	Mel Smoker	Patient safety and quality of care	Performance		3	4	12	3	4		12	26/01/23 - Childrens and Young Persons revovery group in discussions with secondary care 8/02/23: No change	Trusts restricting access to CDS providers and prioritisin theatre space for other specialisties	ng Delays in treatment for patients	Engaging with Medical Directorate and ICBs to facilitate changes	

Dental 13		excess of :	mes for primary care orthodontics is in 12 years in some practices		Mel Smoker Perform			4	3	12	4	3	12	1912/22 Constant num Risk Marking Luissen ta num in The Dalay is individue tautument and partiert paraceceptate watering fire should have block determine watering fail ansessement and parameter and parameter for particular, specifically the answer in 2023. Marking fail Allays in branch and parameter and parameter and parameter for particular, specifically the answer and or parameter and parameters
Dental 14	Al	unable to s	who have had Head and Neck Cancer are see a Dentist for ongoing care needed it of their cancer treatment.	Steve Sylvester	Mel Smoker Patient quality o	safety and of care		з	4	12	2	4	8	Informal discussions are public and team working on a solidar BIOD2 - in charge is some BIOD2 - in c
Dental 15	Al	II ICB Child Frien	ndly Dental Practice scheme at capacity	Steve Sylvester	Mel Smoker Perform		Pastert sufety and quality of care	4	4	16	4	4	16	20022 - I do service a Doort. Bas were new as a 14 Poer There are these particles models in the COP point particles who has been highly index and has been highly index ande
Dental 16	Sorr	TA21 & EX	Phase 1 MDB Procurement providers for XXI are unable to nervult sufficient a to deliver the contracts	Steve Sylvester	Mel Smoker Operati	ional		4	4	16	4	4	16	10/12/2 Developed Taxen Risk Marcing Logistar, Tire van Kirl Hoff, Marcing Logistar, Tire van Kirl Hoff, Marcing Logistar, Statusturg Logistar, Statusturg Logistar, Marcing Logistar, Statusturg Logistar, Statu

#### **Risk Scoring**

**Regional Reporting** 

#### Heatmap/RAG

	Impact											
Likelihood	1 - Very Low	2 - Low	3 - Moderate	4 - High	5 - Very High							
5 - Very Likely	5	10	15	20	25							
4 - Likely	4	8	12	16	20							
3 - Possible	3	6	9	12	15							
2 - Unlikely	2	4	6	8	10							
1 - Rare	1	2	3	4	5							

Low Risk (1-6)
Moderate Risk (8-10)
High Risk (12-16)
Extreme Risk (20-25)

#### Likelihood Score/Descriptors

5	Very Likely	Expected to occur, >50% at least daily
4	Likely	Will probably occur, 21-50%, at least weekly
3	Possible	Should occur at some time, 6-20%, at least monthly
2	Unlikely	Could occur at some time, 1-5%, at least annually
1	Rare	Only occurs in exceptional circumstances, <1%, 1-5 year strategic risk

Risk score/grade	Review Frequency	Reporting Level
1-6	At least quarterly	Primary Care Collaborative
8-10		Primary Care Collaborative
12-16	Monthly	Primary Care Collaborative
20-25	At least monthly	Primary Care Collaborative- DCSLT

Risk Types
1. Patient safety and
quality of care
2. Sector performance
(operational and financial)
3. Innovation ( <u>e.g.</u> new
ways of working, integration of services and
new models of care)
4. Financial risk and
Value for Money (VfM)
5. Legal compliance and
regulatory risk (e.g. statutory and legal
compliance risks)
6. Reputational risk
7. Operational risk,
including technology and information/data risk,
and external event risk.